Patientcentred

care:

Improving quality and safety through partnerships with patients and consumers



AUSTRALIANCOMMISSIONON SAFETYANDQUALITYINHEALTHCARE

This work is copyright. It may be reproduced in whole or part for study or training purposes subject to the inclusion of an acknowledgement of the source. Reproduction for purposes other than those indicated above requires the written permission of the Australian Commission on Safety and Quality in Health Care, GPO Box 5480 Sydney NSW 2001 or mail@safetyandquality.gov.au

ISBN: 978-0-9870617-1-3

Suggested citation:

Australian Commission on Safety and Quality in Health Care (2011), Patient-centred care: Improving quality and safety through partnerships with patients and consumers, ACSQHC, Sydney.

This discussion paper is available on the Commission's website: www.safetyandquality.gov.au

Foreword

This paper is our first published commentary. By

-loring patient-centred approaches undertaken

-loring patient-centred approaches undertaken

-loring patient-centred approaches and future a reduced consideration of the patient as a person.

The patient-centred movement powerfully demonstrates that fully involving the individual patient as a person at all stages with unique needs, concerns and preferences will lead to more efficacious and satisfying outcomes.

The patient-centred approach to health care treats each person respectfully as an individual human being and not as a condition to be treated. It involves not just the patient, but families, carers and other supporters. It is concerned about the patient's comfort and surroundings as well as their beliefs and values.

A patient-centred approach makes care safer and of higher quality. It provides demonstrable personal, clinical and organisational benefits. It also satisfies an ethical imperative - involving patients in their own care and in the planning and governance of the health system is the right thing to do.

The Commission supports organisations building skills and capacity to involve and engage patients and consumers both in their individual care and in system level improvements.

I commend the zeal and dedication of our staff involved in preparing this paper and the enthusiastic guidance of the contributing authors.

W J Beerworth

Chairman

Contents			Case study 1: Victorian Consumer Advisory Committees Approaches and strategies to promote patient-centred care 'Best buys' for improving patient experience Implementing rights-based patient constitutions, charters or codes Patient feedback Improved complaints processes Engaging patients in safety improvement Using information and education	
				анних
Foreword	i		Case study 1: Victorian Consumer	17
Executive summary	1	2.4	Advisory Committees Approaches and strategies to promote	18
C	_		patient-centred care	
Summary of recommendations	2		Best buys' for improving patient experience	18
Introduction	4		Implementing rights-based patient	18 19 19
	4		Patient feedback	10
1 What is patient-centred care?	6		Improved complaints processes	20
1.1 Concepts of patient-centred care	7		Engaging patients in safety improvement	20
1.2 Definitions of associated terms	8		Using information and education	21
Consumer-centred care	8		to facilitate partnerships	
Person-centred care	8		Patient and carer engagement in personal care	22
Personalised care	8		Patient and carer engagement	23
Family-centred care	8		at the governance level	
Relationship-centred care	9		User-centred design and redesign	24
1.3 Evidence for patient-centred care	9		Experience-based design and co-design	24
1.4 Patient-centred care as a dimension of high-quality health care	10		Consultation styles and communication training	24
Drivers engreeshes and initiations	40		Values training	25
2 Drivers, approaches and initiatives	12		Leadership and change management	25
for patient-centred care 2.1 International policy-level drivers	10		strategies	~~
for patient-centred care	13		Staff and practice development Staff satisfaction strategies	26 26
Collecting and publishing patient	13		Accountability strategies	26 26
experience data	ر-		Case study 2: Medical College of	
Performance reporting and choice of provide	ers 13		Georgia Health Inc, Georgia, United States	27
Financial incentives	13	2.5	An overview of leading international	27
2.2 Public value and experience as drivers	16	,	organisations	-/
for patient-centred care			World Health Organization	27
Creating public value	16		Institute for Healthcare Improvement	28
Valuing the experience	17		Picker Institute	28
Driving efficiency gains	17		Institute for Patient- and Family-Centered Care	
2.3 Consumer engagement	17		Planetree	28
and patient-centred care			Studer Group	29

	Kenneth B Schwartz Center	29	5 Making progress on patient-centred	46
	The King's Fund	29	care in Australia	
3	The relevance of patient-centred	30	5.1 Refocusing the way we look at quality in health care	47
	care to the Australian health system		5.2 Performance monitoring and measurement	48
3.1	The relevance of patient-centred care	31	5.3 Education and research	49
	to a mixed healthcare system		5.4 Organisational strategies	50
	National health reform	32	The importance of committed	50
3.3	National strategies and initiatives	32	senior leadership	
	promoting patient-centred care		Engaging patients, families and carers	51
	Australian Charter of Healthcare Rights	32	as partners	
	Australian Safety and Quality Framework for Health Care	33	Data driving change: using regular collection and feedback of patient care experience	52
	National Safety and Quality Health Service Standards	33	Supporting improvement of care delivery and environment	53
	National Primary Health Care Strategy	34	Building staff capacity and a supportive	53
	National Chronic Disease Strategy and National Service Improvement Frameworks	34	work environment Accountability at all levels for improving	54
	National Mental Health Plan	34	patient-centred care	-
	Fifth Community Pharmacy Agreement	34	An organisational culture that strongly	55
	Aboriginal and Torres Strait Islander primary healthcare policies	35	supports learning and improvement	
	F		Acronyms and abbreviations	56
4	Current jurisdictional and	36	_	
	other activities in Australia		References	57
4.1	Listening to patient feedback	37	A	6 -
4.2	The Australian Institute for Patient and Family Centred Care	38	Appendix A Assessing organisational readiness to implement	63
4.3	Examples of state and territory initiatives	38	patient-centred care	
	ACT Health	38		64
	NSW Health	39	organisational status checklist	
	NT Department of Health and Families	39		65
	Queensland Health	40	An assessment tool for hospital	
	SA Health	40	trustees, administrators, providers, and patient and family leaders	
	Tasmanian Department of Health and Human Services	40	(IPFCC 2008)	
	Victorian Department of Health	41	Appendix B Select annotated bibliography	67
	WA Health	_	ppe-iu-n 2 beloet united uter 515-108-up-iy	٠,
4.4	Case studies of current patient-centred care activities in Australia	41 42	Appendix C Supportive resources	76
	Case study 3: Calvary Health Care Simply Better Program	42	Appendix D Australian Charter of Healthcare Rights	85
	Case study 4: Flinders Medical Centre	43		
	Emergency Department Redesign Project	15	Appendix E Australian Safety and Quality	87
	Case study 5: Quality Framework	43	Framework for Healthcare	
	for Australian General Practice			
	Case study 6: Patient First Program	43		90
	Case study 7: Respecting Patient Choices Program	44	for Patient Safety	
	Case study 8: Peer service in mental health project	44	Index	92

Acknowledgments

The Australian Commission on Safety and Quality in Health Care (the Commission) wishes to acknowledge the contributing authors and staff for their work on this paper, including:

Dr Karen Luxford Dr Donella Piper Dr Nicola Dunbar Ms Naomi Poole

The Commission also wishes to acknowledge the valuable contributions made by all of the individuals and organisations that provided submissions and feedback to the draft Patient-Centred Care: Improving Quality and Safety by Focusing Care on Patients and Consumers discussion paper.

The Commission gratefully acknowledges the kind permission of Dr Karen Luxford and the Institute for Family- and Patient-Centred Care for the use of their checklists in this document.

Editing: Biotext, Canberra

Graphic design: True Characters, Melbourne

Executive summary

and communication, continuity and transition, care coordination, involvement of family and carers, and access to care.

Research demonstrates that patient-centred care improves patient care experience and creates public value for services. When health professionals, managers, patients, families and carers work in partnership, the quality and safety of health care rises, costs decrease, provider satisfaction increases and patient care experience improves. Patient-centred care can also positively affect business metrics such as finances, quality, safety, satisfaction and market share.

Australian healthcare organisations are becoming increasingly interested in patient-centred care. Most organisations can readily put patient charters and informed consent policies in place, but many find it hard to actively change the way care is delivered, and may struggle to involve patients, families, carers and consumers and to learn from their experiences.1

Patient-centred care is recognised as a dimension of high-quality health care in its own right. It is identified in the seminal Institute of Medicine report, Crossing the Quality Chasm,² as one of the six quality aims for improving care. Internationally, governments and healthcare services have used a wide variety of strategies to promote patient-centred care. A range of international organisations have also helped to develop and implement these strategies, including Planetree, the Picker Institute and the Institute for Patient- and Family-Centered Care.

I of, Key strategies for building patient-centred care into the ***oms of healthcare organisations include regular ***oms of patient feedback data; ***orers as partners; *ilding staff capacity and a supportive work environment; establishing performance accountability; and supporting a learning organisation culture.

> In Australia, a patient-centred approach is supported by the Australian Charter of Healthcare Rights, the Australian Safety and Quality Framework for Health Care, the National Safety and Quality Health Service Standards, other national policies, reports of state-based inquiries, and a range of jurisdictional and private sector initiatives.

National health reform proposals (such as the proposed Performance and Accountability Framework) provide further incentives to improve patient-centred care by linking it to performance and funding. Another driver for improving patient-centred care is the expected establishment of a National Performance Authority, which would report transparently on a range of performance indicators, including 'patient satisfaction' for Local Hospital Networks, public and private hospitals, and Medicare Locals.

Based on these strategies and frameworks—and taking into account Australia's healthcare system, with its mix of public and private sectors — system-oriented and service-oriented recommendations to foster patientcentred care are outlined in this discussion paper. These recommendations may need to be implemented in different ways, depending on the context of the healthcare organisation.

Summary of recommendations

SYSTEM-ORIENTED RECOMMENDATIONS

Recommendation 1:

Policy makers and regulators should include patientcentred care as a dimension of quality in its own right in strategic and other policy documentation.

Recommendation 2:

A core set of nationally endorsed patient survey questions should be developed to facilitate collation and comparison of patient care experience data in key healthcare settings.

Recommendation 3:

Patient surveys used to assess patient care experience need to include questions specifically addressing recognised patient-centred care domains and assess more than patient 'satisfaction'.

Recommendation 4:

'Improving patient care experience' should be included as an indicator of quality and reflected in healthcare reporting and funding models.

Recommendation 5:

To improve transparency, Australian policy makers and regulators should make data regarding patient care experience in health services publicly available via websites.

Recommendation 6:

Patient-centred care should be a component of undergraduate and postgraduate education programs for all health professionals.

Recommendation 7:

Education programs should engage patients and families as teachers and collaborators, rather than solely as cases to be studied.

Recommendation 8:

Research funding bodies should acknowledge the importance of patient-centred care to the health system, and this should be reflected in the distribution of funding.

SERVICE-ORIENTED

RECOMMENDATIONS

Health service executives and managers should:

Recommendation 9:

Ensure that organisational systems and processes are designed to be patient-centred.

Recommendation 10:

Ensure their healthcare organisation develops a shared patient-centred mission that senior leaders continually articulate to staff to promote the implementation of patient-centred care.

Recommendation 11:

Develop and implement policies and procedures for engaging patients, families and carers in their own care.

Recommendation 12:

Develop and implement policies and procedures for involving patients, families, carers and consumers at a service level, and in policy and program development, quality improvement, patient safety initiatives and healthcare design.

Recommendation 13:

Provide support for patients, families and carers involved in governance to develop the necessary skills and capacity required for effective partnerships with their healthcare organisation.

Recommendation 14:

Ensure that the service meets the National Safety and Quality Health Service 'Partnering with Consumers' Standard.

Recommendation 15:

Ensure that systems are in place for the regular collection and reporting of patient care experience data through quantitative patient surveys and qualitative, narrative-based sources.

Recommendation 16:

Ensure that organisational approaches to quality improvement include feedback about patient care experience — alongside clinical and operational data — when determining health service action plans.

Recommendation 17:

Contribute to the evidence base for patient-centred care by recording and publishing changes in key organisational and patient outcome metrics over time.

Recommendation 18:

Provide organisational support to enable staff to partner with patients and consumers, and to implement any necessary changes based on that partnership.

Recommendation 19:

Support staff through training and education activities tailored to building the capacity of all staff to deliver patient-centred care.

Recommendation 20:

Focus on work environment, work culture and satisfaction of staff as an integral strategy for improving patient-centred care. Workforce surveys and review of staff recruitment and retention rates should be undertaken at regular intervals to monitor the work environment.

Recommendation 21:

Integrate accountability for the care experience of patients into staff performance review processes.

Recommendation 22:

Foster a culture of learning within the organisation, equally learning from successes and failures, including tragic events, to promote patient-centred care.

Introduction

The health and health experiences of Australians compare well with those of other countries. Australia's life expectancy at birth remains among the highest in the world. Death rates are falling for many of our major health problems, such as cancer, cardiovascular disease, chronic obstructive pulmonary disease, asthma and injuries, and survival from these conditions continues to improve.³

Australia ranked highly in a 2010 Commonwealth Fund comparison⁴ of seven international health systems, demonstrating our healthcare professionals' commitment to high-quality care. However, this achievement should be tempered with the awareness that there is still work to be done. In 2008-09, a large proportion of the complaints made to Australian healthcare complaints commissioners were about healthcare professionals' attitude and manner.⁵ It has been suggested that these types of complaints may represent a failure to appreciate that, in some circumstances, the emotional needs of the patient may be as important as their physical state. 6 If health care is to become truly responsive to the needs and desires of the patient, then it will be necessary to refine the skills and capacity of health professionals as evidence for patient-centred care becomes clearer and our understanding of how to implement patient-centred care improves.

In 2010, Australian Health Ministers endorsed the Australian Safety and Quality Framework for Health Care. The framework identifies consumer-centred care as the first of three dimensions required for a safe and high-quality health system in Australia. Including this dimension in the framework reflects a growing recognition of the importance of placing patients and consumers at the centre of the healthcare system. In developing the framework, the Australian Commission on Safety and Quality in Health Care (the Commission) talked to patients, carers, consumers, clinicians,

managers and policy makers, and found strong support for the inclusion of consumer-centred care, as well as for an overall framework that sets the strategic direction for safety and quality in Australian health care over the next decade.

The Australian Safety and Quality Framework for Health Care contains strategies for providing care that is respectful of, and responsive to, the preferences, needs and values of patients and consumers. The Commission has developed a series of guides⁷ that support the use of the framework in practice.

Australia has a widespread and strong commitment to a healthcare system that is based on partnerships with patients and consumers. Many health services and health service providers have taken steps to embed patient-centred care principles into practice, and provide care that addresses the needs and desires of the patient, as well as their immediate treatment requirements. However, for a range of reasons, people can struggle to carry out patient-centred care in practice. It is clear that more specific and focused guidance in this area may be useful.

Patient-centred care is particularly important among vulnerable or disadvantaged populations, such as the young, elderly, disabled or mentally ill; those from culturally and linguistically diverse backgrounds, or rural and remote areas; and Aboriginal and Torres Strait Islander peoples. For many of these people, communication and collaboration with health professionals can be difficult and necessarily involves carers, friends, family, spiritual and pastoral advisers, or the broader community. Patient-centred care principles and approaches are considered to be an opportunity to go some way towards addressing the inequity people in these populations can experience. It is also seen as a way of supporting greater involvement of all people in healthcare processes, and potentially achieving better health outcomes.

Internationally, there has been a focus on patient-centred care for some time, particularly in the US and England. However, the health systems in these countries are different from the Australian system, and questions arise as to whether the lessons learned and approaches that are used internationally can be applied here.

To address these needs, the Commission prepared a draft discussion paper as the first review of patient-centred care for Australia, and in 2010 undertook an extensive open consultation process on the paper. Following this consultation, the Commission refined the discussion paper to clarify key issues raised by stakeholders with the intention of providing a comprehensive resource for healthcare organisations wishing to become more patient-centred.

This discussion paper — Patient-centred care: Improving quality and safety through partnerships with patients and consumers — provides a broad overview of how to improve the delivery of health care by taking a more patient-centred approach. The target audience includes healthcare professionals, managers and policy makers who want to re-orient their systems and practices to be more patient-centred, and to partner with patients, consumers, families and carers. This paper is the first of a proposed suite of activities by the Commission on patient-centred care. Future work in this area may focus on more specific settings and include resources that target consumers.

This paper provides background information, resources and examples of how to take a more patient-centred approach to healthcare delivery. Also included is an in-depth discussion of the concepts and evidence regarding patient-centred care; a comprehensive review of international approaches and activities; and an examination of some of the national, jurisdictional and other policies and activities in Australia that support patient-centred care. Australian examples and case studies have been used to illustrate many of the concepts and approaches described in this paper.

Within this paper, the term 'patient-centred care' is used as it is the most commonly used term in research and literature in this area. The Australian Safety and Quality Framework for Health Care used the term 'patient-focused care' when it was released for consultation in 2009; after consultation, the term was changed to 'consumer-centred care'. Other common terms include person-centred care, personalised care, family-centred care and relationship-centred care.

All of these terms are conceptually similar. They emphasise the formation of partnerships and the central roles of not only the patient, but also the family, carers, friends, spiritual and pastoral advisers, community members and staff in individual aspects of care. The importance of acknowledging those who are significant to the patient has been recently enshrined

in Commonwealth legislation in Australia in the Carer Recognition Act 2010.8

As will be seen in this paper, patient-centred care includes a wide range of concepts and ideas about the way in which patients and consumers are involved in the health system and the delivery of health care. The concept of patient-centred care covers partnerships between a patient and a health professional when care is provided. It also covers partnerships between patients, families, carers, consumers and citizens in health service policy, planning and governance. There are large bodies of literature about many of the specific aspects of patient-centred care, such as communication between patients and doctors during a consultation, or strategies to engage consumers in health service planning. This paper does not go into detail about these specific aspects, but instead provides a broad overview of strategies and approaches that health services can use to make their services more patient-centred and to integrate patientcentred care into their business.

Section 1 of this paper discusses the concepts, origins and evidence about patient-centred care. Section 2 examines the drivers, approaches and strategies to patient-centred care, including key international organisations and initiatives. Section 3 examines the relevance of patient-centred care to the Australian healthcare system, and Section 4 highlights current patient-centred activity in Australia. In light of international and current Australian approaches, Section 5 discusses the way forward for patient-centred care in Australia, and provides recommendations for actions that could be taken to make the health system and healthcare organisations more patient-centred. The appendices include checklists to help organisations assess their readiness to implement patient-centred care, an annotated bibliography highlighting key references on the evidence for patient-centred care and a list of patient-centred care resources.

What is patientcentred care?

This section provides an overview of the concept of patient-centred care as a dimension of high-quality health care, including commonly used terms and their definitions, and a summary of the evidence base for this approach.



1.1 **CONCEPTS OF PATIENT-CENTRED CARE**

Numerous proposed definitions of patient-centred care encompass similar core concepts, but there is no globally accepted definition.10 Modern concepts of patientcentred care are based largely on research conducted in 1993 by the Picker Institute in conjunction with the Harvard School of Medicine. The research identified eight dimensions of patient-centred care that were originally documented in Through the Patient's Eyes: Understanding and Promoting Patient-Centered Care.11 The dimensions are:

- » respect for patients' preferences and values
- » emotional support
- » physical comfort
- » information, communication and education
- » continuity and transition
- » coordination of care
- » the involvement of family and friends
- » access to care.

The framework clearly defined the patient's perspective for the first time and served as the foundation for the Picker surveys measuring patient experiences of health care.12

The concept of patient-centred care clearly recognises the need to include not only the patient in their care, but significant others, including family, friends, carers, spiritual and pastoral advisers, and broader community members.

Patient-centred care also focuses on staff. To succeed, a patient-centred approach should also address the staff experience, because the staff's ability and inclination to care effectively for patients is compromised if they do not feel cared for themselves.13

Organisation-specific concepts of patient-centred care have also emerged. Some organisations identify individual elements of patient-centred care as part of an overall patient-centred care framework. An overview of leading organisations promoting strategies for patientcentred care is presented in Section 2.5.

The World Health Organization (WHO) uses the term 'responsiveness' in preference to 'patient-centred care'. Responsiveness describes how a healthcare system meets people's expectations regarding respect for people and their wishes, communication between health workers and patients, and waiting times.14 WHO states that recognising responsiveness as an intrinsic goal of health systems reinforces the fact that health systems are there to serve people.

Several studies reviewing patient-centred care in the US 15 (cited in Goodrich and Cornwell 16) have identified the core elements as:

- » education and shared knowledge
- » involvement of family and friends
- » collaboration and team management
- » sensitivity to nonmedical and spiritual dimensions of care
- » respect for patient needs and preferences
- » the free flow and accessibility of information.

In reviewing definitions of patient-centred care more generally, Cronin¹⁷ (cited in International Alliance of Patients' Organizations¹⁰) identified the same core elements. Similarly, Robb and Seddon¹⁸ identified the following common concepts in definitions of patientcentred care:

- » informing and involving patients
- » eliciting and respecting patient preferences
- » engaging patients in the care process
- » treating patients with dignity
- » designing care processes to suit patient needs, not providers
- » ready access to health information
- » continuity of care.

According to the International Alliance of Patients' Organizations, the most common element in definitions of patient-centred care is respect for the needs, wants, preferences and values of patients.10 While some definitions acknowledge patients' rights and responsibilities (such as the New Zealand Ministry for Health's definition referred to later in this section), most do not, and safety is rarely included in these definitions or frameworks.10

In 2000, a five-day seminar was held in Salzburg, Austria, where 64 people from 29 countries examined what health care could become in a utopian land called PeoplePower. They envisaged informed and shared decision making, mutual commitments to quality and health outcomes, and patient partnership in governance.19 The phrase 'nothing about me without me' was their guiding principle. This phrase has since been popularised by authors and regulators, and is considered synonymous with efforts to advance a vision for patientcentred care.20

Carer experience of patient-centred care
...patient-centred is an attitude that is
innate in most practitioners—particularly
in those that give direct and personal
service; [it is] an empathy with the patient
and [an ability] to stand in the shoes of the
patient, particularly those who, because
of old age or disability, cannot answer
back. [It is] an inherent attitude that can
instinctively consider how another person
will react when addressed in words or by
some deed. An attitude which accepts that
the patient has a mind as well as a body.

Submission received in the consultation on the
draft patient-centred care discussion paper

DEFINITIONS OF ASSOCIATED TERMS

Several terms used interchangeably or associated with 'patient-centred care' are described below. All terms are based around the fundamental concepts of partnership and collaboration between healthcare professionals and patients, families, carers and consumers.

Consumer-centred care

The term 'consumer-centred care' is sometimes preferred to 'patient-centred care' to acknowledge that care should focus on people who are actual or potential users of healthcare services. For some, the term 'patient' has passive overtones. In contrast, the term 'consumer' is seen as a more active term, encompassing the need to engage people as partners in health service delivery.21 The term 'consumer' also aligns with 'client' and 'user' in business and management models of service delivery.22

Person-centred care

Terms such as 'person-centred care' and 'personcentredness' are often used interchangeably in primary care settings with 'patient-centred care'. The term 'person-centred care' appears more frequently in literature on the care of older people, 23 and focuses on developing relationships and plans of care collaboratively between staff and patients.23-24 This term values the needs of patients, carers and staff, with emphasis on the reciprocal nature of all relationships.25-26

Personalised care

'Personalised care' is the integrated practice of medicine and patient care based on one's unique biology, behaviour and environment. 'Personalised care' uses genomics and other molecular-level techniques in clinical care, as well as health information technology, to integrate clinical care with the individualised treatment of patients.27

Family-centred care

The term 'family-centred care' emerged in the US in the 1980s in response to the needs of families with children who could not leave hospital. These families sought to work more collaboratively with healthcare professionals and successfully advocated for changes to enable them to care for their children in home and community settings.²⁸ More generally, children's hospitals in the US adopted the concept of 'family-centred care' in recognition of input from parents and family members to improve the care of patients who were too young to tell physicians and nurses how they felt. 29-30 In England,

'family-centred care' also relates to children's health care, and encompasses the concepts of parental participation, partnership and collaboration between the healthcare team and parents in decision-making, family-friendly environments that normalise family functioning within the healthcare setting as much as possible, and care of other family members.³⁰

Relationship-centred care

In 'relationship-centred care', all participants appreciate the importance of their relationships with one another. 'Relationship-centred care' is founded upon four principles:

- relationships in health care ought to include the personhood of the participants
- 2. affect and emotion are important components of these relationships
- 3. all healthcare relationships occur in the context of reciprocal influence
- 4. the formation and maintenance of genuine relationships in health care is morally valuable.

In 'relationship-centred care', relationships between patients and clinicians remain central, although the relationships of clinicians with themselves, each other and the community are also emphasised.^{31–32}

1.3

EVIDENCE FOR PATIENT-CENTRED CARE

Patient-centred care is a broad concept, covering a wide range of different strategies and approaches. The evidence base for patient-centred care is similarly varied: some aspects of patient-centred care have been the subject of large bodies of work, while others are still emerging. Overall, however, it is clear that patient-centred care has significant benefits associated with clinical quality and outcomes, the experience of care, the business and operations of delivering health services, and the work environment.

This section provides a summary of some of the research evidence that supports patient-centred care. Additional information about most of the papers referenced here is included in Appendix B.

There is an increasing body of research about the association between patient experience and perceptions of care, and overall measures of quality and safety.^{33–36} Two studies have examined associations among information about the experience of patients collected through the Hospital Consumer Assessment of Healthcare Providers and Systems (H-CAHPS) survey (see Section 2.1), process measures of quality^{33–34} and patient safety indicators³³ across a large number of hospitals in the US. Positive associations were found between patient experience and overall technical quality and safety.^{33–34}

When looking specifically at patients who have had an acute myocardial infarction (AMI), a number of studies have shown an association between patient-centred care and clinical outcomes, including mortality.^{37–40} One study found that better patient-centred care during admission for AMI was associated with a decreased risk of death one year after discharge.⁴⁰

Other clinical benefits that have been found to be associated with better patient experience and patientcentred care include decreased readmission rates, 41 decreased rates of healthcare acquired infections, 42-44 improved delivery of preventive care services, 45 reduced length of stay, 43, 46 improved adherence to treatment regimens⁴⁷ and improved functional status.⁴²⁻⁴³ One study found that factors such as better communication with doctors, a clean and quiet hospital environment and the responsiveness of medical staff were significantly associated with a decreased likelihood of pressure ulcers and post-operative infections.33 In addition, participation of hospital patients in their own care, such as asking questions or participating in decision making, has been found to reduce the risk of experiencing an adverse event.48

A number of studies have looked at the associations between patient-centred care and outcomes in general practice and primary health care. These studies have often focused on communication and collaboration between patients and providers. In these settings, patient-centred care has been associated with a reduction in the number of diagnostic test orders and other referrals, ^{49–51} better adherence to treatment regimens, ^{52–53} greater patient satisfaction ^{50, 52} and greater patient enablement. ⁵⁰

Evidence about the impact of patient-centred care on the business and operation of health services is also beginning to emerge. Some studies have examined the broad effects of the use of patient-centred strategies and approaches within healthcare organisations. Stone⁴⁶ examined data for inpatient units at two similar hospitals in the US over five years. One hospital introduced an extensive program of patient-centred practices, and the other continued their usual practices. The patient-centred inpatient unit consistently demonstrated a shorter average length of stay, a statistically significant lower cost per case, a shift in emphasis from the use of higher-cost staff to lowercost staff, and higher-than-average overall patient satisfaction scores.

In the US, hospitals that have introduced patient-centred approaches have found that inpatient and outpatient throughput increased, community perceptions improved, and the proportion of people reporting that they would be likely to recommend the hospital or to return to it for future care increased.⁵⁴ There have also been reports of hospitals experiencing considerable reductions in malpractice claims following

the implementation of patient-centred care practices, despite an increase in patient care activity.⁵⁴ Sweeney et al⁵⁵ examined the introduction of patient-centred management practices for patients with multiple conditions compared with usual care. A significant decrease in costs was found for the patient-centred group compared with the usual care group, without any shortening of life.

The Medical College of Georgia (MCG) Health System in Augusta, Georgia, was one of the first large health services to explicitly re-orient their systems to be more patient-centred. ⁵⁶ MCG has found that this patient-centred approach positively affects each of the MCG's business metrics (finances, quality, safety, satisfaction and market share). Outcomes that have been identified by MCG include: ⁵⁷

- » patient satisfaction scores for the children's hospital among the highest measured by Press Ganey
- » patient satisfaction scores for the breast health centre increasing from the 40th percentile to the 90th percentile
- » improvements across a range of measures for a new adult neuroscience unit, including increasing the volume of discharges by 15.5 per cent, decreasing length of stay by half, reducing medication errors by 62 per cent and reducing the staff vacancy rate from 7.5 per cent to zero (with a waiting list for nursing positions on the unit).

Increasing patient satisfaction through patient-centred approaches also increases employee satisfaction and this, in turn, improves employee retention rates and the ability to continue practising patient-centred care. ⁵⁴ According to Charmel and Frampton, ⁵⁴ the link between patient satisfaction and employee satisfaction is reflected in the fundamental philosophy of patient-centred care: the importance of staff feeling cared for themselves, so they can best care for their patients.

1.4

PATIENT-CENTRED CARE AS A DIMENSION OF HIGH-QUALITY HEALTH CARE

Over the past two decades, patient-centred care has become internationally recognised as a dimension of the broader concept of high-quality health care. In 2001, the US Institute of Medicine's (IOM) Crossing the Quality Chasm: A New Health System for the 21st Century defined good-quality care as safe, effective, patient-centred, timely, efficient and equitable.²

The report defined patient-centred care as 'care that is respectful of and responsive to individual patient preferences, needs and values, and ensuring that patient's values guide all clinical decisions'. It set out several rules to redesign and improve patient-centred care, including maintaining transparency, sharing knowledge and information freely, customising care based on patients' needs and values, and ensuring that the patient is the source of control and that care is based on continuous, healing relationships.

The IOM report outlined four levels that further define quality care and the role of patient-centred care in each level:

- The experience level refers to an individual patient's experience of their care. At this level, care should be provided in a manner that is respectful, assures the candid sharing of useful information in an ongoing manner, and supports and encourages the participation of patients and families.
- 2. The clinical micro-system level refers to the service, department or program level of care. At this level, patients and family advisers should participate in the overall design of the service, department or program. This could be as full members of quality improvement and redesign teams, and participating in planning, implementing and evaluating change.
- 3. The organisational level refers to the organisation as a whole. The organisational level overlaps with the clinical micro-system level in that the organisation is made up of various services, departments and programs. At this level, patients, families, carers and consumers should participate as full members of key organisational committees for subjects such as patient safety, facility design, quality improvement, patient or family education, ethics and research.
- 4. The **environment level** refers to the regulatory level of the health system. Here, the perspectives of patients, families, carers and consumers can inform local, state, federal and international agency policy and program development. These agencies, along with accrediting and licensing bodies, are in a position to set expectations and develop reimbursement incentives to encourage and support the engagement of patients and families in healthcare decision making at all levels.

According to Charmel and Frampton,⁵⁴ the IOM report reinforces patient-centred care not only as a way of creating a more appealing patient experience, but also as a fundamental practice for providing high-quality care in the US.

Like its evolution in the US, the concept of patient-centred care in England became a dimension of high-quality health care over time. From 1997, when the government began a 10-year program to reform the National Health Service (NHS), strategy documents and policies increasingly referred to the ambition to create a 'patient-centred NHS'.²⁴

By 2004, the NHS had set out its priorities for the following four years in its NHS Improvement Plan: Putting People at the Heart of Public Services. 24 The aim was 'to ensure that a drive for responsive, convenient and personalised services takes root across the whole of the NHS and for all patients' (cited in Goodrich 24). Creating a Patient-led NHS: Delivering the NHS Improvement Plan 58 outlined how the quality agenda could be delivered: patient-led services would ensure that patients were treated with respect, dignity and compassion. 24

The 2008 NHS Next Stage Review⁵⁹ aimed to improve patients' overall experience by putting 'quality at the heart of what we do'. The review committed to providing 'safe, personalised, clinically effective care' and 'locally led, patient-centred and clinically driven change' (cited in Goodrich²⁴). In his introduction to the Next Stage Review, Lord Darzi stated that 'high-quality care should be as safe and effective as possible, with patients treated with compassion, dignity and respect. As well as clinical quality and safety, quality means care that is personal to each individual'.⁵⁹

In 2010, The King's Fund⁶⁰ assessed the effect of the former Labour government's reforms to the NHS on various factors within the service (safety, equity, accessibility, accountability, efficiency, clinical effectiveness, patient experience, promoting health and managing long-term conditions). The study was based on the best available evidence and measured improvements from patients' perspectives. The authors found that, overall, the NHS had improved the standards for high-quality, safe health care. Waiting times for hospital care had been reduced, and access to primary care had improved. There had also been progress in making the NHS more accountable and transparent to government and taxpayers. However, there was still some way to go to firmly embed a strong safety culture in some parts of the service.

The effectiveness of efforts to improve patients' experiences of the NHS presented a mixed picture in this study. Overall, patient experience scores were high and public satisfaction, as a whole, had been improving steadily since 2002. Patient experience had been built into regulation and many providers were routinely monitoring survey results. However, the surveys revealed several weaknesses, including limited progress in delivering greater choice of treatments, information

about care, involvement with decisions, and problems with aspects of privacy for inpatients, especially mental health inpatients. To address this issue, The King's Fund recommended that organisations find more effective ways of encouraging and acting on feedback from patients, carers and staff, ensuring that patients' experiences have a real impact on the quality of care. 60

In New Zealand, the Code of Health and Disability Services Consumers' Rights Regulation 1996⁶¹ (the Health and Disability Code) and the New Zealand Ministry of Health's *Improving Quality* document⁶² established patient-centred care as a priority. 18 Improving Quality talks about 'people-centred' rather than 'patient-centred' care and defines this concept as '...the extent to which a service involves people, including consumers, their families and is receptive to their needs and values'. It includes participation, appropriateness, adherence to the Health and Disability Code, and adherence to other consumer protections such as the Health Information Privacy Code 1994. 18, 62 According to Robb and Seddon, 18 by referencing the Health and Disability Code, this document acknowledges patients' rights as integral to patient-centred care, a common omission in other definitions. In addition, the New Zealand Ministry of Health definition, like its IOM counterpart, implies extending patient-centred care beyond the patientpractitioner interaction. This reflects the importance of collaboration between patients, families, consumers, healthcare practitioners and hospital leaders in all aspects of health care and at all levels of the healthcare system.18

At the federal level in Canada, the term 'collaborative patient-centred care' is used to describe Health Canada's initiatives in the primary healthcare sector to increase patients' self-care⁶³ and educate professionals about working together to achieve patient-centred care.⁶⁴ At the state and provincial level, many jurisdictions are adopting a patient-centred approach as part of their aim to increase healthcare quality.⁶⁵⁻⁶⁶

Overall, many regulators and organisations recognise patient-centred care as a key dimension of quality. Of the Organisation for Economic Co-operation and Development (OECD) member states, the US, England, Canada and Australia include patient-centred care, patient focus or responsiveness as a dimension of healthcare quality in national documents and frameworks. The OECD and WHO included these concepts as a dimension of healthcare performance. ⁶⁷ One study found that in addition to the US, England, Canada and Australia, several European countries (Germany, France, Denmark, the Netherlands and Switzerland) had all implemented various performance measures that include patient-centred health care. ⁶⁸

Drivers, approaches and initiatives for patient-centred care

This section presents an overview of the policies and approaches that drive patient-centred care. It also examines the strategies and initiatives that have been used successfully by healthcare regulators and organisations to put patient-centred care into operation at different levels and in different sectors of the healthcare system.

There is wide agreement about the need to place patients at the centre of their own care, and at the centre of the health system more generally. However, healthcare organisations often have difficulty implementing the type of change necessary to sustain patient-centred care. As a consequence, many organisations are seeking guidance on appropriate strategies for building and maintaining partnerships with patients, families, carers and consumers.

This section provides information on key policy, legislative and theoretical approaches that have driven the emergence of patient-centred care, followed by specific strategies that have been used to promote and embed patient-centred care in health services and health system policy. The section concludes with a summary of leading international organisations that provide tools and resources to support patient-centred care. Case studies of patient-centred approaches that have been adopted by different organisations, including some Australian examples, are provided to illustrate these strategies.

A list of the leading organisations' websites and other relevant websites and resources are included in Appendix C.

2.1

INTERNATIONAL POLICY-LEVEL DRIVERS FOR PATIENT-CENTRED CARE

Internationally, there has been a focus on patientcentred care for some time, particularly in the US and England. These countries have three main policy-level drivers for patient-centred care:

- » mandatory government requirements for service providers to collect and publish patient experience data (other jurisdictions, such as Canada and New Zealand, also collect patient experience data, although there is no mandatory requirement to do so)
- » publicly available information that enables consumers to choose between service providers
- » financial incentives for providers who achieve high measures of patient-centredness.

The different approaches of these policies are discussed below.

Collecting and publishing patient experience data

The US, England and some European countries have implemented patient survey programs to systematically collect patient and carer experience feedback at a national level. In general, the purpose of these programs is to provide a comprehensive picture of quality, monitor trends and drive system change.

England

In England, the former Labour government's vision was for patients and the public to drive the design and delivery of high-quality services. In health care, this was to be achieved by 'putting patient experience centre stage', and a number of regulatory requirements were put in place to meet this aim.^{59,69}

The coalition government has continued this commitment to patient experience. The government's National Health Service (NHS) White Paper, *Equity and Excellence: Liberating the NHS*, published in July 2010,⁷⁰ sets out three key objectives for health care in England: to create a patient-led NHS, to improve healthcare outcomes, and to increase autonomy and accountability within the NHS.

To achieve these objectives, the *NHS Outcomes* Framework 2011/12⁷¹ sets out 10 overarching indicators and 31 'improvement area' indicators, divided between five domains of health care. The first three domains make up the 'clinical effectiveness' component of quality, Domain 4 ensures that people have a positive experience of care and Domain 5 is 'patient safety'.

While some indicators are yet to be developed, most of the indicators under Domain 4 are based upon patient experience surveys, including the General Practice Patient Survey, the Inpatient Survey, the Maternity Survey, and the Outpatient Survey, and the Accident and Emergency Survey.

These surveys are conducted using standardised questionnaires. All of the hospital-based surveys use methods based on the work of Picker Institute Europe, which develops, coordinates and implements healthcare surveys based on the eight Picker dimensions of patient-centred care (see Section 1.1).

Information and some findings from a range of NHS surveys can be found on the Care Quality Commission's website and the Survey Coordination Centre website (see Appendix C). Various disease-specific patient experience surveys have also been conducted in England, including the National Cancer Patient Experience Survey Programme.

Patient experience was also a standard measure in the 2010/11 Quality and Outcomes Framework, and will be a feature of Quality Accounts 2011/12 and local Commissioning for Quality and Innovation (CQUIN) schemes. In late 2010, the National Institute for Health and Clinical Excellence commenced the development of a quality standard for patient experience, which is due to be published in October 2011.

The government's overall objectives in setting these outcomes are to reduce mortality and morbidity, increase safety, and improve patient experience and outcomes for all. The Operating Framework for the NHS in England 2011/12⁷² situates patient experience as 'a key arbiter of all NHS services.' The framework states that the 'core purpose remains the delivery of improved quality for our patients, by improving safety, effectiveness and patient experience'.

The framework requires providers to ensure that appropriate systems are in place to capture the views and experiences of patients, service users and carers. This includes the use of locally and nationally coordinated patient surveys, and a range of additional approaches or sources, such as the use of real-time feedback collected at the point of care (e.g. text messaging, patient experience trackers), and use of complaints data and patient-reported outcome measures. In addition, the framework requires service providers to raise awareness of local feedback options, encourage feedback, and demonstrate to the public how their feedback has been used to improve service quality and patient experience. Furthermore, providers need to ensure that patient experience and feedback are inherent parts of service design, delivery and improvement.

United States

In the US, the Centers for Medicare and Medicaid Services (CMS) and the Agency for Healthcare Research and Quality (AHRQ) routinely collect data on patient experience from service providers. This has been expanded under the *Patient Protection and Affordable Care Act 2010.*⁷³

The US has implemented the Consumer Assessment of Healthcare Providers and Systems survey (CAHPS), developed at Harvard University and based on the Picker principles of patient-centred care. The CAHPS program develops and supports a comprehensive and evolving family of standardised surveys that ask patients to report on and evaluate their healthcare experiences.

Hospitals use the hospital-CAHPS (H-CAHPS) survey to assess seven domains: communication with doctors, communication with nurses, responsiveness of hospital staff, communication about medicines, pain control, cleanliness and quiet of physical environment, and discharge information. The survey provides an overall rating and assesses the likelihood or willingness of patients to recommend the service to others. Other versions of the CAHPS survey are used by clinician group practices and health insurance plans. A shorter, real-time version has also been developed.

The National CAHPS Benchmarking Database (also referred to as the CAHPS Database) is the national repository for data from CAHPS surveys. It holds 11 years of data and is one of the key resources for the AHRQ's National Healthcare Quality Report and National Healthcare Disparities Report. These annual documents help policy makers monitor the nation's progress toward improved healthcare quality.

Various patient experience surveys that focus on specific diseases are also emerging in the US. For example, the National Cancer Institute and the AHRQ are developing a patient care experience survey for cancer patients.

Performance reporting and choice of providers

Both England and the US have public web-based systems that provide information on health service providers' performance, to allow consumers to choose which provider they wish to access. A brief overview of these systems is presented below.

England

Since April 2008, all patients who need a specialist referral have had the right to choose to go to any hospital, including many private and independent sector hospitals that are contracted to provide NHS-funded care. The NHS Choices website publishes information on the specialist options available and performance indicators from the national survey data for each hospital. It also allows patients to rate and comment on their healthcare providers online. Regular surveys of patients' experience of choice provide information about the availability and uptake of provider choice.

The website enables consumers to find both primary and acute care services in their area. It also allows consumers to compare hospitals by various criteria, including cleanliness, car parking arrangements,

general facilities, availability of Accident & Emergency departments, experience of a particular procedure and, in some cases, survival rates.

In addition, the NHS Choices website has a number of areas to which consumers can contribute. Consumers can comment on almost any NHS service, including hospitals, general practices and dentists. Comments will eventually become part of an overall 'ranking' showing the public's opinion of every hospital and general practice that does NHS work in England.

The Coalition government's plan for the NHS sees patient choice as a key mechanism to create a high-quality, patient-centred NHS. The proposed arrangements under the *Health and Social Care Bill* 2010–11 would extend patient choice to include choice of specialist team, general practice and treatment.⁷⁵

United States

The US CMS has implemented a web-based tool called Hospital Compare, which uses data from the Hospital Outcomes of Care Measures, CAHPS and H-CAHPS surveys. This tool sets out information about how well hospitals care for patients with certain medical conditions or surgical procedures.

CMS also launched the Physician Compare web-based tool in December 2010, under the *Patient Protection and Affordable Care Act 2010.* This tool serves as a healthcare professional directory on the official US Government Medicare website, and currently allows individuals to search for a physician or other healthcare professional by specialty, type of professional and location. Additional search criteria allow the user to search by gender, and whether or not the healthcare professional accepts the Medicare-approved amount as payment in full on all claims. Other information available includes languages spoken, group practice locations, education and hospital affiliation. Information on the site is updated monthly.

The Physician Compare web-based tool also includes information about physicians and other professionals who satisfactorily participated in the Physician Quality Reporting System (formerly known as the Physician Quality Reporting Initiative). The tool does not yet contain physician and eligible professional performance information. CMS is required to make information on physician performance publicly available through Physician Compare by January 2013.⁷⁶

Financial incentives

Public and private healthcare sectors around the world are now linking service quality with provider payment. Patient care experience is a key outcome metric for quality reporting, and financial incentives can now help drive services towards patient-centred care. Both England and the US provide financial incentives to some healthcare providers for adopting improved quality

practices, including clinical outcomes and some patient-centred care principles. A recent study of CAHPS data for the primary care sector found that the use of public reporting and performance-based financial incentives can potentially improve patients' experiences of care.⁷⁷

This section focuses on the role of 'pay for performance' (P4P) in driving organisations to focus on patient experience. P4P is defined as 'financial incentives that reward providers for the achievement of a range of payer objectives, including delivery efficiencies, submission of data and measures to payer, and improved quality and patient safety'.⁷⁸

England

In 2004, the NHS began a major P4P initiative—the *Quality and Outcomes Framework*. Under the 2010/11 framework, general practitioners (GPs) received income increases based on their performance. This was measured using 134 quality indicators that cover patient experience, organisation of care and clinical care for ten chronic diseases.

Early analysis of the *Quality and Outcomes Framework*⁷⁹ showed that overall, family practices achieved high scores in their performance indicators in the first year. Practitioners included in the study earned an average of AU\$40 000 more by collecting nearly 97 per cent of the points available. However, a small number of practices achieved high scores by excluding large numbers of patients through exception reporting. More research is needed to determine whether practices such as these exclude patients for sound clinical reasons, or to increase their income.⁷⁹

It is unclear whether the high-performance scores reflect improvements that were already underway, or whether improvement was accelerated by the introduction of the scheme. Overall, the *Quality and Outcomes Framework* has seen modest improvements in measured quality of care, but has had little effect on inequalities in chronic disease management.

The NHS introduced a new CQUIN framework in April 2009 to improve the quality of care in hospitals and other healthcare organisations. The CQUIN makes a proportion of providers' income (1.5 per cent of contract value in 2011/12) conditional on an agreed package of goals and indicators that address quality, safety, innovation and patient experience. To earn from the CQUIN, providers of acute, ambulance, community, mental health and learning disability services that use national contracts must agree on a CQUIN scheme with their funding body.

The 2011/12 CQUIN schemes must include locally agreed quality improvement and innovation goals. In addition, acute providers must include two national goals in 2011/12: to improve responsiveness to personal needs of patients, and to reduce avoidable death, disability and chronic ill health from venous thromboembolism (VTE)

by engaging patients and carers in understanding the risk of VTE and how this can be reduced. 83 From 2011/12, providers that fail to meet agreed patient satisfaction goals may have a proportion of contract payment withheld (up to 10 per cent over time).

From April 2012, the NHS Commissioning Board will assume responsibility for leading the use of the CQUIN framework. The Board's work will be informed by an independent evaluation report, expected in March 2012.

United States

The US provides financial incentives to physicians and hospitals that provide data on quality measures, including patient experience, to the CMS.

In 2008, the CMS introduced the Reporting Hospital Quality Data for Annual Payment Update Initiative to strengthen the relationship between payment and quality of service. Under this scheme, hospitals publicly report a range of inpatient data—including quality, mortality and H-CAHPS patient care experience items—via Hospital Compare. In April 2010, the CMS announced that acute care hospitals that choose not to participate in the voluntary reporting program, or do not meet the established reporting deadlines, will have a 2 per cent reduction in reimbursement for their Medicare patients (those aged 65 and over or who meet other special criteria).

For individual physicians, the *Tax Relief and Health Care Act 2006*⁸⁴ established a physician quality reporting system. This includes an incentive payment for eligible physicians who report on quality measures for services covered by Medicare. Payments are also available for better care coordination between home, hospital and offices for patients with chronic illnesses. Ambulatory surgical centres and other health organisations must also comply with Medicare P4P reporting and performance targets under the *Patient Protection and Affordable Care Act 2010*.73

Many private providers and purchasers have embraced P4P approaches as an essential way to meet quality improvement goals. For example, in 2009, a large private insurer—Blue Cross Blue Shield Massachusetts—introduced an 'alternative quality contract' with healthcare providers. Participating providers receive performance-based incentives for a broad set of quality and outcome measures for outpatient and inpatient care. Organisations that meet defined targets can earn up to an additional 10 per cent of their total budget. The same performance incentives and measures are used for every provider, including nationally accepted measures of clinical processes, clinical outcomes and patient experiences.

2.2

PUBLIC VALUE AND EXPERIENCE AS DRIVERS FOR PATIENT-CENTRED CARE

There is strong support for viewing patient-centred care as a core component in ensuring appropriate outcomes from health service delivery. A range of emerging views regarding the need to create public value, focus on the experience of care, and improve efficiency as part of the context for improving patient-centred care are described below.

Creating public value

One of the common themes resulting from high-profile public inquiries into the health systems in Australia (see Section 4) and England was a reported loss of trust in administrators and clinical colleagues from patients and the community. ⁸⁵ To regain public trust and confidence, public sector organisations are adopting a 'public value' approach to service delivery.

Public value is the return created by public services for the taxes that people pay. The expected return is more than simply 'value for money'. Consumers expect to be able to trust those delivering the services, the actual service delivered, the way they are included in the process of creating and delivering the service, and the measurement of any process or product of delivery from the service. ⁸⁶ The power of a public value approach lies in its advocacy of a greater role for consumers in decision making. It also drives public managers to seek out what consumers want and need, creating a quality service. ⁸⁷

Research conducted by the Work Foundation in England, and sponsored by the NHS Institute for Innovation and Improvement and the Quality Improvement Agency, cites a range of methods and activities that enable public services to involve consumers and create public value. 88 These are in line with the patient-centred activities outlined in Section 2.4, and include formal mechanisms such as client surveys, facilitating patient engagement on committees and boards, providing information to consumers, and improving communication and interactions with consumers.

In England, the NHS collaborates with people who use their services to create public value. NHS organisations are required to regularly and systematically collect and analyse feedback from service users to inform decisions about service commission and delivery. ⁶⁹ In addition, public services are empowering individual citizens to shape their services, by establishing local networks that support the involvement of consumers and patients in commissioning, providing and scrutinising local care services. The Local Government and Public Involvement in Health Act 2007⁸⁹ and the proposed Health and Social Care Bill 2011⁹⁰ empower local networks to canvass the views

of patients and citizens on their need for and experience of care services, and to make recommendations to health services based on those experiences. Furthermore, overview and scrutiny committees question whether services meet the needs of local people, and whether the experience of patients is leading to improvements in service delivery.

Valuing the experience

Another approach to seeking out what consumers want and need comes from industry models of business economy theories. In 1999, Harvard Business School published Pine and Gilmore's The Experience Economy.91 This book proposed that society was moving away from the industrial economy view to a modern view, where the 'product' that a customer receives in a service economy is 'the experience'. In this new economy, organisations create engaging, personal and memorable experiences for customers. The application of this approach to the healthcare sector leads to a greater focus on patient care experience as an outcome of the 'service' received. Such approaches in health care are typified by work such as If Disney Ran Your Hospital.92 Based on the idea that Disney does not provide a service, but instead provides an 'experience', this work emphasises that 'hospitals would do well to emulate the most vital things that earn Disney the love of their guests and employees'.92

Driving efficiency gains

The need for improved efficiency in service delivery is also driving the engagement of patients and consumers more generally. Limited resources in the form of underfunding, low staffing levels and low morale in already overstretched systems are a perceived barrier to the practice of patient-centred care. However, recent research indicates that a patient-centred approach can make health service delivery more efficient. Hy harnessing consumers' contributions, insights and collaboration, patient-centred care is becoming a critical strategy for public services to deal with rising service costs. 193-194

2.3

CONSUMER ENGAGEMENT AND PATIENT-CENTRED CARE

The importance of involving people in all levels of the healthcare system has been recognised for some time, particularly since the *Declaration of Alma Ata* in 1978. This declaration states that 'people have the right and duty to participate individually and collectively in the planning and implementation of their health care'.⁹⁵

Consumer engagement is a broad and complex field, with different types and levels of engagement that reflect

the different goals, audiences and purposes for seeking engagement. ⁹⁶ Different types of consumer engagement range from processes to inform or disseminate information, which have a low level of engagement, to formal partnerships with a high level of public involvement and influence. ⁹⁷ In addition to engagement as patients at the individual service provision level, people are often engaged as consumers of health care, or as citizens, to represent the views of patients and the broader community at the service, program, policy and governance levels of organisations.

All of us are patients or potential patients, and citizens or members of the wider community. However, a distinction can usefully be made between what we want when we are using the health service and what we hope for as citizens or taxpayers. 98 Involving consumers as patients means tackling the clinical agenda and changing the culture of care, while involving consumers as citizens means opening up debate about the pattern and nature of service provision. 99

Involving consumers as citizens has been described as:97

- » an ethical and democratic right
- » a way of adding legitimacy to decision-making
- » a way of improving health outcomes and policy outcomes
- » a way of improving relationships with consumers
- » a way of serving political or democratic processes.

There are clear differences between the processes that empower individuals to contribute to improving the safety and quality of health care, and the processes that enable the public to hold the health system to account. However, without engaging patients in making decisions about their own care and in improving the safety and quality of the health system as a whole, consumer engagement can be seen as little more than tokenistic. Coulter suggests it makes little sense to encourage people to get involved in collective action while treating them merely as passive recipients of health care in day-to-day interactions.99 If people are disengaged and disempowered when they are patients, they are unlikely to feel encouraged to participate as active healthy citizens; most will just want to forget about their experiences of the health service and get on with their everyday lives. Involving patients more actively is the basic building block for consumer engagement.99

Aiming to have active and informed consumers as equal partners in decision-making processes at all levels of the healthcare system is therefore the central concept for both consumer engagement and patient-centred care.

CASE STUDY 1: VICTORIAN COMMUNITY ADVISORY COMMITTEES

In Australia, Victoria was the first jurisdiction to enact legislation mandating community involvement committees to advise area health board governors on health service decision making. This occurred in 2002 under amendments made to the *Health Services Act 1998*.

Community advisory committees have a broad remit to provide direction and leadership in relation to the integration, recognition and reflection of consumer views in all areas of health service operations, planning and policy development.

An external evaluation of community advisory committees revealed that facilitators for the success of the community advisory committees included:100

- » adopting diverse approaches to consumer involvement
- » allocating adequate time
- » having clear lines of accountability
- » ensuring appropriate and reliable resourcing and support for the advisory committee towards achievement of the committee's agreed role
- » providing appropriate information and training to all participants and ensuring that members had a clear understanding of the health sector and health service bureaucracy
- » ensuring meetings were open and inclusive
- appreciating the circumstances of those participating (e.g. that the meeting environment may be intimidating)
- » reporting to participants so that they know what resulted from their involvement.

The evaluation found that a community advisory committee-type forum could only facilitate part of the required agenda for strategic and operational involvement between health services and consumers. There was a need to clarify the facilitative role of community advisory committees in relation to where they fit in the overall strategy of consumer engagement at all levels of the health service. Without this broader context, community advisory committees risked attempting an unmanageable and perhaps inconsistent agenda, resulting in frustration for council members and other consumer involvement stakeholders.

2.4

APPROACHES AND STRATEGIES TO PROMOTE PATIENTCENTRED CARE

A variety of approaches have been used internationally and in Australia to promote patient-centred care. A review by the Picker Institute Europe identified the most effective strategies, or 'best buys', for facilitating patient-centred care. Links to many of the strategies, approaches and resources discussed in this section are provided in Appendix C.

'Best buys' for improving patient experience

The Picker Institute Europe reviewed the body of evidence for strategies to engage patients and consumers in health care. They evaluated 31 systematic, high-quality narrative reviews on various initiatives to improve patients' experience, including consultation styles, communication skills training, service user involvement in evaluations, and studies of direct and indirect patient feedback (including patient experience and satisfaction surveys). Each study was graded according to four outcomes: impact on patients' knowledge, impact on patients' experience, impact on service use and costs, and impact on health behaviour and health status.

According to the Picker review, the most effective ways, or 'best buys', to improve patient experience are patient-centred consultation styles, communication training for health professionals and patient feedback (e.g. surveys, focus groups, complaints) with public reporting of performance data. The body of evidence for these strategies includes:

- » Impact on patients' knowledge: training health professionals to communicate information about medicines improves patients' knowledge and understanding. Longer consultations in primary care can increase patients' confidence to take action in relation to their health. Educational material can be helpful for carers.
- » Impact on patients' experience: patient surveys can stimulate quality improvements, but provider organisations need additional help to implement changes. Patient feedback surveys need to be well planned and carefully implemented. Patient-centred communication and longer consultations in primary care increase patient satisfaction. Communication skills training for clinicians can lead to improved communication, reduced anxiety and greater patient satisfaction.
- » Impact on service use and costs: public reporting of hospital performance data can stimulate providers to improve quality. If it is well disseminated and published in a format that patients can understand, this type of information influences public perceptions of a hospital's reputation, making it more likely that patients will want to go there. One review suggested that improved continuity of care reduces costs.
- » Impact on health behaviour and health status: communication skills development for clinicians may improve health outcomes, but some reviews have reported conflicting findings. Reviews of patientcentred consultations found mixed results in relation to impact on health status.

Implementing rights-based patient constitutions, charters or codes

Several countries provide patients with rights-based constitutions, charters or codes. These documents facilitate patient-centred health care by informing patients about the level of care they are entitled to expect. The NHS Constitution in England¹⁰² includes the following statements that have a bearing on patients' experience:

- » You have the right to be treated with a professional standard of care, by appropriately qualified and experienced staff, in a properly approved or registered organisation.
- » You have the right to expect NHS organisations to monitor, and make efforts to improve, the quality of health care they provide, taking account of the applicable standards.
- you have the right to be treated with dignity and respect.
- » You have the right to privacy and confidentiality.
- » The NHS will strive to ensure that services are provided in a clean and safe environment that is fit for purpose, based on national best practice.
- » The NHS will strive for continuous improvement in the quality of services you receive, identifying and sharing best practice in quality of care and treatments.

New Zealand has a similar code in place (*Code of Health and Disability Services Consumer Rights Regulation 1996*). ⁶¹ In February 2011, Scotland passed the *Patient Rights (Scotland) Bill (SP Bill 42)*. ¹⁰³ The Bill aims to create new statutory rights for those using the NHS (Scotland), including a statutory maximum treatment time guarantee, the introduction of a legal right to complain and the creation of a national Patient Advice and Support Service.

In the US, the American Hospital Association has developed the *Patient Care Partnership Brochure*.¹⁰⁴ Medicare patients are given a copy of the *Important Message From Medicare Notice*,¹⁰⁵ outlining patients' rights. Many organisation- and institution-specific patient rights constitutions or charters also operate in the US and Canada.

In Australia, the Australian Charter of Healthcare Rights provides a national statement of the rights of patients and people using the healthcare system. ¹⁰⁶ In addition, a number of Australia's states and territories have legislated charters of healthcare rights and human rights in place. ^{107–109}

Patient feedback

One method of driving improvements in the overall quality of an organisation is the collection and reporting of patient feedback.

Surveys are one of the most common tools for collecting patient feedback. Rather than relying on paper-based or telephone surveys, innovative methods such as handheld devices, bedside terminals and online tools are now being used to survey patients. Further information about Australian patient experience surveys is provided in Section 4.1.

When conducting surveys, it is important to ensure that survey participants reflect the entire population range and include people with disabilities, people from different cultures and age groups, and people from vulnerable populations. Surveys must also be accessible to these groups, and capture meaningful data that can be used to identify and address processes and practices that may compound any disadvantages or unique issues experienced by these populations.

In addition to surveys, deeper, qualitative methods of obtaining patient feedback can include:^{13, 110}

- » collection of patient stories, interviews and focus groups in which participants share their experiences and insights on a one-time basis
- » use of patient diaries to record healthcare experience through illness and treatment
- » panels of volunteers (e.g. patients, families, carers) to obtain views on a particular topic, concept or idea, such as hospital service changes, services provided in the community, ambulance services or intermediate care
- » staff-accompanied patient walk-throughs during their visit to a department to understand the whole patient journey from the user's perspective
- » anonymous shoppers reporting to management on processes of care.

In 2006 in New Zealand, the Health and Disability Commissioner began inviting the public to provide stories about great healthcare experiences, which were complied into *The Art of Great Care*. This document, and these stories, are being used to promote a focus on quality in New Zealand by describing the type of care that people want to experience.

In hospitals, staff can gather 'real-time' information from patients by engaging them in a conversation about their stay during patient safety rounds. The Dana-Farber Cancer Institute in the US has developed an online toolkit to help organisations initiate patient safety rounds. ¹¹³ In addition, the UK Department of Health publication, *Understanding What Matters—A Guide to Using Patient Feedback to Transform Services*, ¹¹⁰ sets out best practice in terms of collecting, analysing and using patient feedback. In Australia, WA Health has developed a toolkit for collecting and using patient stories for service improvement. ¹¹⁴

The aim of the reforms is to improve the way healthcare organisations listen to, respond to and learn from people's experiences. To ensure that complaints are resolved with a more personal and responsive approach, a single, comprehensive complaints system is in place across health and social care. England's Department of Health has published a guide and advice sheets on their new approach to complaints. $\ensuremath{^{\mbox{\tiny 110}}}$

US providers have complaints processes in place at the institutional and organisational levels. In addition, the Joint Commission's Office of Quality Monitoring evaluates complaints made by patients, carers and staff about its accredited providers that relate to safety and quality of care issues, such as patient rights and care of patients.

Australia also follows the model of locally managed organisation or service-level complaints processes and systems. Each state and territory also has a healthcare

complaints commissioner, who is responsible for responding to complaints escalated from local level mediation processes as well as directly from the public.

.ts commissioner, who is responsib.
.ing to complaints escalated from local.
cion processes as well as directly from the p.

A patient's view on making complaints

'My mother was very concerned about us
'ng anything to upset the apple cart,
'e she said, "I live in a country town.

very few specialists. I'm lucky

vee a heart specialist. If I cause
'ny refuse to see me. I won't
'ee. I may need him in the
'what she was saying.
'ed at some point to
'nd if I'm known
'ninant, they
'" and she
'just let

vch A path

'My moth,
doing anyth
because she sai.
There are very few
enough to have a hea
problems he may refuse,
have anybody to see. I may
future," basically is what she
She also said, "I may need at son.
go back to that hospital, and if I'm
as a difficult person, a complainant,
might not agree to look after me," and s.
said, "Quite frankly, I would rather just let
things lie."

100 Patient Stories Project Data from research
funded by the Commission

"gaging patients in safety improvem"

+-centred care involves more than ensuring
"xperience of care. It is also about enga"
vartners to identify safety issues ar
"rovement and risk reduction.
"ve safety and manage inr
"stems of care and pro"
a growing interest
"es, including i
"nts, and ir

patients as partners to identify safety issues and develop management.116-117

Patients can identify adverse events and incidents that occur in hospitals. 118-119 They have a unique perspective regarding the care they receive, and can provide information and insights that healthcare professionals might not otherwise have known.120 When adverse events occur, it is often the patient, their family or carer who has the most complete, first-hand knowledge of the events that happened—in contrast with the healthcare team, whose membership may change often. In addition, patient involvement in incident investigation meetings following an adverse event improves the dynamics of these meetings, and anchors people's sense of what went wrong to appropriate recommendations for change.117

Active patient participation in decision-making processes and the care of chronic illness has the potential to decrease medical errors and adverse events. 121 One study found an inverse relationship between patient's participation in their care and adverse events. 48 The results indicated that even passive participation contributed to a reduction in risk. This suggests that patients who participate in their own care are more likely to observe, identify and communicate potential issues, thereby mitigating risk and increasing the safety and quality of care. 48

While it is clear that patient participation in care improves safety and quality, research also suggests that patients are often discouraged from communicating their concerns to health professionals. Other obstacles to participation in care include health literacy levels, age and sociodemographic factors. ¹²¹

To promote patient engagement in safety initiatives, healthcare professionals need to be receptive and open to patient concerns and questions, and facilitate patients' attempts to be involved in their care. Brusque, dismissive or disinterested attitudes in the healthcare professional can inhibit patients from carrying out many of the roles proposed as means by which they might improve their safety. 116 Encouragement by doctors and nurses is crucial to engaging patient participation in promoting safe care. For example, patient willingness to ask challenging questions is increased when clinicians encourage active patient participation in care. 122

Opportunities that encourage patients to communicate their knowledge are increased if patients are provided with clear information about how to participate in their care, and if they are partnered in structured activities such as team handovers, ward rounds, team practice review meetings and near-miss discussions. For example, the US Joint Commission's Speak Up program encourages patients to take an active role in preventing healthcare errors by seeking to be a participant in the healthcare team. The Joint Commission has developed a range of educational brochures, posters and videos for patients on a variety of safety topics such as preventing error in your care, preventing infection, planning follow-up care and knowing your rights as a patient. 123 The Australian brochure 10 Tips for Safer Care has a similar role.124 To this end, the NSW Clinical Excellence Commission has established a Partnering with Patients program, which aims to foster the inclusion of patients and families as care team members to promote safety and quality.

There is also a growing recognition of the need to openly disclose adverse events to patients. Open disclosure is the 'open discussion of incidents that result in harm to a patient while receiving health care'. Open disclosure and incident management are important tools for learning organisations to assess and review processes, with a view to understanding patients' and carers' perceptions of their care when things go wrong. The *Australian Open Disclosure Standard*, ¹²⁵ published in 2003, aims to facilitate more consistent and effective communication after an adverse event. Open disclosure policies are also in place in Canada, ¹²⁶ England ¹²⁷ and the US. ¹²⁸

Using information and education to facilitate partnerships

The provision of information and education improves patient, family and carer capacity for involvement, understanding, participation and partnership in an individual's health care. It can also build an individual's capacity for meaningful engagement with healthcare organisations at the governance level.

Patient-centred approaches to engaging patients and families in their own care through the provision of information can include printed and electronic information, educational programs, telephone support and reminders, personalised information packages and sharing medical records.

The Picker review found that printed and electronic information and educational programs can benefit patients' knowledge and understanding of their condition, increase patients' sense of empowerment, satisfaction and ability to cope, and may help reduce anxiety. Among cancer patients, question prompt sheets and audiotapes of consultations improve the recall of medical information given in clinical consultations. Studies of stroke patients show that information strategies that actively involve patients and their care givers are more effective in reducing anxiety and depression.¹⁰¹

Telephone helplines, telecounselling and telemonitoring can also reduce social isolation, increase decision-making confidence and self-efficacy, and improve satisfaction. Telephone reminders can help to increase attendance rates and improve medication adherence, and reminder packaging may improve adherence to self-administered long-term medication.¹⁰¹

Personalised information packets given to patients on discharge are also useful.¹³ These provide customised health information to meet the specific needs of the patient and family. Packets may include fact sheets, recent articles, and information on local support groups and relevant complementary modalities.¹³

A certain level of health literacy is required for patients, families and carers to take full advantage of this type of information. Health literacy is an important driver of health behaviour, and lower health literacy is associated with poorer health outcomes. 129 In 2006, the Australian Bureau of Statistics found that approximately 59 per cent of Australians aged 15–74 had a health literacy level below minimum standard. 130 While efforts to improve health literacy have been discussed at an educational and societal level, health professionals and healthcare organisations can take actions to reduce literacyrelated barriers. These actions relate to the content and design of information and material, the physical and social environment of healthcare organisations, and the processes used to communicate with patients, consumers, families and carers. 131-132

.cords ha
.service prov.
.ledical records a.
.nt and carer engage.
.it to the bedside during t.
.linical tests and procedures
.ient. Future treatment plans th.
.arers and providers deciding on the
.f treatment together. Some providers I
.ntroduced patient progress notes, which giv.
.ent a formal place in the medical record to note
.ny comments or concerns, further encouraging a
partnership approach to care.

'ncouraging greater access to records

'Open Notes project is being conducted
'Beth Israel Deaconess Medical
the Geisinger Health System and
orview Medical Center in the US.
.100 doctors and 25 000 patients
ting in the project, which aims
at happens when a patient's
's made more accessible and
've patient.

't, primary care doctors
'o access their medical
ite hosted by their
The patients can
'ence and phone
'a any questions
ut of the notes.
'ing pre- and
'veys
tof
'yo

The Top 5 pv
support pv
tof
'li invol'
elicit'
ma'
Tv Traditionally, medical records have remained the providers now use medical records as a teaching tool and to encourage patient and carer engagement. The medical record is brought to the bedside during treatment and the results of clinical tests and procedures are shared with the patient. Future treatment plans then involve also introduced patient progress notes, which give the partnership approach to care.13

En.

The C, at the b.
Center, th.
the Harbon
More than 1.
are participath
to explore what
medical record is.
transparent for the.
As part of the project,
invite their patients to.
notes via a secure websit.
healthcare organisation. T.
then use email corresponden
conversations to follow up on.
they have regarding the contem.
The project will be evaluated usin,
post-design doctor and patient surv.
to explore the feasibility and impact to providing patients with ready access to doctors' notes in an electronic form.

Open Notes Project
www.myopennotes.org

'e Australian Government committed to the nt of a Personally Controlled Electronic Heal'
'ery Australian who wants one. This secrical history record will be a subset of 'ete medical history and will be ar nt and any authorised medice'
greater transparency of, ar 'r patients, and will cor information transfe ons.

Patient and carer engagement in

Health services and health professionals are adopting strategies to empower and facilitate patients, families and carers to be actively engaged in their own care. 134 A strong driver for this work has been the focus on shared decision making and self-management approaches, particularly for people with chronic and complex conditions. 135-137 Key interventions for increasing patient engagement in their own care that have been examined include:138

described and implemented in tral Coast Sector of North Sydney and Coast Area Health Service in NSW, dges the value of this information and supportive comotes personalised care.

am was initially designed to with cognitive impairment. Ingaging with carers to important tips and vies for patients. The Coast Sector of Morth Sydney and care service in the company of the for ease of access by

"ards in three sees reported mfortable in perceived the ealth service."

"ards in three sees reported mfortable in perceived the ealth service."

"ards in three sees reported mfortable in perceived the ealth service."

"and more research and the sees reported mfortable in perceived the ealth service."

"and more research and the sees reported mfortable in perceived the ealth service."

"and more research and the sees reported mfortable in perceived the ealth service."

"and more research and the sees reported mfortable in perceived the ealth service."

"and more research and more research and will be research and will Us.

Car infor aid he. progran. the Centh Central Co acknowled, It formalises effective com care, which pro
The Top 5 progra. support patients w. It involves staff eng. elicit their five most management strategi. Top 5 tips are recorded a patient's bedside chart for clinical staff.

Following a pilot in five wa Central Coast hospitals, nurs being more confident and con caring for patients, and carers 1 hospital as providing a better he Patients were reported as being m compliant with treatment, being less distre and having quicker recoveries and sh lengths of stay. Nurses also reported grown's statisfaction, better time managem. feeling safer in the work environment an receiving fewer complaints from families.

Central Coast Area Health Service Personal communication

Healthcare organisations can also look at how they put in place systems to encourage active involvement of family members and carers in the care process. The family or carer knows the patient best, and their presence can help reassure patients in times of uncertainty, anxiety or vulnerability. Family members and carers can also provide information about the patient's history, routines or symptoms that may assist in their treatment. Strategies can include patient-directed visitation, which removes restrictions on visiting times, allowing the patient to decide on the visiting times that best suit them. This strategy benefits overall patient experience and decreases anxiety by 65 per cent.13 A related strategy is the capacity for patients and families to trigger an escalation of care, calling a medical emergency or rapidresponse team when they are concerned about changes in a patient's condition.139 Other examples include care partner programs that aim to meet the personal, emotional, spiritual, physical and psychosocial needs of the patient by encouraging members of their support system to be involved in their care.13

Patient and carer engagement at the governance level

There is a strong tradition of consumer engagement in health service policy and planning internationally and in Australia. In addition to the input provided by consumers and citizens, patients, families and carers who have recently received health care in a particular service can have their own unique perspectives, and they also have roles in governance and planning processes. This can occur through their appointment to new advisory councils and committees, or through integration of consumer advisers into existing councils and committees.

Patient and family advisory councils can serve as the 'patient voice'. Such councils provide the institutional infrastructure for including patient and family members' perspectives in hospital organisational decision making. Councils typically include patients, family members, executive leaders and staff, working in partnership to assure delivery of the highest standard of comprehensive and compassionate care. Patients and carers may be members of existing advisory and governance committees, 13 or new committees being established. Activities may include:

- » providing information to hospital leaders and staff about patients' needs and concerns
- » helping plan patient care areas and new programs
- » making changes that affect patients and family members
- » encouraging patients and families to be involved and to speak up
- » strengthening communication among patients, family members, carers and staff.

In England, and more recently in Australia, the establishment of patient improvement advisory groups ('Critical Friends Groups') in general practices has proven a successful method of using patients' perspectives to improve quality in the primary healthcare sector. 140 The method involves a group of patients and practitioners working in partnership to identify and address issues of concern within the practice. Some practices have established groups to improve organisational issues such as telephone systems, information and dissemination practices, appointment structure, waiting room facilities, the role of the practice nurse and procedures for repeat prescriptions. Other practices use these groups to help practices interpret results of systematic patient feedback.

The Massachusetts Health Care Quality Act 2010¹⁴¹ requires all acute, chronic and long-term care hospitals in the US state of Massachusetts to establish a patient and family advisory council. These councils allow patients and family members to advise the hospital on matters including, but not limited to, patient and provider relationships, institutional review boards, quality improvement initiatives, and patient education on safety and quality matters, to the extent allowed by state and federal law. Membership of a council is determined by each hospital, and orientation, training and continuing education must be provided to council members.

Tra.

The Hamble from the develop the for consum.

It is a staff to develop the for staff to develop the for staff to develop and effective consuments of the highest standard in the highest standard in the highest standard in the safety an interest of the safety and interest of the

User-centred design and redesign

In both the US and England, approaches based on design of the physical care environment are emerging as successful tools to improve patients' experience of care. The approach in the US focuses on user-centred design. This is based on environmental psychology principles, in which spatial design is seen as an important part of patient experience and wellbeing. 143 Typically, a team of architects, psychologists and sociologists are employed to collect data from staff and patients via observations and interviews. The team maps the entire patient journey, from the first point of contact to the last, and then identifies core patterns and areas for improving the environment to better meet patients' emotional experience of care.

The leading case study for user-centred design is Kaiser Permanente's work in the US. This work improved staff efficiency, and interaction and information flow both among staff, and between staff and patients. It also provided examination rooms that could accommodate $% \left\{ \mathbf{r}^{\prime}\right\} =\left\{ \mathbf{r}^{\prime}\right\}$ carers and families, increased attention to privacy and comfort in patient rooms, and provided more comfortable waiting areas, easier check-in procedures and improved signage to help patients orient themselves more intuitively.143

In England, several NHS services have been redesigned. Service redesign focuses on technical aspects of the patients' journey, such as efficiency. Like user-based design, service redesign is based on insights from observing service users or mapping all the steps of a patient's journey. Service redesign brings together the patient and all key staff involved in the patient's journey to identify the problems throughout the journey. 144 Solutions are typically generated by staff. This is in contrast to user-centred design, which employs professionals outside the health service to map the patient journey and generate solutions.

Experience-based design and co-design

Experience-based design (EBD) is a methodology developed by Bate and Robert¹⁴⁵ for working with groups of patients and staff to improve services. It draws on knowledge and ideas from design sciences and professions in which products or buildings are improved by making the user integral to the design process.

In a healthcare setting, EBD focuses on how patients and staff move (or are moved) through the service and how they interact with its various parts. EBD works with patients to identify developments that could improve their care experience. It can also involve co-designing solutions with patients to address identified problems and improve patients' experience of using the service.

Involving patients and staff on an equal footing is more integral to EBD than to 'patient involvement' projects, in which patients are often treated as objects for study, rather than partners. How the service 'feels' functional and safe the service is. Bate and Robert provide a step-by-step guide to the EBD method, and illustrate it with a 12-month pilot program in head and neck cancer services at Luton and Dunstable NHS Foundation Trust, funded by the NHS Institute for Innovation and Improvement. Their method has also been used in other clinical specialties and hospitals in

A is seen as equally in

d safe the service is. Bate

ep-by-step guide to the EBD.

it with a 12-month pilot progran.

As cancer services at Luton and Dun.

Jation Trust, funded by the NHS Institution.

een used in other clinical specialties and hosy.

England and Australia.

*atient feedback influencing design

ign and construction of the new

usland Children's Hospital has been

ed by a Youth Advisory Group. The

tablished in 1997, comprises young

'd 12-18 who have experienced

'patient and outpatient

nents.

* every one to two months

'es related to the

Children's Hospital

'patient's perspective,

service delivery

's of the building,

ite visits

'g processes

uling working

volicy officers

ds, retail

mily-

Pat.

Desig.
Queens.
informea.
group, esta
people aged.
antion flow both
d patients. It also
at could accommodate
thetention to privacy
'd provided more
-heck-in procedures
its orient themselves

The group meets.
and discusses issue.
development of the C.
from a consumer and p.
including architecture, s.
and participated in planning
throughout the hospital, inclu.
with architects, designers and p.
to contribute to the design of ware,
areas, adolescent relaxation and fan.
focused spaces.

"S

Queensland Children's Hospital Youth
Advisory Council
www.health.qld.gov.au/childrenshospital.
content/establish_yaf.asp

"sultation styles and
munication training

ors and professionals in England, the US ar
've investigating the effects of patient-c'
n styles on continuity of care, lengt'
and increased support for patir
issues and concerns.io Rece

1 that the degree to whi
are courteous, parti
ivier of patient
arch in Enr
-all pati
ass demonstrates that overall patient satisfaction strongly correlates with patients' assessment of clinicians' interpersonal skills.148-149

> To improve these outcomes, initiatives to develop communication skills—such as training courses and coaching strategies that teach staff about the need

to establish a connection with patients—have been introduced successfully in a number of settings.13 Strategies include developing and using verbal communication guidelines for staff, and scripting tools and cues for effectively communicating with patients. The Patient-Centered Care Improvement Guide¹³ provides examples of scripting tools and cues. Nonverbal tools include patient and family communication boards with information such as the care team member names, and the scheduled date and time for specific procedures.

.ction with patients.sfully in a number of se.
.ide developing and using ve.
.ion guidelines for staff, and scrip.
.or effectively communicating with p.
.ent-Centered Care Improvement Guide* p.
.ples of scripting tools and cues. Nonverbal tc.
.ude patient and family communication boards v.
.iformation such as the care team member names, an the scheduled date and time for specific procedures.

\text{\carer's experience of communication}
\text{\carer's exper A can

'We arn
confuse,
worried h,
for a long ti
In due course
addressed my v
but I tagged alon,
man and his wife
and an appendage.
Then ominous silenc,
introductions; the doct
No introductions or exps
who might have been the
observer sitting in on the capatient had become an item v
belt. The doctor faced the desk
back to us. At no time did he tur.
face us. I entered the discussion o.
I felt that my comments were need,
my wife, because of her mental state,
not answer all the questions. I felt tha
presence was not welcome.'
Submission received in the consultation
on the draft patient-centred care
discussion paper

ful communication strategy that has facilitated
ntred care is the 'Just Ask Campaign' run
n Westchester Hospital in the US. Hospita'
a sign in every patient room encourar
'amilies with the message: 'If you'
'tit'. Displaying photographs of
g clothing that clearly disti
nd follow-up discharge nesses of communicating
tred care.'3

'onals thin'
e best pr
ling
Ir

A successful communication strategy that has facilitated

between providing the best practice evidence-based clinical care and providing patient-centred care. It is important to acknowledge that adopting a patientcentred approach to health care does not negate the need to adhere to best practice clinical standards and

guidelines. To achieve patient-centred care, health professionals need to negotiate a balance between care that aligns with the best clinical evidence available and care that is acceptable to the patient.150 Healthcare professionals must be good communicators who effectively involve their patients in decisions about

Another factor associated with success in patientcentred environments is that employees' behaviour consistently reflects the organisation's values. Only when employees' personal values simulate the core values of the organisation can the culture transform to a patient-centred model.¹³ At Sharp Coronado Hospital outside San Diego, California, every employee has completed the sentence 'I come to work to make a difference by...'; the laminated statement is adhered to their name badge as a constant reminder that what they do is meaningful and makes a difference to people's lives. Directly involving staff in determining organisational values and defining the behaviours that embody those values fosters a culture of patient-centred care. Making the organisation's values visible can remind staff of the patient-centred behaviours expected

Leadership and change management strategies

Enabling patient-centred care may require a change to the way organisations have traditionally provided care. Leadership and change-management strategies, including patient and staff rounding, fireside or chairside chats, and clinician, employee and board engagement, are successful ways to strengthen the foundation of patient-centred care.13 Staff rounding involves service-provider leaders moving through their organisation and hearing what is happening first hand from patients and staff. Patients and staff can express their views in a personalised, dialogue-based way that enables managers to understand the hospital experience. Similarly, staff may be invited to meet with managers to share ideas or express concerns in fireside or chairside chats. The strength of these communication methods is the direct interaction they afford between management, frontline staff and patients.13

Further strategies to establish and sustain patientcentred care include engaging staff and clinicians by involving them in patient-centred advisory councils, and identifying and recognising role models and champions of other patient-centred practices.13

Board- or governance-level strategies to facilitate patient-centred care include starting each board meeting with a patient story shared by the patient or a staff member, having a board member on a patient advisory council, and including patient-centredness as a defined and measurable organisational aim.13

Staff and practice development

In patient-centred care, the model of care delivery is not limited to a focus on the patient—considerable attention should also be paid to the experience of care providers.13 In a patient-centred environment, all employees are care givers, each affecting the patient's experience. Accordingly, there is a need to engage employees in fostering an atmosphere of patient-Jee-of-the
Jin a newslett.

Atient, family met.
Jo be an opportunity in the management team.
Lebes to staff recognition inclustion and the state of the second manager writes an inspirational we age that is recorded on a dedicated phone line a.
Je accessed by any employee. Scrolling screensave.

Jee and vareness of person-centred culture.

Jee and vareness of person-centred care

Enhancing Practice Program was

Joped by Northern Health, Victoria, in

Varation with the Council on the Ageing

we the way health services work

Tyeople and their carers.

a comprises a series of workshops,

health professionals and

at provide participants with

to reflect on current practice,

tunities for adopting a

at approaches to acre. The

llenge attitudes and

lige practice change.

dience includes

who work with

more than 1500

n 23 Victorian

the program.

the National

"to fthe Best

Care for

o was the

vorian

Accountability strates.

To become patient-centred, or

to encourage and promote e'

to encourage and promote centredness. Successful strategies to recognise and reward employees include an employee-of-the-month program, or public acknowledgment in a newsletter of a staff member's impact on a patient, family member or another share their story with the management team. More informal approaches to staff recognition include a simple acknowledgment or thank you during manager rounding. Some providers have successfully adopted a strategy in which a senior manager writes an inspirational weekly message that is recorded on a dedicated phone line and can be accessed by any employee. Scrolling screensaver messages and staff rounding are further tools to help

Staff satisfaction strategies

.rate.

.ror healthc.
.ney need to cr.
.nich their most in
.ued and treated with
.spect that the organisath
.patients and families. An i.
.ds commitment and engageme.
.isfaction surveys and monitor stat
.gh staff rounding and experience-bass
.gn. Directly involving employees in the d
.peration of patient-centred processes may als.
.taff satisfaction.

***uproving patient and staff experience**
.1. NSW Health commenced the
.1. This program is part of the patient
.2. Experience strategy and has been
.2. It is program is part of the patient
.2. Experience strategy and has been
.2. It is program is part of the patient
.2. Experience strategy and has been
.2. It is program is part of the patient
.2. Experience strategy and has been
.2. It is program is part of the patient
.3. Experience strategy and has been
.4. It is program is part of the patient
.4. It is a strategy and has been
.4. It is a strategy and has been
.4. It is program is part of the patient
.4. It is a strategy and has been
.4. It is a strategy and ha According to Shaller, 15 for healthcare organisations to be patient-centred, they need to create and nurture an environment in which their most important asset, their workforce, is valued and treated with the same level of dignity and respect that the organisation expects them to provide to patients and families. An important way to achieve this commitment and engagement is to conduct staff satisfaction surveys and monitor staff experience through staff rounding and experience-based codesign. Directly involving employees in the design and operation of patient-centred processes may also improve

in professional codes of conduct to define the way patient-centred care is provided. Some patient-centred

organisations in the US have incorporated performance measures to support the behaviours defined in codes of conduct¹³ and performance incentives based on improving quality metrics, including patient experience.¹⁵²

CASE STUDY 2: MEDICAL COLLEGE OF GEORGIA HEALTH INC., GEORGIA, UNITED STATES

Located in Augusta, Georgia, Medical College of Georgia Health Inc. is a non-profit health service affiliated with the Medical College of Georgia (MCG). The medical district, including the university hospital, is the largest employer in Augusta and has more than 25 000 staff. MCG Health is a 632-bed tertiary medical centre with more than 22 000 admissions per year and 455 000 outpatients.

MCG Health's focus on patient-centred care started in 1998, when they built a new paediatric medical centre. Patients and families were encouraged to participate in their care and were recognised as care team members and partners in improving care delivery. Because of this, MCG Health is considered a pioneer in the field of patient- and family-centred care. Once the benefits of a patient-centred focus were recognised in the paediatric area, strategies were broadened to include adult services. Strategies used by MCG Health include:

- » establishing patient advisory councils; MCG Health has the largest patient adviser program in the US, with more than 130 trained patient and family advisers involved in safety and quality, planning, and design throughout the organisation
- » removing all visiting hours; family and carers can visit at any time preferred by the patient (including in the intensive care unit), and they are not viewed as 'visitors', but as 'partners'
- » giving patients and family the right to call out rapid response teams
- » ensuring that patient advisers sign-off on plans for facility redesign and new fit-outs
- » introducing executives' and patient advisers' rounding on wards
- » providing patients' feedback to staff on a regular basis
- » conducting training across the whole organisation in patient-centred customer service values
- » redesigning patient accounts based on patient advice.

More recent undertakings include the redesigning of the MCG Medical Center Breast Cancer unit by patients. Within three years, patient care experience ratings for the unit moved from the 40th to the 74th percentile, rated using surveys administered by an independent vendor, Press Ganey. The neurology intensive care unit was also renovated with patients' input, introducing a 24/7 family access policy. Within five years, patient satisfaction ratings increased from the 10th to the 95th percentile and the length of stay of patients decreased by 50 per cent.

At this point, staff acknowledged that the chief executive officer 'saw the business case' and became a patient-centred care 'convert'. During this time, MCG Health witnessed a number of clinical and operational benefits, including a falling overall staff vacancy rate—from eight per cent to zero per cent—with a current waiting list of several hundred people. With a continued focus on patients in 2010, planning has begun for a new cancer centre with patient input into the design (Pat Sodomka, Sebrio Vice President, Patient and Family Centered Care, MCG Health Inc. and Director, Center for Patient and Family Centered Care, Medical College of Georgia, pers comm, 2009).

2.5

AN OVERVIEW OF LEADING INTERNATIONAL ORGANISATIONS

Several international private or not-for-profit organisations advocate specific patient-centred frameworks or approaches to health service improvement built around a combination of the strategies outlined in Section 2.4. Both public and private health service providers have used these frameworks. The leading organisations advocating patient-centred healthcare frameworks are described below. Websites and links to resources on strategies to promote patient-centred care are provided in Appendix C.

World Health Organization

The World Health Organization (WHO) is the United Nation's leading authority on global health matters. Its role includes setting standards and developing evidence-based policy.

WHO advocates for a 'responsive' healthcare system that meets people's expectations. ¹⁴ It also advocates for involving patients and carers as partners in initiatives to improve the safety and quality of care, ¹⁵³ particularly through its Patients for Patient Safety program. This program works with a global network of patients, consumers, care givers and consumer organisations to support patient involvement in patient safety programs at local, regional and international levels. WHO has established a global network of Patients for Patient Safety champions, including 13 Australian champions, who work in partnership with health professionals and policy makers across the world to identify problems, design solutions and implement change.

The program is designed to ensure that the perspective of patients and families is a central reference point in shaping the safe and high-quality delivery of health services, including health service decision making. It aims to develop opportunities for patient voices to be heard in creating public awareness about inherent

healthcare risks, educate the public about systems approaches to risk management, report errors or healthcare failures in ways that contribute to systemic learning, and disseminate research and share solutions that can prevent patient harm.¹⁵³

Institute for Healthcare Improvement

The Institute for Healthcare Improvement (IHI) is an independent not-for-profit organisation, founded in 1991 and based in Cambridge, Massachusetts. The IHI aims to identify best practices and bring about system changes that enable healthcare providers to become and remain patient-centred.

The IHI has developed tools to assist healthcare providers, including an 'improvement map', which sets out the processes for improving patient-centred care in a variety of areas. These include advanced care planning, communication with patients and families after an adverse event, daily goal setting and planning for treatment and discharge, multidisciplinary patient rounding for coordinated care, shared decision making, patient experience programs, and leadership programs to enable patient-centred care.

The IHI website provides a variety of publications, resources, tools and improvement stories to help organisations achieve patient-centred care. In 2011, the IHI published a white paper to assist hospitals in achieving exceptional patient and family experiences of inpatient care. ¹⁵⁴

Picker Institute

The Picker Institute was established in 1994 as an independent non-profit organisation based in Boston, Massachusetts. The institute's goal is to foster a broader understanding of the practical and theoretical implications of patient-centred care by focusing on the concerns of patients and other healthcare consumers. The Picker Institute pioneered the use of scientifically validated, nation-wide surveys and databanks on patient-centred care to educate doctors and hospital staff on improving patient services from a patient's perspective. The surveys cover the eight Picker dimensions of patient care (set out in Section 1.1). Patients report on what happened to them rather than rate how satisfied they were, as Picker research has shown that simple patient satisfaction questionnaires do not produce useful results. Picker Institute surveys are used by regulators in the US, England, Canada and Australia to measure patient-centred care.

In addition to their surveys, the Picker Institute provides education programs, research grants and publications on patient-centred care topics, including *The Patient-Centered Care Improvement Guide*, and a self-assessment tool written in conjunction with Planetree (see below). The institute has also established the Always Events program, which aims to establish a set of events

that patients can expect to occur at every healthcare experience, through demonstration projects and the establishment of a collaborative learning network.

A variety of resources on patient-centred care is provided on the website of the institute's sister organisation, Picker Institute Europe.

Institute for Patient- and Family-Centered Care

The Institute for Patient- and Family-Centered Care (IPFCC) is a non-profit organisation founded in the US in 1992. The institute's mission is to advance the understanding and practice of patient- and family-centred care. In partnership with patients, families and healthcare professionals, the organisation integrates these concepts into all aspects of health care. It provides education, consultation, technical assistance, materials development, information dissemination and research on patient-centred care.

The IPFCC website provides numerous practical resources, including assessment tools, guidance publications and multimedia resources produced by the institute and other leading patient-centred care organisations. Notable publications include:

- » Advancing the Practice of Patient- and Family-Centered Care: How to Get Started⁵⁷
- » Partnering with Patients and Families to Design a Patient- and Family-Centered Health Care System— Recommendations and Promising Practices¹⁵⁵ (this publication sets out practical examples of patientcentred care in the US)
- » Partnering with Patients and Families To Design a Patient- and Family-Centered Health Care System: A Roadmap for the Future—A Work in Progress.²⁸

Planetree

Planetree was founded in 1978 by Angelica Thierot, a patient who was disheartened by her own experience of hospital care, which she felt was de-personalising. Her vision of a different type of hospital was one in which patients would become active participants in their own care and wellbeing. Planetree began by developing information libraries for patients, and has grown into a non-profit organisation providing education and information on patient-centred care. It facilitates efforts to create patient-centred care in healthcare organisations in the US, Canada and Europe.

Based on the feedback of thousands of patients and hospital staff members, Planetree has identified core principles that are essential to practising patient-centred care, published in *Putting Patients First: Designing and Practicing Patient-Centered Care*. 156

The Planetree method for fostering the core principles includes:

- » working with senior leaders to develop and share ownership of an organisational strategy to drive improvement in patient-centred care
- » multidisciplinary and multiprofessional training for all staff in the core principles, mainly through one and two-day retreats
- » expressing the desired culture and values in an accessible language
- » positively and continuously reinforcing Planetree values and principles with awards and recognition for individuals
- » emphasising the built environment.

In 2008, Planetree partnered with the Picker Institute to produce *The Patient-Centered Care Improvement Guide.* ¹³ This and other Planetree publications are available on the Planetree website.

Studer Group

The Studer Group is a US-based organisation that partners with healthcare providers to increase performance by improving employee and clinician satisfaction and experience. Improved performance is achieved through 'the nine principles': commit to excellence; measure the important things; build a culture around service; create and develop leaders; focus on employee satisfaction to improve patient satisfaction; build individual accountability; align behaviours with goals and values; communicate at all levels; and recognise and reward success.

The Studer Group provides on-site resources to demonstrate their approach. Resources include tools for improving communication interactions between staff and patients, such as 'key words at key times', 'discharge phone calls' and 'patient rounding' tools. Other tools concentrate on improving patient experience by first improving staff satisfaction and experience, such as the 'staff rounding for outcomes' and 'thank you notes' tools. There are also several supplemental resources available to nonpartners, including leadership development resources and evidence-based tools.

Kenneth B Schwartz Center

The Kenneth B Schwartz Center is a non-profit organisation based at Massachusetts General Hospital in the US. It was founded by the family of a patient (Schwartz) in 1995, with a mission to 'provide support and advance compassionate health care, in which caregivers, patients and their families relate to one another in a way that provides hope to the patient'.

The centre runs programs to educate, train and support care givers in 'the art of compassionate health care', and to strengthen the relationship between patients and care providers, including Schwartz Center Rounds,

the CarePages website and the Patient Voice for Compassionate Care program.

Schwartz Center Rounds are a multidisciplinary forum in which care givers discuss difficult emotional and social issues that arise in caring for patients. An evaluation of Schwartz Center Rounds¹⁵⁷ found many benefits, including improving communication and teamwork among care givers, and communication between care givers and patients. This study also found that participating in rounds increased the likelihood of staff attending to the psychosocial and emotional aspects of care, and improved their beliefs about the importance of empathy.

The CarePages website contains patient blogs that connect friends and family during a health challenge, helping them support each other. The Patient Voice for Compassionate Care: Schwartz Center Dialogues is a pilot program that brings the experiences and perspectives of patients and their families directly to care providers in a series of facilitated discussions. The program helps patients and their families become stronger advocates and partners in their own care. At the same time, it improves the communication skills of care givers and the quality of compassionate care they deliver. The panel of patients and care givers identifies specific changes necessary to improve communication, and reconvenes over time to ensure that the changes are made and the patient focus is sustained. Information on each of these programs is available on the Schwartz Center website.

The King's Fund

The King's Fund is a charity that seeks to understand how the health system in England can be improved. The King's Fund Point of Care program aims to transform patients' experience of care in hospital by enabling healthcare staff to deliver the quality of care they would want for themselves and their own families. The King's Fund works with staff, hospital boards, and patients and their families to research, test and share new approaches to improving patients' experiences.

Seeing the Person in the Patient: Point of Care Review Paper¹⁶ considered current debates and dilemmas in relation to patients' experience of care in acute hospitals in England. Based on this research, The King's Fund developed a framework to analyse the factors that affect patients' experience at four levels: individual interactions with staff members, the 'clinical microsystem' of the team, unit or department; the hospital; and the wider healthcare system (including NHS priorities). These levels are similar to the Institute of Medicine's levels of patient-centred care described in Section 1.4.

The relevance of patient-centred care to the Australian health system

This section describes the applicability of patient-centred care approaches and strategies to the Australian health system, provides an overview of the current policy reform context of patient-centred care in Australia, and highlights national-level initiatives, strategies and policies that promote patient-centred care. strategies and policies that promote patient-centred care.



Australia has a mixed healthcare system, in which funding for, and delivery of, health care is divided between the public (represented by the Australian Government, six states and two territories) and private sectors. Despite the complexity of the Australian healthcare system, many of the strategies and approaches to patient-centred care used internationally are highly applicable to Australia.

3.1

THE RELEVANCE OF PATIENT-CENTRED CARE TO A MIXED HEALTHCARE SYSTEM

Australia's health system is primarily funded through general taxation and a small, compulsory, tax-based health insurance levy. Medicare, the tax-funded national health insurance scheme, offers patients free public hospital care and subsidised access to their doctor of choice for out-of-hospital care. The Australian Government's Pharmaceutical Benefits Scheme provides subsidised access to a wide range of medicines for all Australians. The public system administers public hospitals and a range of other primary and community health services.

The private sector includes the majority of doctors, both general practitioners (GPs) and specialists, pharmacists, numerous private hospitals and day hospitals, a large diagnostic services industry and a number of private health insurance funds. Ownership of private hospitals is mostly centralised, with more than two-thirds of all private beds owned by large for-profit chains or the Catholic Church. 158 GPs and pharmacists are largely self-employed and funded by Medicare and the Pharmaceutical Benefits Scheme through fee-for-service, the Practice Incentive Program and the Fifth Community Pharmacy Agreement (including the Pharmacy Practice Incentive and Accreditation Program). These last two programs offer financial incentives for accredited GPs and pharmacists, respectively, to improve the quality and accountability of their services.

Both the public and private sectors are seeking to improve the overall quality of health care. These initiatives are supported and driven at the national policy level via recent national reforms that seek to embed patient experience as a measurable and reportable component of quality. National-level service strategies and plans provide a further lever for the government to promote a patient-centred approach to health care in Australia.

The current reform of accreditation for health care facilities, led by the Commission, is another aspect of Australia's healthcare system that could further facilitate patient-centred care. To date, the requirement for healthcare organisations to be accredited has varied across public, private, acute, primary and aged care sectors. For example, in the acute sector, most public hospitals are required to seek accreditation and most large hospitals seek accreditation. 158 In addition, in some jurisdictions, private hospitals are also required to be accredited as part of their licensing arrangements. Many private hospitals will voluntarily seek to meet accreditation standards as private health insurers pay higher reimbursement rates to accredited facilities.158 However, accreditation coverage is far from consistent or complete.

The Commission's work on accreditation reform and the development of the National Safety and Quality Health Service (NSQHS) Standards will create greater consistency of accreditation by introducing the NSQHS Standards to be used in accreditation processes for hospitals and day procedure services. The application of the NSQHS Standards will incorporate patient-centred principles into accreditation requirements, particularly through Standard 2, 'Partnering with Consumers'.

3.2

NATIONAL HEALTH REFORM

National policy reforms set out in the *National Health* and *Hospitals Network Agreement*¹⁵⁹ and the *Heads of Agreement*—*National Health Reform*¹⁶⁰ focus on the need to improve access to services, quality of service delivery, financial responsibility, patient outcomes and patient experience.

The reforms include a new *Performance and Accountability Framework*, which consists of national performance indicators, new hospital and health service performance reports, and national clinical safety and quality standards to be developed by the Commission.

The Performance and Accountability Framework builds on the requirement for public reporting on objectives and outcomes of the National Healthcare Agreement.

This agreement identifies a number of long-term goals for Australian governments to improve health outcomes, including that 'Australians have positive health and aged care experiences, which take account of individual circumstances and care needs'.

161

Progress towards these objectives and outcomes will be measured by nationally comparative information that indicates patient satisfaction levels on key aspects of the care they receive. In the first National Healthcare Agreement report in 2010, agreement had not yet been reached on how to measure this progress, and no data were included. In the future, progress reports will be drawn from patient experience surveys in each jurisdiction, and a national population-based survey conducted by the Australian Bureau of Statistics on the healthcare experiences of individuals in private households (see Section 4.1).¹⁶¹

The draft national performance indicators set out in the *Performance and Accountability Framework* provide potential levers for improvement in patient-centred care, and quality more generally. The current draft indicators acknowledge the need to measure traditional operational and clinical performance indicators (such as hospital waiting times and hospital acquired infections) alongside patient-reported quality based on the patient experience. The indicators also facilitate transparent monitoring across services and jurisdictions, and will set a minimum standard of healthcare quality to which each provider should adhere.

A new, independent National Health Performance Authority is expected to be established as part of the health reform governance framework. The National Health Performance Authority is anticipated to have responsibility for producing performance reports for Local Hospital Networks, hospitals (both public and private) and Medicare Locals.

Nationally consistent, activity-based funding is anticipated to be introduced for public hospital services. This is likely

to be based on a nationally efficient price determined by a new Independent Hospital Pricing Authority (IHPA). When setting the nationally efficient price, the IHPA is expected to consider a range of issues, such as reasonable access, clinical safety, efficiency and fiscal considerations, as well as healthcare needs of people living in rural Australia, and Aboriginal and Torres Strait Islander peoples. ¹⁵⁹ By linking this funding to performance, the nationally efficient price is expected to consider clinical indicators and outcomes, as well as patient experience. ¹⁵⁹ Improving the patient experience through patient-centred care, therefore, has important implications for both patients and providers.

The Commission will also play a role in the implementation of health reform. In 2011, the Commission was established as an independent permanent body with an expanded remit to develop, monitor and implement national standards for improving clinical safety and quality in hospitals and healthcare settings. These standards will complement the NSQHS Standards.

3.3

NATIONAL STRATEGIES AND INITIATIVES PROMOTING PATIENT-CENTRED CARE

A variety of national service-level initiatives, strategies and policies set out a patient-centred approach to health care. National initiatives include the Australian Charter of Healthcare Rights and the Australian Safety and Quality Framework for Health Care. National strategies, such as the National Primary Health Care Strategy, the National Chronic Disease Strategy, the Fourth National Mental Health Plan and the Fifth Community Pharmacy Agreement, all state that a patient-centred approach to health care is needed to improve the quality of health care in Australia. Current Aboriginal and Torres Strait Islander policies also reflect patient-centred principles and focus on family and community.

Australian Charter of Healthcare Rights

The Australian Charter of Healthcare Rights underpins the provision of safe and high-quality care, and supports a shared understanding of the rights of patients and consumers between those seeking health care and those providing health care. From 1993, each Australian state was required to develop public hospital patient charters to inform patients of their rights under the Australian Health Care Agreements. In July 2008, Australian health ministers endorsed a single national charter as a clear statement of a minimum set of standards, rights, expectations and entitlements that is uniformly applicable across all states and territories, and in all settings of care. A copy of the Australian Charter of Healthcare Rights can be found in Appendix D.

Australian Safety and Quality Framework for Health Care

In 2010, Australian Health Ministers endorsed the Australian Safety and Quality Framework for Health Care. The overall aim of the framework is to provide direction for improving the healthcare system. In developing the framework, the Commission recognised that consumer engagement is vital to safety and quality improvements, and should underpin the strategic plan for the system. 164

The Commission has developed a series of guides for policy makers, managers, healthcare teams, executives and boards to support the use of the framework. A copy of the Australian Safety and Quality Framework for Health Care—including the actions proposed for health systems and providers to achieve patient-centred care—can be found in Appendix E.

National Safety and Quality Health Service Standards

The NSQHS Standards provide an explicit statement of the expected level of safe and good-quality care to be provided to patients. There are ten standards, nine

of which have components relevant to patient-centred care. Two of the standards—Standard 1, 'Governance for safety and quality in health service organisations' and Standard 2, 'Partnering with consumers'—describe explicitly how healthcare organisations should engage with consumers at a governance and organisational level. These two standards form the foundation on which each of the other standards should be applied. In addition, seven of the remaining standards include criteria that describe different requirements for engaging with patients, through communication and the provision of information for each of the specific topic areas. Further detail on these standards can be found in Table 1

In 2010, health ministers agreed that all health services that pose a high risk of harm to patients be accredited against the NSQHS Standards.^a Under these arrangements, accrediting agencies will assess health services using the NSQHS Standards, with the transition period to commence from 1 July 2011 and full implementation by early 2013. In addition, all other health services may use the NSQHS Standards as part of their internal safety and quality assessment processes.

Table 1

Excerpt from the National Safety and Quality Health Services Standards: standards and criteria relevant to patient-centred care

STANDARD	CRITERIA RELEVANT TO PATIENT-CENTRED CARE
Standard 1: Governance for safety and quality in health service organisations	Patient engagement and rights: Patients' rights are respected and their engagement in their care is supported.
Standard 2: Partnering with consumers	Consumer partnership in service planning: Governance structures are in place to form partnerships with consumers and/or carers.
	Consumer partnership in designing care: Consumers and/or carers are supported by the health service organisation to actively participate in the improvement of the patient experience and patient health outcomes.
	Consumer partnership in service measurement and evaluation: Consumers and/or carers receive information on the health service organisation's performance and contribute to the ongoing monitoring, measurement and evaluation of performance for continuous quality improvement.
Standard 3: Preventing and controlling healthcare associated infection	Communicating with patients and consumers: Information on healthcare associated infection is provided to patients, carers, consumers and other service providers.
Standard 4: Medication safety	Communicating with patients and carers: The clinical workforce informs patients about their options, risks and responsibilities for an agreed medicines management plan.
Standard 6: Clinical handover	Patient and carer involvement in clinical handover: Health service organisations establish mechanisms to include patients and carers in clinical handover processes.

a High-risk services are defined as those that undertake invasive procedures into a body cavity or dissecting skin while using anaesthesia or sedation.

STANDARD	CRITERIA RELEVANT TO PATIENT-CENTRED CARE
Standard 7: Blood and blood products	Communicating with patients and carers: Patients and carers are informed about the risks and benefits of using blood and blood products, and available alternatives when a treatment plan is developed.
Standard 8: Preventing and managing pressure injuries	Communicating with patients and carers: Patients and carers are informed of the risks, prevention strategies and management of pressure injuries.
Standard 9: Recognising and responding to clinical deterioration in acute care	Communicating with patients and carers: Patients, families and carers are informed of recognition and response systems, and can contribute to the processes of escalating care.
Standard 10: Preventing falls and harm from falls	Communicating with patients and carers: Patients and carers are informed of the identified risks from falls and are engaged in the development of a falls prevention plan.

Source: Australian Commission on Safety and Quality in Health Care 2011¹⁶⁵

National Primary Health Care Strategy

The National Primary Health Care Strategy¹⁶⁶ is a highlevel action plan consisting of ten elements aimed at improving primary health care through patient-centred approaches. It sets the policy direction to better connect hospitals, primary and community care to meet patient needs, improve continuity of care and reduce demand on hospitals. Key priority areas identified in the National Primary Health Care Strategy include:

- » improving access and reducing inequity
- » better management of chronic conditions
- » increasing the focus on prevention
- » improving quality, safety, performance and accountability. 166

Although the Australian primary healthcare system serves many patients well, it has not been specifically designed to cater for the particular health needs and cultural requirements of some groups, such as people from culturally and linguistically diverse backgrounds, Aboriginal and Torres Strait Islander people, or disadvantaged and marginalised populations. These groups may find it difficult to access appropriate services within the system, or to know which services to access and when. 166 The National Primary Health Care Strategy recognises that an inclusive and patient-centred focus is a key element of all future reforms in primary health care.

National Chronic Disease Strategy and National Service Improvement Framework

Promoting patient-centred care is a central aim for improving health service delivery, outlined in the *National Chronic Disease Strategy* (NCDS) and the *National Service Improvement Framework* (NSIF) for asthma, cancer, diabetes, heart disease, stroke and vascular disease. 167 NCDS and NSIF place the patient, their family and carers at the centre of a broader health promotion and disease prevention framework. This includes the core principle of achieving patient-centred

care and optimising self-management. Self-management helps people to take responsibility for their own health, to make informed decisions, and to maximise their wellbeing and quality of life. 167

National Mental Health Plan

The Fourth National Mental Health Plan—An Agenda for Collaborative Government Action 2009–2014¹⁶⁸ outlines reporting requirements for service providers against agreed standards of care, including consumer and carer experiences and perceptions. A priority in this plan is ensuring that consumers and carers can access information about service provider performance across the range of health quality domains and compare this to national benchmarks. The principles behind this priority include the following patient-centred concepts:

- » Consumers, their carers and families should be actively engaged at all levels of policy and service development. They should be fully informed of service options, and anticipated risks and benefits.
- » Consumers and carers should be able to access information in a language they understand, or have access to interpreters.
- » Families and carers should be informed to the greatest extent, consistent with the requirements of privacy and confidentiality, about the treatment and care provided to the consumer, the services available and how to access those services. They need to know how to get relevant information and necessary support.

The Fourth National Mental Health Plan also calls for the recognition of social, cultural and geographic diversity, and experience of consumers and carers. 168

Fifth Community Pharmacy Agreement

The Fifth Community Pharmacy Agreement¹⁶⁹ commenced on 1 July 2010. It encourages pharmacists to adopt a more patient-centred approach to care by providing support for pharmacists to identify, resolve and document medicine-related issues experienced

by patients. The agreement also sets out a new patient service charter that outlines the role and responsibilities of the pharmacist and the pharmacy.

Aboriginal and Torres Strait Islander primary healthcare policies

In addition to the National Primary Health Care Strategy, Aboriginal and Torres Strait Islander primary healthcare policies reflect patient-centred principles and aim to be community- and family-centred. Patient-centred care in this context requires a critical focus on the cultural safety of, and cultural respect for, the patient, their family and community, and may involve considerations that are not normally associated with health care in mainstream services.

The Office for Aboriginal and Torres Strait Islander Health provides direct grants to approximately 280 Aboriginal community controlled health services and Aboriginal medical services. These primary healthcare services are initiated and operated by local Aboriginal communities. They deliver holistic, comprehensive and culturally appropriate health care to the community that controls it, through locally elected boards of management.

The National Aboriginal Community Controlled Health Organisation is the peak national Aboriginal health body representing Aboriginal community-controlled health services throughout Australia. According to the organisation, this model of primary health care is in keeping with the philosophy of Aboriginal community control and the holistic view of health that this entails:

Aboriginal health is not just the physical well being of an individual but is the social, emotional and cultural well being of the whole community in which each individual is able to achieve their full potential thereby bringing about the total well being of their community.170

In a review on the link between primary health care and health outcomes for Aboriginal and Torres Strait Islander people, Griew¹⁷¹ identified the following patient-centred factors in association with successful local-level primary healthcare interventions:

- » genuine local Aboriginal and Torres Strait Islander community engagement to maximise participation, up to and including full community control
- » a multidisciplinary team approach employing local community members
- » service delivery that harmonises with local Aboriginal and Torres Strait Islander ways of life.

Current reform activity in relation to Aboriginal and Torres Strait Islander primary health care emphasises the importance of a family-centred approach. Family in this context has a broad community focus and recognises, for example, that Aboriginal and Torres Strait

Current jurisdictional and other activity in Australia

This section highlights some of the patient-centred care initiatives being undertaken in Australia, ranging from the adoption of patient experience measurement strategies at a state level to the implementation of co-design and peer support projects at a local level.





Many of the initiatives described in this section evolved from the need to address the safety and quality of health service delivery, revealed by several high-profile, statebased inquiries. These inquiries involved patient care and adverse clinical incidents at the King Edward Memorial Hospital in Perth, 173 Canberra Hospital, 174 Royal Melbourne Hospital, 175 Campbelltown and Camden hospitals, 176 Bundaberg Hospital and Queensland Health, 177-178 Royal North Shore Hospital, 179-180 and, more generally, acute care services provided by NSW Health. 181-182

Hindle and colleagues¹⁸³ reviewed the common features of the King Edward, Royal Melbourne, and Campbelltown and Camden inquiries. They found that quality monitoring processes were deficient, and that patients and families were not informed members of the team. Patients were not adequately involved in care planning and did not always have an adequate basis for informed consent. They were also unsure of their rights and were frequently afraid of exercising them, sometimes treated in inconsiderate ways, and seldom received sympathetic or helpful support when they made complaints.

The inquiries concerned allegations of poor clinical practice and poor communication between healthcare organisations and patients, their families or other carers, which led to a loss of trust in the health system by patients and the community.85

The response to each of these inquiries was an increased focus on patient-centred care. Several strategies to achieve this are in place within jurisdictions, including improving communication with patients, improving complaints processes, promoting healthcare rights, increasing reporting of hospital activity and performance in relation to patient safety and quality issues, and increasing consumer involvement in health service planning and design.184-185 These and other approaches are discussed in this section. Links to many of the organisations, programs and projects discussed in this section are included in Appendix C.

4.1 **LISTENING TO** PATIENT FEEDBACK

Most Australian jurisdictions conduct surveys to measure patients' experience across different settings. In New South Wales, Patient and Care Experience surveys are conducted annually. Condition-specific surveys are also emerging, with a care experience survey for cancer patients conducted through the Cancer Institute NSW. 186 Other state surveys include the Victorian Patient Satisfaction Monitor survey, the Queensland Health Patient Satisfaction survey, the Healthcare Survey (Australian Capital Territory), and the Patient Evaluation of Health Services survey (South Australia and Western Australia). Tasmania has a Patient Satisfaction System in place as part of its overall framework for patient-centred care.187

Survey results are used to identify trends, monitor performance, benchmark results against similar service providers, and inform health service planning and patient safety and quality initiatives. 188 Patients and consumers can be part of the process of interpreting the results of surveys, and can provide direction about priority areas for change.189

The Australian Bureau of Statistics (ABS) also collects population-based information about healthcare experiences from private households. Part of the annual Multipurpose Household Survey identifies healthcare issues at a national level across a range of key delivery areas, including hospitals (inpatient and emergency), general practice, specialists, dentistry, and pathology and imaging. Unlike the state and territory surveys, the ABS asks household members to recall a range of experiences over the preceding 12-month period, rather than a specific service experience. This survey of consumer experience, first conducted in July-December 2009 on a sample of about 6500 people, has the advantage of

capturing information about people who could not access services and exploring the underlying reasons. Future ABS surveys will be expanded to capture the experiences of more than 20 000 people from 2010–11.

The Council of Ambulance Authorities has conducted annual national patient satisfaction surveys since 2002, ¹⁹⁰ which measure the quality of the ambulance service as perceived by its customers. The survey results are compared across states and territories, and are used to set performance benchmarks.

The private sector, including general practitioners (GPs), insurers and hospitals, also conduct surveys to capture patients' experiences and feedback. There is little publicly available information on the content or scope of private sector surveys.

The fourth edition of the Royal Australian College of General Practitioners Standards¹⁹¹ requires GPs to obtain feedback from patients. More specifically, the standards require that general practices actively seek feedback about patients' experiences of the practice at least once every three years, demonstrate improvements they have made in response to analysis of patient feedback, and provide information to patients about practice improvements made as a result of their input.

Another source of feedback is complaints from patients and consumers. Complaints received at a local level provide an important source of information about the experience of patients receiving care. In addition, all states and territories have a healthcare complaints commission, or similar body, which works with patients and providers to manage complaints.

4.2

THE AUSTRALIAN INSTITUTE FOR PATIENT AND FAMILY CENTRED CARE

The Australian Institute for Patient and Family Centred Care was founded in 2009 by a diverse group of healthcare professionals, patients and families. The institute's aim is to bring patients, families and healthcare professionals to the same table to transform the experience and improve the outcomes of health care.

The institute's three key pillars of work are:

- » Partnership: Building strong healthcare partnerships between patients, families and healthcare workers.
- » Culture: Promoting a supportive and transparent workplace culture within healthcare environments across Australia.
- » Environment: Encouraging the development of improved healing environments through meaningful art, architecture and design.

The activities of the institute include:

- » advancing the understanding and practice of patientand family-centred care through training and consultancy
- » partnering with universities, hospitals and insurers to support research into patient- and family-centred practice
- » advising healthcare program planners and decisionmakers on policy that supports patient- and familycentred practice
- » maintaining a library of information on patient- and family-centred practice accessible via a website portal.

4.3

EXAMPLES OF STATE AND TERRITORY INITIATIVES

State and territory health departments have made a range of advances in driving and supporting patient-centred care. In some states and territories, these initiatives form part of broader consumer engagement frameworks designed to involve a variety of people at all levels of health care. In other states and territories, patient engagement is implemented separately to citizen engagement. Some examples of state and territory-based frameworks and patient-centred initiatives are described below.

ACT Health

In 2003, ACT Health established the Consumer Feedback Project. This involved comprehensive consultation with the community and resulted in the report *Listening and Learning, ACT Health's Consumer Feedback Standards*. ¹⁹² The standards encourage consumers to provide feedback, both positive and negative, on their health experience. The importance of the standards was reinforced with the release of ACT Health's *Consumer Feedback Management Policy*. ¹⁹³ in 2008. The policy outlines the general principles and approach of all ACT Health staff to manage consumer feedback.

Patient experience and redesign programs are currently underway in the Australian Capital Territory. More than 200 patients and carers across a wide range of clinical settings have told their stories through these programs. This information is used for planning and designing new facilities, redesigning processes, and identifying gaps and barriers to healthcare delivery. ACT Health is developing a framework for partnering with patients in all aspects of the redesign program.

Information from the patient experience and redesign programs is a valuable educational tool. Staff are directly involved in documenting the information received from interviews, and the patient experience reports are provided to senior staff and executives. ACT Health also partners with the University of Canberra to educate nursing students about patient experiences

and the importance of a patient-centred approach to care delivery.

Since 2006, ACT Health has implemented the Respecting Patient Choices Program, a national advance care planning program developed by Austin Health in Victoria. The Respecting Patient Choices Program enables individuals to discuss and document decisions about their future health care in preparation for a time when they are unable to participate in their medical care decisions. There is significant community support for this program, and the program coordinator has built networks within the residential aged care sector and other groups to increase community awareness of advance care planning.

In early 2011, ACT Health released the discussion paper *Towards a Patient and Family-Centred Model of Care in the ACT*⁹⁴ for consultation, and is also exploring options for piloting this work.

NSW Health

The NSW Health Caring Together: Health Action Plan¹⁸⁴ was developed in response to the New South Wales inquiries, and focuses on ensuring the patient is the centre of the healthcare system.

The action plan includes strategies to improve communication with patients, such as identifying categories of staff and types of uniforms worn, and developing patient fact sheets to support care in emergency departments. New emergency department nurse positions that focus on the needs of the patients in the waiting room have been established. These nurses are providing better communication on waiting times, initiating basic treatments and completing required admissions documentation.¹⁸⁴

The action plan also led to the establishment of a range of programs that support patient-centred approaches to care. The Take the Lead project is designed to further facilitate the roles of the nursing unit manager and midwifery unit manager in improving the patient journey and carer experience. Some health services in New South Wales have also implemented Studer Group training and programs. The Essentials of Care Program also has a patient-centred focus, and aims to encourage patient participation in decisions about their care, focusing on patients' needs and their experiences of care provided, and encouraging the appreciation of the value of contributions from all involved in care.

NSW Health has implemented a variety of other patient-centred initiatives, including two experience-based co-design programs in hospital emergency departments and departments that receive patients from the emergency departments. 94, 195-196 Evaluations of these programs revealed that co-design was successful in:

» enabling frontline staff to better appreciate the impact of healthcare practices and environments on patients and carers

- » engaging consumers in 'deliberative' or in-depth processes that were qualitatively different from conventional consultation and feedback
- » achieving practical solutions that realise the wishes, advice and insights of consumers and frontline staff
- » enabling staff to reflect on and improve their work practices
- » involving consumers in developing or reviewing clinical pathways, leading to better experiences and patient flow.^{94, 195-196}

NT Department of Health and Families

The Northern Territory has implemented a range of strategies that facilitate patient-centred approaches to care, including conducting patient surveys and engaging patients in the evaluation of services. In addition, families and carers are involved in the development of patient pathways and service models.

All Northern Territory hospitals have boards with consumer representation, and consumers are included in reference and advisory groups. In addition, the Northern Territory has commenced implementation of shared electronic records enabling health information to follow patients as they move between healthcare providers, improve the quality of care in emergency presentations, reduce duplication of diagnostic tests and improve medication management.

Specific projects to improve patient-centred care with Aboriginal and Torres Strait Islander patients have included:

- » placing Aboriginal Community Workers in remote Northern Territory communities, helping to make communication between health professionals and patients easier.
- » expanding mobile dental services through the establishment of a new vehicle based in Alice Springs, which will provide dental services to approximately 8000 people in surrounding communities.
- » developing the Indigenous Renal Palliative Care Pathways, in conjunction with consumers and their families, to facilitate use of and engagement with renal palliative care processes for Aboriginal and Torres Strait Islander patients, particularly those living within remote communities.

In addition, a range of resources tailored to the needs and preferences of Aboriginal and Torres Strait Islander peoples have been developed, including:

- » Talking Posters which, through embedded microchips, enable the provision of health and related information in local Aboriginal dialect.
- » Looking from a new perspective to create a brighter future. A DVD developed and designed to help health services staff understand Aboriginal and Torres Strait Islander peoples' culture and needs, and improve methods of communication and service delivery.

» Waiting. Going to hospital. Coming home strong. A DVD and flip chart designed to help health professionals communicate better with Aboriginal and Torres Strait Islander patients, and to reduce misconceptions that Aboriginal and Torres Strait Islander peoples have about hospitals, and to help break down communication barriers between patients and medical staff. The DVD aims to help inform patients about what to expect, their rights and responsibilities, and the services that they are entitled to during their stay in hospital.

Queensland Health

Queensland Health acknowledges that measuring and reporting on the quality of care from a patient's perspective is essential to providing a high-quality healthcare system.

The Queensland Health strategy for improving the patient experience is informed by collecting consumer feedback through surveys, collecting patient stories and using an enterprise-wide computer-based information system to record and manage complaints and compliments. This information allows Queensland Health to assess performance along the patient journey at the service delivery level and to determine where it may be appropriate to concentrate improvement activities. It also allows the impact of these improvement activities to be measured over time. Management action plans developed in consultation with local health community councils identify areas and activities for improvement, and statewide trend analysis informs organisational strategic planning.

The Productive Ward and Transforming Care programs are key quality and safety improvement initiatives sponsored by the Patient Safety and Quality Improvement Service. Both focus on patient-centred care as a core pillar of their improvement strategy. The programs promote practice that improves the amount of time nurses spend delivering direct patient care (typically doubled). The programs also provide strategies to improve the quality of patient-centred care and have established links with priority state and national safety initiatives. These include clinical handover, malnutrition, patient identification, recognition and management of the deteriorating patient, and falls and pressure ulcer prevention. Both programs have developed patient satisfaction and patient experience assessments to measure improvement in these initiatives. The programs are spreading across all health service districts in Queensland.

Queensland Health actively engages consumers in service planning, design implementation and assessment, from executive committee membership to consultation in the assessment of survey outcomes. Health Consumers Queensland was established in 2008 to contribute to the continued development and reform of health systems and services in Queensland.

It provides the Minister for Health with information and advice from a consumer perspective, and supports and promotes consumer engagement and advocacy.

SA Health

In 2009, SA Health released its Consumer and Community Participation Policy Directive¹⁹⁷ and Consumer and Community Participation Guideline.¹⁹⁸ These aim to ensure consumers and the community have the ability to participate in healthcare decisions, and to support managers and staff to develop a culturally appropriate organisation that is responsive to the perspectives and needs of a diverse range of consumers and community members. SA Health is reviewing these documents in collaboration with the Health Consumers Alliance of South Australia, and will relocate them under a broader, patient-centred care framework along with other relevant policies and strategies.

SA Health's policies, plans and strategies have benefited from consumer input through the many consumer advisory boards and committees, and consumer representatives on clinical networks and advisory structures.

In South Australia, efforts to improve the experience of patients are also informed by patient surveys and computer-based information systems. The Patient Evaluation of Hospital Services survey collects information on patient satisfaction. The new South Australian Consumer Experience Surveillance System is focused on measuring the patient experience as part of a more deliberate strategy to focus on patient-centred care. The Safety Learning System is a new computer-based information and reporting program that enables clinicians and staff to more systematically and accurately collect information (including consumer feedback and complaints), and report on adverse events to target areas for change and to provide data for measuring improvements in health practice.

Tasmanian Department of Health and Human Services

In 2010, the Tasmanian Department of Health and Human Services developed the consumer, carer and community engagement framework *Your Care, Your Say,* which encourages consumers, carers and the community to take part in decision making. The framework establishes four key objectives for the department:

- » No wrong door: Making sure you can access the services you need.
- » Listening and responding: Involving you in developing responsive, accessible and sustainable health care services.
- » Capacity building: Making sure service providers have the skills, knowledge and know-how to practise engagement meaningfully.

» Relationship building: Developing trust and understanding to develop healthier communities.

Implementation of the action plan is currently being progressed.

One of the initiatives currently being undertaken focuses on improving health literacy in the effort to help patients understand their condition, which facilitates better decision making through informed consent. The project has two objectives:

- » improving the capability of the department in providing health information to clients in the most efficient and easily understandable way possible
- » developing strategies to improve the health literacy among the wider community to assist the broader effort of improving the health outcomes for Tasmanians.

In April 2011, the department held a patient-centred care statewide video conference with the aim of improving the understanding of the patient's perspective and facilitating cross-professional learning and information sharing. The conference focused on concepts of patient-centred care, making progress on patient-centred care in Australia, and examples of state and territory initiatives.

Consumers have been recruited to all clinical networks. The networks encourage the exchange of information and collaboration between clinicians and consumers in order to coordinate and plan for the efficient use of resources. There are four clinical networks in existence, with another three under development and a further three identified for development in the future.

Victorian Department of Health

In 2006, the Victorian Department of Health developed the consumer participation in health policy *Doing it With Us Not For Us.* ¹⁹⁹ The policy encompasses consumer engagement at the individual service, program, organisational and system levels. For the period 2010–13, a new strategic direction for this policy has been developed. ²⁰⁰ New standards, indicators and targets to facilitate engagement at all levels of health care have been developed in collaboration with health services, consumers, carers and community members, and will be applicable to all public health sectors in Victoria.

The Victorian Mental Health Reform Strategy 2009–19 and accompanying action plan and quality framework, developed in partnership with patients and carers, ²⁰¹ promotes mechanisms to support patient-centred care in mental health services.

The Victorian Department of Health currently has several patient-centred projects and initiatives in place. For example:

» A series of 'productive ward' projects developed by the UK NHS are currently being piloted. The projects aim to increase safety, patient complaints, length of stay,

- staff morale and retention, and the time staff spend caring for patients.
- » In 2008, a stroke project aimed at improving the care pathway for stroke patients and their carers was developed and implemented in consultation with patients and carers.
- » A paediatric palliative care parent reference group is being established to ensure the needs and views of parents are included in the government's policy for paediatric palliative services.

Several health services have conducted process-redesign projects as part of the broader Improving the Patient Experience Program. Under this program, emergency departments in Victoria upgraded their physical amenities and reviewed their internal environment. Three complementary activities to improve communication were initiated, including consistent patient-friendly signage improvements, consistent consumer-focused information materials, and communication workshops for frontline emergency department staff. Workshops were also held to improve clinicians' understanding of the patient's perspective of emergency department care and consumer communication needs.

As part of the Victorian Department of Health's Consumer Participation and Information Program, the Health Issues Centre is funded to support and promote consumer participation. Similarly, the Centre for Health Communication and Participation is funded to identify evidence-based improvements to clinical practice, focusing on consumer interactions with healthcare professionals, services and researchers. Links to these organisations can be found in Appendix C.

The Victorian Quality Council has also introduced an initiative to improve the patient journey. This initiative aims to promote the principles of patient-centred care across the Victorian health sector, improve and standardise information flow, and provide decision support tools to assist interfacility transfer of noncritical patients.

WA Health

In 2007, Western Australia adopted a *Consumer, Carer* and *Community Engagement Framework*²⁰² to support area health services and WA Health staff and managers to effectively engage with health consumers across the state. The intent of the framework is to support staff to develop and evaluate productive consumer engagement strategies that promote quality and safety in health care.

The accountability and reporting of consumer, carer and community engagement activity is incorporated in the WA Strategic Plan for Safety and Quality in Health Care 2008 – 2013 Placing Patients First. ²⁰³ Specific strategies incorporated in the strategic plan to enhance patient-centred health care include ensuring consumers, carers

and the community are well informed, improving communication between patients and healthcare providers, and facilitating the involvement of consumers and community members in the planning and delivery of health care.

Examples of WA Health programs underpinned by a patient-centred focus include the Patient First Program, described in case study 6, and the Four Hour Rule Program. The Four Hour Rule Program commenced in 2009 and is an extensive redesign program aimed at improving the patient experience across the hospital journey from emergency admission to ward discharge. As part of the implementation of this program, patient experiences were gathered through 'Voice of the Patient' interviews to ensure that patient feedback about their care experiences was incorporated in the program.

In addition to patient-centred programs, all WA Health hospitals have procedures in place for managing both complaints and feedback provided by healthcare consumers. These procedures are underpinned by the *Western Australian Health Complaint Management Policy 2009*, ²⁰⁴ which recognises the role of complaints and feedback in highlighting areas of improvement to promote safe quality health care.

WA Health has also measured patient satisfaction since 1997. The annual Patient Evaluation of Health Services involves interviews with over 7000 health consumers and measures their satisfaction across a range of domains including information and communication, and the right to be involved in care and treatment. Results gleaned from the evaluation are used to fulfil quality assurance commitments and inform service improvement initiatives.

In 2009, the inaugural Australian Patients for Patient Safety Workshop was held in Perth. The workshop was a collaboration between the WHO Alliance for Patient Safety, the London, Chicago and Australian networks for Patients for Patient Safety, the Health Consumers Council WA, and the Office of Safety and Quality of the WA Department of Health.

The workshop brought together a group of 40 health consumers—many of whom had experienced medical error or health system failure—and healthcare providers and health policy makers from around Australia. The workshop resulted in *The Perth Declaration for Patient Safety*, which builds on the London Declaration drafted at a similar workshop in London in 2005. The London Declaration is a pledge of partnership between consumers and providers. The Perth Declaration for Patient Safety continues and aligns with this pledge of partnership, forming an Australian community of committed people who seek to ensure that future health users are not harmed. A copy of *The Perth Declaration for Patient Safety* is in Appendix F.

4.4

CASE STUDIES OF CURRENT PATIENT-CENTRED CARE ACTIVITIES IN AUSTRALIA

The following case studies demonstrate examples of patient-centred programs and activities currently being undertaken in Australia.

CASE STUDY 3: CALVARY HEALTH CARE SIMPLY BETTER PROGRAM

In 2006, staff at Calvary Health Care—a mixed public and private service in the Australian Capital Territory—reviewed their patient care experience results and identified a need for significant improvement. Analysis of surveys conducted through private vendor Press Ganey highlighted that patient satisfaction had fallen below the 10th percentile and that staff satisfaction was similarly poor. Assessment by Better Practice Australia identified a 'culture of blame' in the service.

In recognition of the link between patient satisfaction and staff satisfaction, Calvary launched the Simply Better Program. The program used the Studer Group model to develop strategies to improve both the care experience of patients and the work environment experience of staff. Targets for performance improvement were established in both areas at the outset. Importantly, the Simply Better Program was linked to Calvary's strategic plan, organisational mission, values and service accreditation processes. The program:

- » obtained commitment from the executive team and departmental managers
- » developed measurable goals and performance indicators (e.g. preventable harm of patients and unplanned employee leave)
- » built a service culture by training and coaching staff, and regularly providing staff with patient feedback
- » provided managers with executive leadership training
- » focused on staff satisfaction, including staff rounding, to engage staff in the improvement processes
- » built accountability by reviewing measurable goals and meetings with line managers to identify barriers and enablers
- » aligned behaviours with goals and values
- » provided open communication about strategies and program orientation for new staff
- » instituted a process of formal recognition of staff excellence.

Specific strategies to improve patient care experience included rounding on patients and using patient whiteboards. Communication skills were improved through discharge calls and use of the AIDET tool (Acknowledge, Introduce, Duration, Explanation, Thank you) for all patient communication.

Calvary Health Care considers the Simply Better Program an organisational success, as witnessed through a range of improved outcomes. Improvements in patient care experience data occurred across the organisation, including the public mental health hospital (increased from the 69th to the 99th percentile), public day surgery (from the 56th to the 74th percentile) and private day surgery (from the 47th to the 88th percentile). In three years, employee satisfaction significantly improved from a 'culture of blame' to one of 'reaction/consolidation'. Following the implementation of the Simply Better Program, Calvary achieved its best ever accreditation survey results and the program was awarded an 'outstanding achievement' during the accreditation. Other health services across the country have liaised with Calvary and are now adopting similar strategies.

In 2009, Calvary Health Care's Simply Better Program was awarded the Overall Winner of the ACT Quality in Healthcare Awards (Naree Stanton, Support Services Manager, Calvary Hospital, Canberra, ACT, pers comm, 2009). 205

CASE STUDY 4: FLINDERS MEDICAL CENTRE EMERGENCY DEPARTMENT REDESIGN PROJECT

Flinders Medical Centre established the Redesigning Care Program in 2003. The primary objective of this program was to address the increasing demand for emergency services in the hospital. The program used 'lean thinking' (ensuring that every step of a process adds value and serves the customer's needs) to redesign hospital processes across the entire spectrum of clinical care.

Redesigning Care aimed to redesign patient flow and eliminate duplication and delays to ensure that each step in the process added value and improved outcomes for patients and staff.¹⁹⁵ The program focused on the patient journey from admission to discharge, and one of its basic rules was to 'see things through the patient's eyes'. The work at Flinders Medical Centre has been undertaken in the emergency department, under the banner of experience-based design (EBD) based on Bate and Robert.¹⁴⁵

In 2007, the Flinders Medical Centre Redesigning Care Program designed and implemented an approach to capture patient, carer and staff experience in the emergency department. The aim was to understand and improve patient, carer and staff experience when accessing the centre's emergency department and to trial the EBD methodology as a sustainable approach to patient and carer participation within the centre's management system. ¹⁹⁵ In 2007–08, data were generated through observations of 29 staff interviews, 22 patient and carer real-time experiences, and 14 patient and carer reflective interviews.

Some of the solutions for action related to recommendations for physical changes, and resulted in a capital works redevelopment of the centre's emergency department to increase patients' comfort and improve their environment. For example, a physical redevelopment reduced noise levels and improved visual management by staff. The main solutions outside the redevelopment involved communication processes and teamwork. The program was considered successful in improving teamwork and communication between staff and patients.⁹⁴

CASE STUDY 5: QUALITY FRAMEWORK FOR AUSTRALIAN GENERAL PRACTICE

To support their work to improve quality, the Royal Australian College of General Practitioners has developed a quality framework for general practice. The framework can be used to plan or evaluate quality initiatives, map the current activities, clarify roles and responsibilities and stimulate broader thinking about quality. It includes six domains (professionalism, knowledge and information management, competence, capacity, financing and patient focus) and four levels of analysis and action (national, regional, setting of care and consultation) within each domain.

The patient focus domain is based on the wishes of patients and consumers to fully participate in improving quality at all levels²⁰⁷, in particular as:²⁰⁷

- » active partners in the consultation
- » customers in the setting of care
- » the population of a region whose collective health is dependent on the region's services
- » consumers, community members and taxpayers who have a legitimate stake in the health systems of the country.

Examples of activities included in the patient focus domain of the quality framework include: 206-207

- » ensuring a practice is accessible
- » using a patient-centred approach during consultations
- » collecting feedback from patients
- » involving carers and consumers in regional and jurisdictional projects, programs and policy development
- » ensuring that health inequalities are addressed by activities at all levels of action
- » encouraging national advocacy by and for carers and consumers.

CASE STUDY 6: PATIENT FIRST PROGRAM

WA Health has developed the Patient First Program, ²⁰⁸ which places patients and their carers at the forefront of clinical processes by increasing patients' understanding of their condition. This facilitates better decision making through informed consent.

The program aims to increase patients' health literacy and facilitate a quicker recovery time. Increasing patients' awareness of the risks inherent in their health care was seen as a way to mitigate those risks, while also giving patients the ability to self-manage their health issues.

The program involved volunteer consumers ('Patient First Ambassadors') distributing a Patient First booklet directly to patients. The booklet provides information for patients on topics such as informed consent, issues to consider when making decisions about treatment, understanding the risks associated with treatment and procedures, patients' rights and responsibilities, medication safety, falls prevention, avoiding infection, improving emotional wellbeing, maximising recovery, health records, and information and privacy.²⁰⁹

Using a Patient First Ambassador to distribute the booklets improved communication with patients; the idea of one consumer talking to another consumer was a powerful tool. The ambassador was not affiliated with the hospital or health department, but identified with the health consumer, and was therefore able to discuss the topics in the booklet in a nonthreatening way.²⁰⁸

CASE STUDY 7: RESPECTING PATIENT CHOICES PROGRAM

The Respecting Patient Choices Program is an advanced care planning program that encourages patients to have input in their current and future health care. The program was originally developed in the US, in the context of advances in medical technology that gave medicine the ability to prolong life through artificial or mechanical means. In this environment, healthcare professionals and families can often find themselves having to decide when to withhold or withdraw lifesustaining treatments, because the patient can no longer communicate their own decisions.

The program has developed a systematic approach to advanced care planning. This approach supports the integration of the program into routine health care, and ensures that patient preferences are known and respected. The quality-assured system of discussing and documenting a patient's preferences for their future health care prepares for a time when patients might not be able to make decisions about their end-of-life care. This can be done in consultation with health providers, family members and other important people in patients' lives, helping to ensure that future medical care is in line with the patient's wishes and views.

The Respecting Patient Choices Program was adapted by Melbourne's Austin Hospital in 2002–03, and has now been taken up by many other healthcare organisations within Australia. It was trialled as a community

extension program in a number of residential and community aged and palliative care facilities in 2004–05. An evaluation of this community extension program found that advanced care planning had significant impact on the level to which consumers were involved in choices regarding their future care, and on the skill and confidence of health service providers in attending to patients' wishes.²¹⁰

CASE STUDY 8: PEER SERVICE IN MENTAL HEALTH PROJECT

The Southern Mental Health Peer Service was established in southern Adelaide as a three-month pilot project in peer support, and operated from 2006 to 2008. 211 The consumer-driven and operated project was funded and supported by Southern Mental Health, Flinders University and SA Health's Mental Health Unit. Metro Home Link, a state government funded service, was also involved in the project.

Initially, a group of consumers of mental health services, mental health managers, nongovernment representatives and general practice staff came together to discuss consumer needs and priorities, and how these might be better met. Consumers identified the first days after leaving hospital as significant. They reported feeling that limited support was available at a time when they were vulnerable to relapse. Consequently, Southern Mental Health Peer Service was established with the goal of early discharge and future hospital avoidance.

Peer support workers who had experience and training in support work, and had experienced a mental health condition that required hospitalisation, were employed by the service. Workers were paid, supported by a coordinator and participated in a six-week specific peer worker course. Over the course of the project, they partnered with peer volunteers interested in becoming peer support workers.

Following consumer consent, referrals were made through a statewide call centre by GPs, consumers, carers, staff from inpatient wards and emergency departments, short stay units and hospital in the home services.

Over the first 12 months, the project provided support to more than 300 people, with referrals from the full range of service providers across the southern half of Adelaide as well as from some country regions.

It was concluded that using peers to provide support to consumers at this stage of their recovery seemed highly effective, had personal benefit to consumers and peers, provided savings to the system in terms of bed-days, and had the potential for encouraging mental health service culture and practice towards a greater recovery focus.

It was further concluded that with adequate support, training and supervision, people with mental illness who are living well can provide meaningful emotional and practical support to other consumers.

The program and its impact has informed other government initiatives, such as the Personal Helpers and Mentors initiative, managed through the Australian Government Department of Families, Housing, Community Services and Indigenous Affairs.



Making progress on patient-centred care in Australia



This section describes a range of recommendations and linked strategies that healthcare organisations could consider in their approaches to implementing patient-centred care.

Many healthcare professionals and organisations are seeking to provide care that reflects patient-centred principles; however, there is little Australian guidance on how to achieve this. Like any activity that seeks to improve quality, patient-centred care needs to be integrated into existing systems and processes at the policy, organisational and service delivery level.

Current and emerging policy points to the need for healthcare services to be patient-centred, but the services can often find it difficult to transform care delivery. Most services can readily put patient charters and informed consent policies in place, but many also find it hard to actively change the way care is delivered, and may struggle to engage patients and learn from their experience. A range of Australian organisations are currently implementing projects that focus on different aspects of patient-centred care and are achieving positive outcomes. However, there is little guidance on how to implement patient-centred care in health services in Australia in a systematic way.

This section highlights strategies and makes recommendations that practitioners, managers and policy makers can consider using to build partnerships with patients, families, carers and consumers. Sections 5.1, 5.2 and 5.3 consider the key issues of redefining quality in health care, performance monitoring, education and research, and provide systemoriented recommendations. Section 5.4 includes serviceoriented recommendations that healthcare organisations should consider when looking to design systems that are patient-centred. Organisations need to consider how these recommendations might be applied within their own context, as each organisation is unique and will have its own set of challenges and variables. Checklists to help organisations assess their readiness to implement patient-centred care are presented in Appendix A.

5.1 REFOCUSING THE WAY WE LOOK AT OUALITY IN HEALTH CARE

Patient-centred care requires a change in the way policy makers and regulators think about the quality of health care. The traditional approach to health care focuses on clinical, therapeutic and diagnostic effectiveness and cost-effectiveness as measures of health outcomes. ¹⁰ In contrast, patient-centred care takes a broader view. For example, the *Next Stage Review* ⁵⁹ defines quality as consisting of patient safety, clinical effectiveness and patients' experience.

In the US, the Institute of Medicine report identified patient-centred care as one of six quality aims for improving care.² Berwick¹⁵⁰ states that incorporating patient-centred care into health care as we currently know it will involve some radical, unfamiliar and disruptive shifts in control and power, out of the hands of those who give care and into the hands of those who receive it. The starting point for change will therefore be affirming patient-centred care as a dimension of quality in its own right, and not just through its effect on health status and outcomes.¹⁵⁰

In Australia, it is recognised that the traditional model of patient safety, drawn from high-risk industries, is not patient-centred enough to be used as a comprehensive approach to improving safety and quality. Patient-centred care is now considered to be an integral dimension of quality health care in Australia; the Australian Safety and Quality Framework for Health Care positions patient-centred care as one of three dimensions of safety and quality (see Section 3.3). Various state and territory safety and quality frameworks acknowledge the importance of patient-centred concepts as a dimension of quality, although they do not generally define patient-centred care as a dimension of quality in its own right.

Recommendation 1

Policy makers and regulators should include patient-centred care as a dimension of quality in its own right in strategic and other policy documentation.

5.2

PERFORMANCE MONITORING AND MEASUREMENT

All Australian states and territories are increasing their activity in measuring healthcare quality, especially patient experiences. Most jurisdictions collect a variety of patient feedback, including national and local surveys, complaints data and web-based feedback.

Patient surveys are a commonly used method of collecting patient care experience data. Survey data is useful for monitoring care because it identifies trends and provides an overall snapshot of patient experience across the system. However, it should not be viewed as the only method; a variety of other methods for collecting patient experience data can provide a deeper level of understanding at the individual care and service levels.

Most state and territory regulators have performance improvement divisions in place that seek to facilitate the implementation of patient-centred care strategies. For example, the collection of patient experience data via patient interviews, and redesign and co-design projects has seen the implementation of significant changes to service delivery that have been driven by both patients and staff.213

Regulators and policy makers should continue to facilitate the implementation of a wide variety of patient experience methodologies to augment patient survey data. Information on additional ways to source patient care experience data can be found at Section 2.4.

In the US, the Institute for Patient- and Family-Centered Care (IPFCC)¹⁵⁵ states that processes and outcomes associated with quality, safety and experience of care should be measured to embed patient-centred care in the health system. They also suggest that there is a need to link funding and accreditation to patient satisfaction measurements, and to assure accountability by requiring accurate public reporting.155

According to the International Alliance of Patients' Organizations,10 the traditional approach to defining quality has lacked the necessary concepts of patientcentred care because there are no indicators or criteria to determine the level of patient-centredness in health care. Where indicators do exist, they are not standardised across service providers or jurisdictions, making the necessary comparisons and benchmarking needed to improve service delivery difficult to achieve.

In Australia, approaches to the measurement of safety and quality in health care are increasingly being standardised across the country. Patient care experience is an essential part of safety and quality measurement, 214 and there is a need, over the medium to short term, to move towards standardised measures of patient care experience and indicators that can be used at service and policy levels nationally, across states and territories, and within health regions.

Currently, different organisations, jurisdictions and healthcare sectors use different survey tools, preventing the analysis of national-level data. To standardise patient experience data collection, the data collection tools should address the principles of patient- and family-centred care, including responsiveness, respect, information sharing and collaboration. Working towards national agreement on a set of core items that could be included in patient experience surveys would facilitate analysis of national-level data.

Steps are being made in this direction. The proposed new national-level Performance and Accountability Framework indicators are likely to measure traditional operational and clinical performance indicators, as well as patient-reported quality based on the patient experience.159 These performance indicators are expected to facilitate transparent monitoring across services and jurisdictions, setting a minimum standard of healthcare quality to which each provider should adhere. 159

healthcare settings.

Recommendation 3

Patient surveys used to assess patient care experience need to include questions specifically addressing recognised patient-centred care domains and assess more than patient 'satisfaction'.

Recommendation 4

'Improving patient care experience' should be included as an indicator of quality and reflected in healthcare reporting and funding models.

Australia is also improving transparency. State and territory health departments are increasingly reporting patient survey and experience data, with survey results publicly available on departmental websites. In 2010, the New South Wales Bureau of Health Information 186 released a detailed report of patient care experience data for overnight and day-only patients. Data on specific area health services and individual hospital performance were made publicly available.

Some international initiatives may be useful models to improve transparent data reporting in Australia. For example, in England, individual stories of patient and family perceptions of care complement survey data (see Section 2.1). In the US, the Department of Health and Human Services publicly reports all hospital data for patient care experience, processes of care (e.g. infection prevention), and outcomes of medical conditions and surgical conditions (e.g. mortality, surgical readmission rates) through the Hospital Compare website.

Recommendation 5

To improve transparency, Australian policy makers and regulators should make data regarding patient care experience in health services publicly available via websites.

5.3

EDUCATION AND RESEARCH

As was noted in Part 1.5, the focus of patient-centred care is being able to take account of patients' desire for information and involvement in decision making by providing a supportive and collaborative environment.

Changes in the expectations of patients and consumers, and the dynamics of relationships between health professionals, patients, families and carers can have significant implications for the education and professional development needs of healthcare professionals.²¹⁵ To promote an approach to health care delivery in which power and decision making are shared

and face-to-face relationships are more equal, healthcare professionals need to be equipped with new skills. This may require further education and training.

Efforts to break down professional boundaries through education and training provide opportunities for the re-engineering of professional roles in ways that offer the chance to draw on interdisciplinary models of care.²¹⁵

This approach is beginning to be adopted internationally. In Canada, the Patient-Centered Care: Better Training for Better Collaboration program, funded by Health Canada, developed the theoretical and practical basis required for healthcare professionals to work more collaboratively to ensure effective patient-centred practice. An evaluation of the program found that although the implementation of an interfaculty training curriculum on interprofessional collaborative practice is challenging in many ways, it offers a true opportunity to prepare future health professionals for contemporary patient-centred practice. ²¹⁶

If patient-centred care is to be integrated into the way the health system does business, then there is a need to ensure that healthcare professionals receive training in both the principles and application of patient-centred care in parallel to their clinical training.

Recommendation 6:

Patient-centred care should be a component of undergraduate and postgraduate education programs for all health professionals.

A English study²¹⁷ exploring the extent to which doctors are being trained to work in partnership with patients found that patient partnership is becoming more prominent on the agenda in medical education.

Patient partnership appears in codes, statements of principle and some of the new curricula, and methods of assessing the relevant competencies are being developed. Despite promising developments in some medical schools and certain postgraduate training programs, the study found skill gaps, a general lack of awareness of needs, and few examples of good practice.²¹⁷

The study identified a range of barriers to change, including knowledge of how best to teach skills for partnering with patients. Hasman and Coulter²¹⁷ suggest that to address this, and other challenges to partnering with patients, greater use should be made of patients as teachers. Both real and simulated patients have a potentially useful part to play in clinical education, and there is considerable scope for extending and developing their role.

Recommendation 7:

Education programs should engage patients and families as teachers and collaborators, rather than solely as cases to be studied.

Much of the research and literature on patient-centred care originates from the US or Europe. Australian research has been undertaken on different components of patient-centred care, such as communication, care coordination and methods of consumer and citizen engagement. However, not a great deal is known about the implementation of patient-centred care practices within the Australian healthcare system, including the extent to which patient-centred approaches are practiced and the effectiveness of these approaches. To advance patient-centred care in Australia, further research is needed in this area.²¹⁵

In addition, more general emerging principles and practices for patient-centred care in the Australian context require further research and evaluation to determine best practice. This type of research can provide a basis for the development of tools, models and educational strategies to promote consistent and evidence-based patient-centred practices.

Recommendation 8:

Research funding bodies should acknowledge the importance of patient-centred care to the health system, and this should be reflected in the distribution of funding.

5.4 ORGANISATIONAL STRATEGIES

In order to successfully implement patient-centred care at the organisational level of health care, it is important that it becomes part of the way an organisation does business. Therefore, a systematic approach to integrating the principles of patient-centred care needs to be adopted across an organisation as a whole.

This section considers practical strategies that healthcare service providers can introduce to integrate patient-centred care into their governance framework. It also considers tools for assessing current care delivery. Key characteristics^{15, 152} of organisations that have

successfully transformed care delivery to patientcentred care include:

- » having committed senior leadership
- » engaging patients, family and carers as partners
- » using data to drive change, including regular collection and feedback of patient care experience
- » supporting change responsively
- » building staff capacity and a supportive work environment
- » ensuring accountability at all levels for improving patient care experience
- » fostering an organisational culture that strongly supports learning and improvement.

Each organisation needs to reflect on its current care provision and feedback from patients before deciding how to implement strategies to improve patient-centred care. In addition, efforts should be made to ensure that any strategies or programs that are put in place are evaluated to allow the Australian healthcare community to benefit from the experience of individual healthcare organisations.

Recommendation 9:

Healthcare service executives and managers should ensure that organisational systems and processes are designed to be patient-centred.

The importance of committed senior leadership

Schall and colleagues²¹⁸ state that one of the most important lessons learned when implementing patient-centred care at their facility is that 'leadership matters'. At the organisational level, the way in which leaders set policies and reinforce the importance of improvement greatly affects progress towards improvement. When leadership and staff priorities do not align, it is hard to deliver high-quality services.

Key characteristics of a successful patient-centred care organisation are a strategic vision and mission that clearly articulate the organisation's focus on patients, and leaders who continuously convey this message to all levels of the service. 15, 152

Changing organisational culture from a provider focus to a patient focus can be one of the most difficult barriers to establishing a patient-centred approach to care. Various change management strategies can be used to shift culture over time, including Kotter's Eight Steps:²¹⁹ incorporating creating a vision for change, communicating the vision, removing barriers, creating short-term wins, and anchoring the change in organisational culture. Programs by organisations

such as the Studer Group (see Section 2.5) may help engage staff, particularly those in priority areas for improvement.

Designating a senior manager or executive with responsibility for implementing patient-centred care, and designating champions who model patient-centred behaviours, sends a clear message to an organisation about the importance of this approach.

In their own behaviours and values, leaders set the tone for implementation of patient-centred care. Building credibility as a champion of patient-centred care requires a leader to understand the hospital experience from both patient and staff perspectives. It means nput from staff, poliders at patient-centred hour spending considerable time in the cheduling frequent rounding on all shifts, cone experiences of patients and staff first-hand.

Recommendation 10:

Palthcare service executives and should develop a shared mission that senior confidence in the cheduling frequent rounding on all shifts, cone experiences of patients and staff first-hand. communicating openly, and soliciting and responding

Engaging patients, families and carers as partners

Promoting patient engagement can focus on servicelevel improvement as well as individual care. For example, a service aiming to improve safety by partnering with individual patients may actively engage patients in handovers, ward rounds, medication and incident reviews and team practice review meetings, as well as planning and managing their own care.

Strategies for partnering with patients, families and carers at the service level include:

- » partnering in service redesign and co-design projects
- » engaging patients, families and carers in incident investigation systems and processes
- » establishing and involving patients, families and carers in open disclosure processes following adverse events
- » involving patients, families and carers in educational programs for healthcare professionals and administrative leaders
- » establishing an escalation of care for deteriorating patients
- » establishing a patient liaison office involving patient representatives

- » establishing patient and family advisory councils or committees
- » involving patients, families and carers in key organisational committees such as
 - strategic planning
 - safety and quality improvement
 - medical review
 - risk management
 - facility planning and design
 - staff interview panels and induction
 - information technology
 - ethics and research.

In addition, engaging patients in open disclosure processes following adverse events may not only improve the individual patient experience, but can enable overall system improvement by providing an opportunity for services to assess and review how processes and services may be improved.

Appointing patients to advisory or governing structures is emerging as a successful consumer involvement activity that links the individual experience level of health care to the organisational and systems levels. For example, experience-based co-design in England facilitated provision of patient feedback to the board of governors through patients telling their stories at board meetings. 220 In Canada, Patients For Patient Safety champions are appointed to consumer advisory committees of safety and quality structures, 221 and at the provincial level, patients and families are invited onto patient safety advisory committees at hospitals. In England, the Local Government and Public Involvement in Health Act 2007⁸⁹ empowers local networks to assess the healthcare needs and experiences of patients and citizens, and make recommendations to governors of health services.

Implementing organisational procedures and policies to support these patient engagement strategies will ensure that the strategies are sustainable.152 In this regard, the Commission has developed service-level standards for healthcare providers that outline principles for partnering with patients, families and carers to improve quality of care within services. Ten NSOHS Standards have been developed as part of a model national accreditation scheme that was approved by the Australian Health Ministers' Conference in November 2010. Of these, the 'Partnering with Consumers' Standard requires the effective and meaningful engagement of consumers in organisational planning. The NSQHS Standards provide a framework for a patient-centred service culture by involving consumers in the review, design and implementation of services.

It is important to note that ensuring that individuals have the necessary skills and capacity to engage with health service providers as equal partners is critical to effective engagement. Facilitating access to relevant orientation and training for patients, families, carers

and consumers engaging in partnership with an organisation is an important step in the engagement process. All the standards developed as part of the set of NSQHS Standards need to be considered in the context of partnering with patients.

As with any new strategies to foster patient-centred care, healthcare services should be encouraged to evaluate the effect of these partnerships to help build the evidence base for partnering with patients, families and carers to improve quality of care.

Recommendation 11:

Healthcare service executives and managers should develop and implement policies and procedures for engaging patients, families and carers in their own care.

 $\mu_{H_{1}}$

Recommendation 12:

Healthcare service executives and managers should develop and implement policies and procedures for involving patients, families, carers and consumers at a service level, in policy and program development, quality improvement, patient safety initiatives and healthcare design.

Recommendation 13:

Healthcare service executives and managers should provide support for patients, families and carers involved in governance to develop the necessary skills and capacity required for effective partnerships with their healthcare organisation.

Recommendation 14:

Healthcare service executives and managers should ensure that the service meets the National Safety and Quality Health Service 'Partnering with Consumers' Standard.

Data driving change: using regular collection and feedback of patient care experience

Internationally, few organisations have adequate systems for coordinating patient experience data collection and assessing its quality, or for learning from and acting on the results in a systematic way. ²⁴ To gain a clear picture of patient care, it is important to use a range of sources to collect information about patient experience. Patient surveys that are conducted on a regular basis and reported throughout the organisation, from executive level to the ward, provide staff and management with feedback about care from the service users' perspective.

Patient surveys and complaints data are useful to reveal large-scale trends²⁴ and help those responsible for service planning and governance. Although they are an important foundation, patient experience survey scores can present a limited picture. Detailed information about specific aspects of patients' experiences is likely to be more useful for monitoring performance of hospital departments and wards.¹⁵²

These measures need to be complemented with patients' personal stories, which can have a direct impact on those responsible for care. 94 The stories allow carers to 'see the person in the patient'; they bring patients' experiences, feelings and concerns to life in a way that connects with service providers' own experiences, feelings and concerns. 16, 94 Other methods for collecting care experience information include conducting focus groups, providing patient journals, having patient adviser rounds on wards, using 'mystery shoppers' who report to governance committees, and asking patients to tell their stories at executive meetings.

It is also important to ensure that surveys and other methods of collecting data on patient experience are accessible for, and applicable to, people from diverse and vulnerable populations. This includes the young, the elderly, people with a disability, Aboriginal and Torres Strait Islander peoples, and those from culturally and linguistically diverse backgrounds.

Recommendation 15:

Healthcare service executives and managers should ensure that systems are in place for the regular collection and reporting of patient care experience data through quantitative patient surveys and qualitative, narrative-based sources.

Recommendation 16:

Healthcare service executives and managers should ensure that organisational approaches to quality improvement include feedback about patient care experience — alongside clinical and operational data — when determining health service action plans.

Recommendation 17:

Healthcare service executives and managers should contribute to the evidence base for patient-centred care by recording and publishing changes in key organisational and patient outcome metrics over time.

Supporting improvement of care delivery and environment

 ω_{m}

To successfully establish a patient-centred care approach, organisations need to address changes in response to the areas of need identified through patient feedback, 152 and consult with patient advisers and other relevant experts before deciding on strategies. Surprisingly, in successful patient-centred services, the improvements that patients suggested were not necessarily expensive. 152 Patient advisers in a number of US services are viewed as the force for making health care more affordable.

Examples of responsive changes¹⁵² made by hospitals to improve patient-centred care include:

- » hourly rounding by nurses on wards
- » providing welcoming facilities for families
- » reviewing hospital signage from a patient perspective

- » providing a new style of hospital gown to afford dignity to patients
- » engaging volunteers to act as concierges or patient navigators
- » redesigning waiting areas
- » introducing communication strategies to keep patients and families informed.

Although these examples demonstrate the types of responsive changes that organisations can make, it is not an exhaustive list. Other strategies are presented in Sections 2 and 4, along with resources to help foster a patient-centred approach. Further resources, including checklists, are summarised in the appendices.

Resourcing the improvement of the quality of the physical care environment strongly correlates with improved patient care experience and other health and business outcomes. The Planetree approach in the US links architecture and physical space with healing, and outlines principles for healthcare service design (see Appendix C). When considering new healthcare facilities, or when renovating or reviewing existing facilities, organisations should consider patient-centred principles and seek patient adviser input; for example, through experience-based redesign and co-design strategies (see Section 2.4).

Recommendation 18:

Healthcare service executives and managers should provide organisational support to enable staff to partner with patients and consumers and to implement necessary changes based on that partnership.

Building staff capacity and a supportive work environment

The strategies to support staff highlighted in Section 2, such as staff and practice development, values training, communication skills training and staff satisfaction programs, will also help establish patient-centred care in Australia. This has already begun in some jurisdictions. For example, a Studer Group program to improve communication between staff and patients is underway in New South Wales.

According to Coulter,²²³ the basic competencies required by individual health professionals to be patient-centred include:

- » understanding the patient's perspective, expressing empathy and providing appropriate support
- » guiding patients to appropriate sources of information on health and health care

- » educating patients on how to protect their health and prevent occurrence or recurrence of disease
- » eliciting and taking account of patients' preferences
- » communicating information on risk and probability
- » sharing treatment decisions
- » providing support for self-care and self-management
- » working in multidisciplinary teams
- » managing time effectively.

Patient-centred organisations focus on increasing their staff's skills to support patient-centred care delivery. 152 Strategies to achieve this include:

- » training staff in communication skills, including skills for communicating with patients with specific communication needs
- » adopting communication techniques such as AIDET (Acknowledge, Introduce, Duration, Explanation, Thank you)
- » training all staff in patient-centred values and customer service techniques
- » tailoring the workforce through selection of staff committed to the mission of the organisation
- » integrating discussion of patient-centred values into staff orientation sessions (e.g. the chief executive could open the induction session with a discussion of values)
- » actively involving patients and families in education programs for healthcare professionals, managers and executives
- » holding education sessions for healthcare professionals in which patients and families share their experience of care
- » involving patients and families in educating junior healthcare professionals and in affiliated programs for undergraduates.

Focusing on the work environment for staff is an important component of 'caring for the care givers'. Exemplary patient-centred health services achieve this by visibly celebrating the successes of staff in improving patient care experience, such as public acknowledgment by the chief executive, awards for achievements, or features in annual reports, hospital intranet and newsletters. 152 This approach recognises that the workforce is the healthcare service's most important asset, and aligns with the move towards 'person-centred' organisations in which both patients and staff are valued.15 Person-centred organisations also use feedback from staff surveys on the work environment to improve the work culture and processes. Giving significant attention to staff satisfaction also acknowledges the link between employee satisfaction and patient satisfaction.224

Recomment.

Healthcare service managers should support training and education activate tailored to building the capacity or staff to deliver patient-centred care.

Recommendation 20:

Ithcare service executives and rs should focus on work rt, work culture and raff as an integral rurveys ent and raken work environment.

Accountability at all levels for improving patient-centred care

Successful patient-centred care organisations establish clear lines of accountability for staff at all levels, making each person responsible for improving the patient care experience. 152 Individual accountability can be reinforced through performance reviews. A range of strategies can be put in place to promote staff accountability, including:

- » incorporating responsibility for improving patient care experience in job descriptions
- » explaining at orientation for new staff that they are responsible for patients' experience of care
- » considering patient feedback during staff performance reviews, including sharing patient stories
- » linking promotions or performance bonuses to improving quality indicators, including care experience
- » incorporating patient care experience metrics into unit, departmental and organisational performance monitoring and reporting
- » adopting a motto that embodies accountability, such as 'Every patient—Everyone's responsibility' (Pat Sodomka, Sebrio Vice President, Patient and Family Centered Care, MCG Health Inc. and Director, Center for Patient- and Family-Centered Care, Medical College of Georgia, pers comm, 2009)

- » linking quality metrics, including patient care experience, to performance reviews of organisational governance bodies and chief executives
- » ensuring that the agenda for board or governance committee meetings includes a strong emphasis on reviewing quality issues and quality performance data, including patient care experience. 152, 225

Although a number of options are outlined here, linking improvement of care experience to individual accountability through performance review is a starting point relevant to all staff.

Recommendation 21:

Healthcare service executives and managers should integrate accountability for the care experience of patients into staff performance review processes.

An organisational culture that strongly supports learning and improvement

The leading organisations in patient-centred care have a culture of learning and strongly supporting change and improvement.152 These organisations use many aspects of 'learning organisation' theory and have characteristics described in 'learning culture' frameworks. Learning organisations have systems, mechanisms and processes in place that are used to continually improve the capabilities of those who work with it or for it, to achieve sustainable objectives. 226 They adapt to external environmental forces, constantly improve their own ability to change, promote individual and collective learning, and use these lessons to improve outcomes.

Healthcare organisations can commit to continuous quality improvement using approaches such as Deming's cycle of 'Plan, Do, Study, Act'. Focusing on needs analysis, performance measurement and improvement is crucial. Learning organisations also support a culture that values people, stimulates new ideas, develops teamwork and adopts staff recognition systems. Successful patient-centred organisations often take the next step of linking individual performance accountability to organisational performance.

Learning organisations are also open to learning from past failures. 227 A number of healthcare services have refocused care delivery for patient-centred care in response to tragic events in their facility, such as the death or severe harm of a patient during care. Such tragic events can cause an organisation to examine its own values and processes for promoting safety and quality improvement. Listening to patients and

involving patients and families in improving care delivery is a positive response to these tragedies.

Events at the Dana-Farber Cancer Institute, Boston, US, are well known in the area of 'turning tragedy into a positive outcome'. In 1994, Betsy Lehman, a health journalist from the Boston Globe newspaper, died after she was accidentally given an overdose of cyclophosphamide—a chemotherapy drug she was receiving for breast cancer treatment. The institute began asking patients and families 'how can we do better?' The incoming chief operating officer, James B Conway, organised town hall meetings for patients and was inundated with people providing feedback about improving care delivery. Sixteen years on, the Dana-Farber Cancer Institute is a leader in patientcentred care, with patient advisers involved throughout the organisation.

Learning from others' experiences can be invaluable, and healthcare organisations can link with service providers who have experience of learning from patients and families.

ed care, with ganisation.

ning from others' experience healthcare organisations can link violets who have experience of learning troud families.

Pecommendation 22:

The service executives and wild foster a culture of organisation, presses and organisation, organisat

Thum,

Acronyms and abbreviations

ABS Australian Bureau of Statistics

AHRQ Agency for Healthcare Research

and Quality (United States)

CAHPS Consumer Assessment of Healthcare

Providers and Systems

CMS Centers for Medicare and Medicaid

Services (United States)

the Commission Australian Commission on Safety

and Quality in Health Care

CQUIN Commissioning for Quality

and Innovation (England)

EBD experience-based design

GP general practitioner

H-CAHPS Hospital Consumer Assessment of

Healthcare Providers and Systems

IHI Institute for Healthcare Improvement

(United States)

IHPA Independent Hospital

Pricing Authority

IOM Institute of Medicine (United States)

IPFCC Institute for Patient- and Family-

Centered Care (United States)

MCG Medical College of Georgia

(United States)

NCDS National Chronic Disease Strategy

NHS National Health Service (England)

NSIF National Services

Improvement framework

NSQHS National Safety and Quality Health

Standards Service Standards

OECD Organisation for Economic

Co-operation and Development

P4P pay for performance

WHO World Health Organization

References

- Goodrich J, Cornwell J. Seeing the Person in the Patient: The Point of Care Review Paper. London: The King's Fund, 2008a-63.

 Goodrich J, Cornwell J. Seeing the Person in the Patient: The Point of Care Review Paper. London: The King's Fund, 2008a-63.

 Goodrich J, Cornwell J. Seeing the Person in the Patient: The Point of Care Review Paper. London: The King's Fund, 2008a-63.
 - Institute of Medicine. Crossing the Quality Chasm: A New Health System for the Twenty-first Century. Washington: National Academies Press, 2001.
 - Australian Institute of Health and Welfare. Australia's Health 2010. Australia's health series no.12. Cat. no. AUS 122. Canberra:
 - 4. Davis K, Schoen C, Stremikis K. Mirror, Mirror on the Wall: How the Performance of the US Health Care System Compares Internationally. 2010 Update: The Commonwealth Fund, 2010:1-21.
 - Australian Commission on Safety and Quality in Health Care. Windows into Safety and Quality in Health Care 2009. Sydney: ACSQHC, 2009:1-98.
 - Vincent C, Young M, Phillips A. Why do people sue doctors? A study of patients and relatives taking legal action. Lancet 1994;343:1609-1613.
 - Australian Safety and Quality Framework for Health Care. Australian Commission on Safety and Quality in Health Care, 2010. (Accessed 13 April 2011, at www.safetyandquality.gov.au/internet/ safety/publishing.nsf/Content/NSQF)
 - Carer Recognition Act. Australia, 2010.
 - Institute for Patient- and Family-Centered Care website. Institute for Patient- and Family-Centered Care. (Accessed 7 July 2010, at www.ipfcc.org)
 - International Alliance of Patients' Organizations. What is Patient-Centred Health Care? A Review of Definitions and Principles. Second ed. London: IAPO, 2007:1-34.
 - Gerteis M, Edgman-Levitan S, Daley J, Delbanco T. Through the Patient's Eyes: Understanding and Promoting Patient-Centered Care. San Francisco: Jossey-Bass, 1993.
 - NRC Picker. Eight Dimensions of Patient-Centred Care. NRC Picker, 2008.
 - Frampton S, Guastello S, Brady C, Hale M, Horowitz S, Bennett Smith S, et al. The Patient-Centered Care Improvement Guide. Derby, USA: The Planetree Association and The Picker Institute, 2008.
 - 14. World Health Organization. The World Health Report 2000 Health Systems: Improving Performance. Geneva: World Health Organization, 2000:1-215.
 - Shaller D. Patient-Centered Care: What Does It Take? United States: Picker Institute and The Commonwealth Fund, 2007.

- healthcare. Part 6: keeping the patient front and centre to improve healthcare quality. The New Zealand Medical Journal 2006;119(1242).
- Delbanco T, Berwick D, Boufford J, Edgman-Levitan S, Ollenschlager G, Plamping D, et al. Healthcare in a land called PeoplePower: nothing about me without me. Health Expectations 2001;4(3):144-150.
- 20. Davis K, Schoenbaum SC, Audet AM. A 2020 vision of patientcentered primary care. Journal of General Internal Medicine 2005;20(10):953-957.
- Silow-Carroll S, Alteras T, Stepnick L. Patient-Centered Care for Underserved Populations: Definition and Best Practices. Washington DC: The WK Kellogg Foundation, 2006:1-42.
- 22. Law M, Baptiste S, Mills J. Client-centred practice: what does it mean and does it make a difference? Canadian Journal of Occupational Therapy 2005;62:250-257.
- Slater L. Person-centredness: a concept analysis. Contemporary Nurse 2006;23(1):135-144.
- Goodrich J. Exploring the wide range of terminology used to describe care that is patient-centred. Nursing Times 2009;105(20):14-17.
- Kitwood T. Dementia Reconsidered: The Person Comes First. Buckingham: Open University Press, 1997.
- Bellchambers H, Penning C. Person-centered approach to care (PCA): a philosophy of care and management for carers. Contemporary Nurse 2007;26 (2):196–197.
- US Department of Health and Human Services. Personalized Health Care: Pioneers, Partnerships, Progress. Washington DC: US Department of Health and Human Services, 2008:1-302.
- Conway J, Johnson B, Edgman-Levitan S, Schlucter J, Ford D, Sodomka P, et al. Partnering with Patients and Families to Design a Patient- and Family-Centered Health Care System: A Roadmap for the Future — A Work in Progress. Bethesda, Cambridge: Institute for Patient and Family-Centered Care, Institute for Healthcare Improvement, The Robert Wood Johnson Foundation, 2006.
- In focus: health care institutions are slowly learning to listen to customers. The Commonwealth Fund, 2010. (Accessed 18 August 2010, at www.commonwealthfund.org/Content/ Newsletters/Quality-Matters/2010/February-March-2010. aspx?view=newsletter print)

- Franck L, Callery P. Re-thinking family-centred care across the continuum of children's healthcare. Child: Care, Health and Development 2004;30(3):265–277.
- Beach M, Inui T. Relationship-centered care: a constructive re-framing. Journal of General Internal Medicine 2006;21:3-8.
- Safran D, Miller W, Beckman H. Organizational dimensions of relationship-centered care: theory, evidence and practice. *Journal* of General Internal Medicine 2006;21:9–15.
- Isaac T, Zaslavsky AM, Cleary PD, Landon BE. The relationship between patients' perception of care and measures of hospital quality and safety. Health Services Research 2010;45(4):1024–1040.
- Jha AK, Orav EJ, Zheng J, Epstein AM. Patients' perception of hospital care in the United States. New England Journal of Medicine 2008;359(18):1921–1931.
- Sequist T, Scheider E, Anastario M, Odigie E, Marshall R, Rogers W, et al. Quality monitoring of physicians: linking patients experiences of care to clinical quality and outcomes. *Journal of General Internal Medicine* 2008;23(11):1784–1790.
- Schneider EC, Zaslavsky AM, Landon BE, Lied TR, Sheingold S, Cleary PD. National quality monitoring of Medicare Health Plans. Medical Care 2001;39:1313–1325.
- Arnetz JE, Winblad U, Höglund AT, Lindahl B, Spångberg K, Wallentin L, et al. Is patient involvement during hospitalization for acute myocardial infarction associated with post-discharge treatment outcome? An exploratory study. Health Expectations 2010;13(3):298–311.
- Fremont AM, Cleary PD, Lee Hargraves J, Rowe RM, Jacobson NB, Ayanian JZ. Patient-centered processes of care and long-term outcomes of myocardial infarction. *Journal of General Internal Medicine* 2001;16(12):800–808.
- 39. Glickman SW, Boulding W, Manary M, Staelin R, Roe MT, Wolosin RJ, et al. Patient satisfaction and its relationship with clinical quality and inpatient mortality in acute myocardial infarction. Circulation: Cardiovascular Ouality and Outcomes 2010;3:188–195.
- Meterko M, Wright S, Lin H, Lowy E, Cleary PD. Mortality among patients with acute myocardial infarction: the influences of patient-centered care and evidence-based medicine. *Health* Services Research 2010;45(5p1):1188–1204.
- 41. Boulding W, Glickman S, Manary M, Schulman K, Staelin R. Relationship between patient satisfaction with inpatient care and hospital readmission within 30 days. *American Journal of Managed Care* 2011;17(1):41–48.
- 42. The AHRQ Innovation Exchange: Patient- and family-centered care initiative is associated with high patient satisfaction and positive outcomes for total joint replacement patients. DiGioia, AM. (Accessed 30 May 2010, at www.innovations.ahrq.gov/content.aspx?id=1764)
- DiGioia AI, Greenhouse PK, Levison TJ. Patient and family-centered collaborative care: an orthopaedic model. Clinical Orthopaedics and Related Research 2007;463:13–19.
- 44. Edgcumbe D. Patients' perceptions of hospital cleanliness are correlated with rates of methicillin-resistant *Staphylococcus aureus* bacteraemia. *Journal of Hospital Infection* 2009;71(1):99–101.
- Flach SD, McCoy KD, Vaughn TE, Ward MM, BootsMiller BJ, Doebbeling BN. Does patient-centered care improve provision of preventive services? *Journal of General Internal Medicine* 2004;19(10):1019–1026.
- 46. Stone S. A retrospective evaluation of the impact of the Planetree patient-centered model of care on inpatient quality outcomes. Health Environments Research and Design Journal 2008(summer):55–69.
- 47. Arbuthnott A, Sharpe D. The effect of physician–patient collaboration on patient adherence in non-psychiatric medicine. Patient Education and Counseling 2009;77(1):60–67.

- 48. Weingart SN, Zhu J, Chiappetta L, Stuver SO, Schneider EC, Epstein AM, et al. Hospitalized patients' participation and its impact on quality of care and patient safety. *International Journal for Quality in Health Care* 2011;13(3):269–277.
- 49. Epstein RM, Franks P, Shields CG, Meldrum SC, Miller KN, Campbell TL, et al. Patient-centered communication and diagnostic testing. Annals of Family Medicine 2005;3(5):415–421.
- Little P, Everitt H, Williamson I, Warner G, Moore M, Gould C, et al.
 Observational study of effect of patient centredness and positive approach on outcomes of general practice consultations. BMJ 2001;323(7318):908–911.
- Stewart M, Brown J, Donner A, McWhinney I, Oates J, Weston W, et al. The impact of patient-centered care on outcomes. *Journal of Family Practice* 2000;49(9):796–804.
- Beach MC, Sugarman J, Johnson RL, Arbelaez JJ, Duggan PS, Cooper LA. Do patients treated with dignity report higher satisfaction, adherence, and receipt of preventive care? *Annals of Family Medicine* 2005;3(4):331–338.
- 53. Schneider J, Kaplan SH, Greenfield S, Li W, Wilson IB. Better physician-patient relationships are associated with higher reported adherence to antiretroviral therapy in patients with HIV infection. *Journal of General Internal Medicine* 2004; 19(11):1096–1103.
- Charmel P, Frampton S. Building the business case for patientcentred care. Healthcare Financial Management 2008; March (62 (3)):80–85.
- Sweeney L, Halpert A, Waranoff J. Patient-centered management of complex conditions can reduce costs without shortening life. *American Journal of Managed Care* 2007;13:84–92.
- 56. Hobbs S, Sodomka P. Developing partnerships among patients, families, and staff at the Medical College of Georgia hospital and clinics. *Joint Commission on Accreditation of Healthcare Organizations* 2000;26(5):268–276.
- 57. Institute for Patient- and Family-Centered Care. Advancing the Practice of Patient- and Family-Centered Care: How to Get Started. Bethesda: Institute for Patient- and Family-Centered Care, 2008:1–28.
- National Health Service. Creating a Patient-led NHS: Delivering the NHS Improvement Plan. London: National Health Service, 2005.
- Professor the Lord Darzi of Denham KBE. High Quality Care for All: NHS Next Stage Review Final Report. London: UK Department of Health, 2008:1–92.
- 60. The King's Fund. A High-performing NHS? A Review of Progress 1997–2010. London: The King's Fund, 2010.
- Code of Health and Disability Services Consumers' Rights Regulation. Auckland: Health and Disability Commissioner, 1996.
- New Zealand Ministry of Health. Improving Quality: A Systems Approach for the New Zealand Health and Disability Sector. Wellington: NZ Ministry of Health, 2003.
- 63. Mable A, Marriott J. Sharing the Learning: Health Transition Fund: Synthesis Series: Primary Health Care. Ontario: Health Canada, 2002.
- 64. Herbert D. Changing the culture: interprofessional education for collaborative patient-centred practice in Canada. *Journal of Interprofessional Care* 2005;19(S1):1–4.
- Dagnone T. For Patients' Sake: Patient First Review Commissioner's Report to the Saskatchewan Minister of Health. Regina, Canada: Ministry of Health, 2009:1–68.
- 66. Bill 46, Excellent Care for All Act, Ontario Canada, 2010.
- 67. Arah OA, Westert GP, Hurst J, Klazinga NS. A conceptual framework for the OECD Health Care Quality Indicators Project. *International Journal for Quality in Health Care* 2006;18(suppl. 1):5–13.
- 68. Groene O, Skau JKH, Frølich A. An international review of projects on hospital performance assessment. *International Journal for Quality in Health Care* 2008;20(3):162–171.

- UK Department of Health. Helping the NHS Put Patients at the Heart of Care: The Patient and Public Engagement Support Programme 2009–2010. London: Department of Health, 2009.
- 70. UK Department of Health. Equity and Excellence: Liberating the NHS. London: UK Department of Health, 2010.
- UK Department of Health. NHS Outcomes Framework 2011/12.
 London: UK Department of Health, 2010.
- UK Department of Health. The Operating Framework for the NHS in England 2011/12. London: UK Department of Health, 2010.
- 73. Patient Protection and Affordable Care Act. United States, 2009.
- Measuring healthcare quality. Agency for Healthcare Research and Quality. (Accessed 31 March 2011, at www.ahrq.gov/qual/ measurix.htm#quality)
- Coulter A. Do patients want a choice and does it work? BMJ 2010;341.
- Physician compare website overview information for professionals. Centers for Medicare and Medicaid Services. (Accessed 29 March 2011, at www.cms.gov/physician-compareinitiative/o1 overview.asp)
- 77. Rodriguez HP, von Glahn T, Elliott MN, Rogers WH, Safran DG. The effect of performance-based financial incentives on improving patient care experiences: a statewide evaluation. *Journal of General Internal Medicine* 2010;24(12):1281–1288.
- 78. P4P website. Agency for Healthcare Research and Quality. (Accessed 7 May 2010, at www.ahrq.gov/qual/pay4per.htm)
- Doran T, Fullwood C, Gravelle H, Reeves D, Kontopantelis E, Hiroeh U, et al. Pay-for-performance programs in family practices in the United Kingdom. New England Journal of Medicine 2006; 355(4):375–384.
- Campbell S, Reeves D, Kontopantelis E, Middleton E, Sibbald B, Roland M. Quality of primary care in England with the introduction of pay for performance. New England Journal of Medicine 2007;357(2):181–190.
- 81. Gillam S. Should the Quality and Outcomes Framework be abolished? Yes. *BMJ* 2010;340.
- Alshamsan R, Majeed A, Ashworth M, Car J, Millett C. Impact of pay for performance on inequalities in health care: systematic review. *Journal of Health Services Research and Policy* 2010; 15(3):178–184.
- 83. Commissioning for Quality and Innovation (CQUIN) guidance for 2011/12. UK Department of Health. (Accessed 29 March 2011, at www.london.nhs.uk/webfiles/Information%20Revolution/NHS%20London%202011–12%20CQUIN%20guidance.pdf)
- 84. Tax Relief and Health Care Act. United States, 2006.
- 85. Dunbar J, Reddy P, Beresford B, et al. In the wake of hospital inquiries: impact on staff and safety. *Medical Journal of Australia* 2007;186:80–83.
- 86. Mahdon M. *Public Value and Health*. London: The Work Foundation, 2006.
- 87. Horner L, Lekhi R, Blaug R. Deliberative Democracy and the Role of Public Managers: Executive Summary of the Final Report of the Work Foundation's Public Value Consortium. London: The Work Foundation, 2006.
- 88. Work Foundation. World Class Public Service: Engaging Citizens and Staff. London: The Work Foundation, 2008.
- Local Government and Public Involvement in Health Act. United Kingdom, 2007.
- 90. Health and Social Care Bill 2011. England, 2011.
- Pine J, Gilmore J. The Experience Economy. Boston: Harvard Business School Press, 1999.
- 92. Lee F. If Disney Ran Your Hospital: 9 1/2 Things You Would Do Differently. Bozeman: Second River Healthcare Press, 2004.

- 93. Boyle D, Harris M. The Challenge of Co-production: How Equal Partnerships Between Professionals and the Public are Crucial to Improving Public Services. London: National Endowment for Science, Technology and the Arts, 2009.
- 94. Piper D, Iedema I, Merrick E. Emergency Department Co-design Evaluation Program 1 Stage 2 Final Report to Health Services Performance Improvement Branch. Sydney: Centre for Health Communication, University of Technology Sydney, for NSW Health. 2010.
- 95. Declaration of Alma-Ata. World Health Organization 1978.
- 96. Ridley J, Jones L. User and Public Involvement in Health Services: A Review (Scotland). Edinburgh: NHS Scotland, 2002.
- Gregory J. Conceptualising consumer engagement: a review of the literature. Melbourne: Australian Institute of Health Policy Studies, 2007:69.
- Weale A. What is so good about citizens' involvement in Healthcare? In: Andersson E, Tritter J, Wilson R, editors. Healthy Democracy: The Future of Involvement in Health and Social Care. London: National Centre for Involvement: National Health Service. 2006.
- 99. Coulter A. Patient engagement: why is it important? In: Andersson E, Tritter J, Wilson R, editors. Healthy Democracy: The Future of Involvement in Health and Social Care. London: National Centre for Involvement: National Health Service, 2006.
- 100. O'Brien B, Alderdice A, Hales J. Evaluation of the Effectiveness of the Community Advisory Committees to Boards of Victorian Public Health Services: Final Report. Melbourne: Victorian Government Department of Human Services, 2008.
- Review of evidence for consumer engagement website.
 Picker Institute Europe. (Accessed 15 May 2010, at www.investinengagement.info)
- National Health Service. The NHS Constitution for England. London: UK Department of Health, 2010.
- 103. Patient Rights (Scotland) Bill (SP Bill 42). Scotland, 2011.
- 104. American Hospitals Association. Patient Care Partnership Brochure. Chicago: American Hospitals Association, 2003.
- 105. Centers for Medicare and Medicaid Services. Important Message From Medicare Notice. Baltimore: Centers for Medicare and Medicaid Services, 2007.
- 106. Australian Charter of Healthcare Rights. Commonwealth of Australia, 2008. (Accessed 13 April 2011, at www.safetyandquality.gov.au/internet/safety/publishing.nsf/ Content/PriorityProgram-01)
- 107. ACT Human Rights Act. Australia, 2004.
- 108. South Australian Charter of Health and Community Services Rights. Australia, 2011.
- Victorian Charter of Human Rights and Responsibilities. Australia, 2006.
- 110. UK Department of Health. Understanding What Matters A Guide to Using Patient Feedback to Transform Services. London: Department of Health, 2009.
- Coulter A, Fitzpatrick R, Cornwell J. Measures of Patients' Experience in Hospitals: Purpose, Methods and Uses. London: King's Fund, 2009.
- 112. Nationwide Health and Disability Consumer Advocacy Service. The Art of Great Care. Auckland: Nationwide Health and Disability Consumer Advocacy Service, 2010.
- 113. Implementing Dana-Farber Cancer Institute Patient Safety Rounds in Your Organization — A Toolkit. (Accessed 15 April 2010, at www.dana-farber.org/Adult-Care/Treatment-and-Support/Care-Quality-and-Safety/Patient-Safety-Resources.aspx)
- 114. WA Health. A Toolkit for Collecting and Using Patient Stories for Service Improvement in WA Health. Perth: WA Health, 2008.

- UK Department of Health. Listening, Responding, Improving: A Guide to Better Customer Care. London: UK Department of Health, 2009.
- 116. Watt I. A Review of Strategies to Promote Patient Involvement, A Study to Explore Patient's Views and Attitudes and A Pilot Study to Evaluate the Acceptability of Selected Patient Involvement Strategies. York: Patient Safety Research Group, University of York, 2009.
- Conway J, Federico F, Stewart K, Campbell MJ. Management of Serious Clinical Adverse Events. IHI Innovation Series White Paper. Cambridge, MA: Institute for Healthcare Improvement, 2010.
- 118. Weingart SN, Pagovich O, Sands DZ, Li JM, Aronson MD, Davis RB, et al. What can hospitalized patients tell us about adverse events? Learning from patient-reported incidents. *Journal of General Internal Medicine* 2005;20(830–836).
- 119. Weingart SN, Price J, Duncombe D, Connor M, Sommer K, Conley KA, et al. Patient-reported safety and quality of care in outpatient oncology. *Joint Commission Journal on Patient Safety and Quality* 2007;33(2):83–94.
- 120. Donaldson LJ. Put the patient in the room, always. *Quality and Safety In Health Care* 2008;17(2):82–83.
- 121. Longtin Y, Sax H, Leape LL, Sheridan SE, Donaldson L, Pittet D. Patient participation: current knowledge and applicability to patient safety. Mayo Clinic Proceedings 2010;85(1):53–62.
- 122. Davis RE, Sevdalis N, Vincent CA. Patient involvement in patient safety: how willing are patients to participate? BMJ 2011; 20:108–114.
- 123. Speak Up Program website. (Accessed 12 April 2011, at www.jointcommission.org/speakup.aspx)
- 124. Ten Tips for Safer Health Care. 2006. (Accessed at www.health.gov.au/internet/safety/publishing.nsf/Content/ BE79FB82644728ABCA2571C0000330FB/\$File/10tipsumclbx.pdf)
- 125. Australian Commission on Safety and Quality in Health Care. Open Disclosure Standard. Sydney: Australian Commission on Safety and Quality in Health Care, 2003.
- 126. Canadian Patient Safety Institute. Canadian Disclosure Guidelines. Ontario: Canadian Patient Safety Institute, 2008.
- 127. National Patient Safety Agency. Being Open: Communicating Patient Safety Incidents with Patients, Their Families and Carers. London: National Patient Safety Agency, 2009.
- Gallagher TH, Studdert D, Levinson W. Disclosing harmful medical errors to patients. New England Journal of Medicine 2007;356(26):2713–2719.
- 129. Berkman ND, Sheridan SL, Donahue KE, Halpern DJ, Viera A, Crotty K, et al. Health Literacy Interventions and Outcomes: An Updated Systematic Review. Evidence Report/Technology Assesment No. 199. (Prepared by RTI International–University of North Carolina Evidence-based Practice Center under contract no. 290–2007–10056–I) AHRQ Publication Number 11–E006. Rockville, MD: Agency for Healthcare Research and Quality, 2011.
- Australian Bureau of Statistics. Health Literacy, Australia.
 Canberra: Australian Bureau of Statistics, 2008.
- Rudd RE. Improving Americans' health literacy. New England Journal of Medicine 2010;363(24):2283–2285.
- 132. DeWalt DA, Callahan LF, Hawk VH, Broucksou KA, Hink A, Rudd R, et al. Health Literacy Universal Precautions Toolkit. AHRQ Publication No. 10–0046–EF. Rockville, MD: Agency for Healthcare Research and Quality, 2010.
- 133. Delbanco T, Walker J, Darer JD, Elmore JG, Feldman HJ, Leveille SG, et al. Open notes: doctors and patients signing on. Annals of Internal Medicine 2010;153(2):121–125.
- 134. Bauman A, Fardy H, Harris P. Getting it right: why bother with patient-centred care? *Medical Journal of Australia* 2003;179:253–256.

- 135. Légaré F, Ratté S, Stacey D, Kryworuchko J, Gravel K, Graham I, et al. Interventions for improving the adoption of shared decision making by healthcare professionals (review). Cochrane Database of Systematic Reviews 2010;5(Art. No: CD006732).
- 136. Newman S, Steed L, Mulligan K. Self-management interventions for chronic illness. *Lancet* 2004;364:1523–1537.
- 137. Jordan JE, Briggs AM, Brand CA, Osbourne RH. Enhancing patient engagement in chronic disease self-management support initiatives in Australia: the need for an integrated approach. Medical Journal of Australia 2008;189(10):S9-S13.
- Coulter A, Ellins J. Effectiveness of strategies for informing, educating, and involving patients. BMJ 2007;335(7609):24-27.
- 139. Gerdik C, Vallish RO, Miles K, Godwin SA, Wludyka PS, Panni MK. Successful implementation of a family and patient activated rapid response team in an adult level 1 trauma center. Resuscitation 2010;81:1676–1681.
- 140. Greco M, Carter M, Powell R, Sweeney K, Jollife J, Stead J. Impact of patient involvement in general practice. Education for Primary Care 2006;17:486–496.
- 141. Massachusetts Health Care Quality Act. United States, 2010.
- 142. Victorian Department of Human Services. Report of the Findings of a Literature Review and Consultation Process into Consumer Leadership. Melbourne: Rural and Regional Health and Aged Care Services Division, Victorian Department of Human Services, 2007.
- 143. Altringer B. The emotional experience of patient care: a case for innovation in health care design. *Journal of Health Services Research and Policy* 2010;15(3):174–177.
- 144. Ben-Tovim D, Dougherty M, O'Connell T, McGrath K. Patient journeys: the process of clinical re-design. *Medical Journal of Australia* 2008;188(suppl.6):S14–S17.
- 145. Bate P, Robert G. Bringing User Experience to Healthcare Improvement: The Concepts, Methods and Practices of Experience-Based Design. Oxford/Seattle: Radcliffe Publishing, 2007.
- 146. Queensland Children's Hospital Youth Advisory Forum. Beyond Beanbags: The Youth Advisory Forum's Vision of the Queensland Children's Hospital. Brisbane, 2009.
- 147. Watson D, Peterson S, Black C. In Pursuit of Quality: Opportunities to Improve Patient Experiences in British Columbian Emergency Departments. Vancouver: University of British Columbia Centre for Health Services and Policy Research, 2009.
- 148. Boudreaux E, O'Hea E. Patient satisfaction in the emergency department: a review of the literature and implications for practice. *Journal of Emergency Medicine* 2004;26(1):13–26.
- 149. Toma G, Triner W, McNutt L. Patient satisfaction as a function of emergency department previsit expectations. Annals of Emergency Medicine 2009;54(3):360–367.
- 150. Berwick DM. What 'Patient-centered' should mean: confessions of an extremist. *Health Affairs* 2009;28(4):w555–565.
- 151. Coulter A. The NHS revolution: health care in the market place What do patients and public want from primary care? BMJ 2005; 331:1199–1201.
- 152. Luxford K, Safran D, Delbanco T. Promoting patient-centered care: a qualitative study of facilitators and barriers in healthcare organizations with a reputation for improving patient experience. International Journal for Quality in Health Care 2011;doi: 10.1093/ intqhc/mzro24.
- 153. Patients for Patient Safety. World Health Organization, 2010. (Accessed 18 August 2010, at www.who.int/patientsafety/patients_for_patient/statement/en/index.html)
- 154. Balik B, Conway J, Zipperer L, Watson J. Achieving an Exceptional Patient and Family Experience of Inpatient Hospital Care. IHI Innovation Series White Paper. Cambridge, Massachusetts: Institute for Healthcare Improvement, 2011.

- 155. Johnson B, Abraham M, Conway J, Simmons L, Edgman-Levitan S, Sodomka P, et al. Partnering with Patients and Families to Design a Patient and Family-Centered Health Care System: Recommendations and Promising Practices. Besthesda, Cambridge: Institute for Patient- and Family-Centred Care and Institute for Healthcare Improvement, 2008.
- 156. Frampton S, Gilpin L, Charmel P. Putting Patients First: Designing and Practicing Patient-Centered Care. San Francisco: Jossey-Bass, 2003.
- 157. Lown BA, Manning CF. The Schwartz Center Rounds: evaluation of an interdisciplinary approach to enhancing patient-centered communication, teamwork, and provider support. Academic Medicine 2010;85(6):1073–1081.
- 158. Healy J, Sharman E, Lokuge B. Australia: health system review. In: Healy J, editor. Health Systems in Transition. Copenhagen: World Health Organization, 2006:1–178.
- 159. Council of Australian Governments. National Health and Hospitals Network Agreement. Canberra: Commonwealth of Australia, 2010.
- 160. Council of Australian Governments. National Health Reform— Heads of Agreement. Canberra: Commonwealth of Australia, 2011.
- 161. Council of Australian Governments Reform Council. National Healthcare Agreement Report on Performance 2008–09. Canberra: Commonwealth of Australia, 2010.
- 162. Piper D, Dunbar N. Healthcare rights: will patients' rights be respected? Windows into Safety and Quality. Sydney: Australian Commission on Safety and Quality in Health Care, 2008.
- 163. Palmer G, Short S. *Health Care and Public Policy*. third ed. South Yarra: Macmillan Publishers Australia, 2000.
- 164. Australian Commission on Safety and Quality in Health Care. National Safety and Quality Framework — A National Framework for Improving the Safety and Quality of Health Care. Sydney: Australian Commission on Safety and Quality in Health Care, 2009:1–9.
- 165. Australian Commission on Safety and Quality in Health Care. Draft National Safety and Quality Health Service Standards. Sydney: Australian Commission on Safety and Quality in Health Care. 2010.
- 166. Department of Health and Ageing. Building a 21st Century Primary Health Care System: Australia's First National Primary Health Care Strategy. Canberra: Commonwealth of Australia, 2010.
- 167. National Health Priority Action Council. National Chronic Disease Strategy. Canberra: Commonwealth of Australia, 2006.
- 168. Department of Health and Ageing. Fourth National Mental Health Plan: An Agenda for Collaborative Government Action in Mental Health 2009–2014. Canberra: Commonwealth of Australia, 2009.
- 169. Department of Health and Ageing. Fifth Community Pharmacy Agreement. Canberra: Department of Health and Ageing, Commonwealth of Australia, 2010.
- Department of Aboriginal Affairs. National Aboriginal Health Strategy. Canberra: Department of Aboriginal Affairs, 1989.
- 171. Griew R. The Link Between Primary Health Care and Health Outcomes for Aboriginal and Torres Strait Islander Australians. Canberra: Office for Aboriginal and Torres Strait Islander Health, Department of Health and Ageing, Commonwealth of Australia 2008
- 172. Griew R. Family-Centred Primary Health Care. Canberra: Office for Aboriginal and Torres Strait Islander Health, Department of Health and Ageing, Commonwealth of Australia, 2007.
- 173. Douglas N, Robinson J, Fahy K. Inquiry into Obstetric and Gynaecological Services at King Edward Memorial Hospital 1990– 2000 Final Report. Perth: Government of Western Australia, 2001.
- 174. ACT Community and Health Services Complaints Commissioner. A Final Report of the Investigation into Adverse Patient Outcomes of Neurosurgical Services Provided by the Canberra Hospital. Canberra: ACT Community and Health Services Complaints Commissioner, 2003.

- 175. Victorian Health Services Commissioner. Royal Melbourne Hospital Inquiry Report. Melbourne: Office of the Health Services Commissioner, 2002.
- 176. Walker B. Final Report of the Special Commission of Inquiry into Campbelltown and Camden Hospitals. Sydney: NSW Attorney-General's Department, NSW Government, 2004.
- 177. Davies G. Queensland Public Hospitals Commission of Inquiry Final Report. Brisbane: Queensland Government 2005.
- 178. Forster P. Queensland Health Systems Review Final Report.
 Brisbane: Queensland Government, 2005.
- 179. Hughes C, Walters W. Report of Inquiry Into the Care of a Patient With Threatened Miscarriage at Royal North Shore Hospital on 25 September 2007. Sydney: NSW Health, 2007.
- 180. Joint Select Committee on the Royal North Shore Hospital. Report on Inquiry into the Royal North Shore Hospital. Sydney: NSW Legislative Council, State of New South Wales, 2007.
- 181. Garling P. First Report of the Special Commission of Inquiry Acute Care Services in NSW Public Hospitals — Inquiry into the Circumstances of the Appointment of Graeme Reeves by the Former Southern Area Health Service. Sydney: State of New South Wales, 2008.
- 182. Garling P. Final Report of the Special Commission of Inquiry into Acute Care Services in NSW: Acute Care Services in NSW Public Hospitals. Sydney: State of New South Wales, 2008.
- 183. Hindle D, Braithwaite J, Travaglia J, Iedema R. Patient Safety: A Comparative Analysis of Eight Inquiries in Six Countries. Sydney: Centre for Clinical Governance Research, University of New South Wales, for the NSW Clinical Excellence Commission, 2006:1–35.
- 184. NSW Health. Caring Together: The Health Action Plan for NSW. Sydney: NSW Health, 2009.
- 185. Queensland Health. Action Plan Building a Better Health Service for Queensland. Brisbane: Queensland Government, 2005.
- 186. Bureau of Health Information. Insights into Care: Patients' Perspectives on NSW Public Hospitals. Sydney: Bureau of Health Information, 2010.
- 187. Department of Health and Human Services Tasmania. Leading the Way: Tasmania's Health Professionals Shaping Future Care. Hobart: Department of Health and Human Services, 2009.
- 188. Draper M, Cohen P, Buchan H. Seeking consumer views: what use are results of hospital patient satisfaction surveys? *International Journal for Quality in Health Care* 2001;13(6):463–468.
- 189. Greco M, Carter M, Powell R, Sweeney K, Jolliffe J, Stead J. Impact of patient involvement in general practice. Education for Primary Care 2006;17:486–495.
- Bogomolova S, Høj S. Council of Ambulance Authorities Patient Satisfaction Survey Results. Adelaide: Ehrenberg-Bass Institute, 2009.
- 191. Royal Australian College of General Practitioners. Standards for General Practice, fourth ed. Melbourne: Royal Australian College of General Practitioners, 2010.
- 192. ACT Health. Listening and Learning ACT Health Consumer Feedback Standards. Canberra: ACT Health, 2004.
- 193. ACT Health. Consumer Feedback Management Policy. Canberra: ACT Health, 2008.
- 194. ACT Health. Towards a Patient and Family-centred Care Model in the ACT. Canberra: ACT Health, 2011.
- 195. Iedema R, Merrick E, Piper D, Walsh J. Emergency Department Co-Design Evaluation Stage 1 — Final Report to Health Services Performance Improvement Branch. Sydney: Centre for Health Communication, University of Technology, for NSW Health, 2008.
- 196. Piper D, Iedema I, Merrick E, Perrot B. Experience-based Co-design Evaluation Program 2 Stage 1 — Final Report to Health Services Performance Improvement Branch. Sydney: Centre for Health Communication, University of Technology, for NSW Health, 2010.

- SA Health. Consumer and Community Participation Policy Directive. Adelaide: SA Health, 2009.
- SA Health. Consumer and Community Participation Guideline.
 Adelaide: SA Health, 2009.
- 199. Victorian Department of Health. Doing it With Us Not For Us—
 Participation in Your Health System 2006–09: Victorian Consumers,
 Carers and the Community Working Together With Their Health
 Services and the Department of Human Services. Melbourne: State
 Government of Victoria, 2006.
- 200. Victorian Department of Health. *Doing it With Us Not For Us:*Strategic Direction 2010–13. Melbourne: State Government of Victoria, 2009.
- 201. Victorian Department of Health. *Because Mental Health Matters— The Victorian Mental Health Reform Strategy 2009–19*. Melbourne:
 State Government of Victoria, 2009.
- 202. WA Health. WA Health Consumer Carer and Community Engagement Framework: for Health Services, Hospitals and WA Health Following Consultation Across WA Health. Perth: WA Health, 2007.
- 203. WA Health. Western Australian Strategic Plan for Safety and Quality in Health Care 2008–2013 Placing Patients First. Perth: WA Health, 2008.
- 204. WA Health. Western Australian Health Complaint Management Policy. Perth: WA Health, 2009.
- ACT Health. Quality in Healthcare Awards. Canberra: ACT Health,
 2009
- 206. Booth BJ, Snowdon T. A quality framework for Australian general practice. Australian Family Physician 2007;36(1/2):8–11.
- 207. Booth BJ, McBride T. Patient focus and the quality framework.

 Australian Family Physician 2007;36(1/2):20–22.
- 208. Cordingley A. Western Australia Health Consumers' Council: Patient First Ambassador Project. Health Issues 2009;98(autumn):12–15.
- 209. WA Council for Safety and Quality in Health Care. Patient First
 Program Booklet. Perth: Office of Safety and Quality in Health Care,
 WA Department of Health, 2006.
- 210. Austin Health. Respecting Patient Choices: An Evaluation of the Community Implementation of the Respecting Patient Choices Program. Melbourne: Austin Health, 2006.
- 211. Lawn S, Smith A, Hunter K. Mental health peer support for hospital avoidance and early discharge: an Australian example of consumer driven and operated service. *Journal of Mental Health* 2008;17(5):498–508.
- 212. Jorm CM, Dunbar N, Sudano L, Travaglia JF. Should patient safety be more patient centred? Australian Health Review 2009; 33(3):390–399.
- 213. McGrath K, Bennett D, Ben-Tovim D, Boyages S, Lyons N, O'Connell T. Implementing and sustaining transformational change in health care: lessons learnt about clinical process redesign. Medical Journal of Australia 2008;188(suppl.6):32–35.
- Wakefield J, Jorm C. Patient safety a balanced measurement framework. Australian Health Review 2009;33(3):382–389.
- 215. Gillespie R, Florin D, Gillam S. Changing Relationships: Findings from the Patient Involvement Project. London: King's Fund, 2002.
- 216. Dumont S, Briere N, Morin D, Houle N, Iloko-Fundi M. Implementing an interfaculty series of courses on interprofessional collaboration in prelicensure health science curriculums. Education for Health 2010;23(1):395.
- Hasman A, Coulter A, Askham J. Education for Partnership: Developments in Medical Education. London: Picker Institute Europe, 2006.
- 218. Schall M, Sevin C, Wasson JH. Making high-quality, patient-centered care a reality. *The Journal of Ambulatory Care Management* 2009;32(1):3–7.

- 219. Kotter J. Winning at change. Leader to Leader 1998;10(fall):27-33.
- 220. UK Cabinet Office. Excellence and Fairness: Achieving a World Class Health System. London: UK Cabinet Office, 2008.
- 221. Burns K. Canadian patient safety champions: collaborating on improving patient safety. *Healthcare Quarterly* 2008;11(sp):95–100.
- 222. Jenkinson C, Coulter A, Bruster S, Richards N, Chandola T. Patients' experiences and satisfaction with health care: results of a questionnaire study of specific aspects of care. Quality and Safety in Health Care 2002;11(4):335–339.
- 223. Arneill B, Frasca-Beaulieu K. Healing environments: architecture and design conducive to health. In: Frampton SB, Gilpin L, Charmel P, editors. Putting Patients First: Designing and Practicing Patient-Centered Care.
- 224. Coulter A. Patients' expectations. In: Carter Y, Jackson N, editors. Medical Education and Training: From Theory to Delivery. Oxford: Oxford University Press, 2009:45–57.
- 225. Rave N, Geyer M, Reeder B, Ernst J, Goldberg L, Barnard C. Radical systems change: innovative strategies to improve patient satisfaction. *Journal of Ambulatory Care Management* 2003;26(2):159–174.
- 226. Senge P. The Fifth Discipline The Art and Practice of the Learning Organization. London: Random House, 1990.
- 227. Peters T. Thriving on Chaos: Handbook for a Management Revolution. New York: Harpers Paperbacks, 1987.

Appendix A

Assessing organisational readiness to implement patient-centred care

Appendix A contains two tools that may assist health services to analyse their own needs regarding patient-centred care practices.

- » Tool 1: a checklist for assessing organisational readiness for implementing patient-centred care, written by Dr Karen Luxford in the context of the Australian healthcare system.
- » Tool 2: an assessment tool from the Institute for Patient- and Family-Centered Care (IPFCC) entitled Where Do We Stand? An Assessment Tool for Hospital Trustees, Administrators, Providers and Patients and Family Leaders.

.....

Question 14. Are successes by staff visibly celebrated? Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Tool 1: Patient-centred care organisational status checklist		
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	CHECK YOUR ORGANISATIONAL READINESS FOR IMPROVING	,	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	PATIENT CARE EXPERIENCE:		
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 1. Are you collecting patient care experience data?	Y	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 2. How are you collecting patient care experience data?		
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 3. Why are you collecting patient care experience data?	••••••	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 4. How often are you collecting patient care experience data?	••••••	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 5. How are you using the data/information collected?		
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 6. Is the data about patient care experience being reported?	Y	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 7. To whom is it being reported?		
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Question 21. What did the organisation learn from these events?	Question 8. Is staff satisfaction monitored?	Y	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 9. Is there a 'scorecard' of performance metrics monitored by the organisation?	Y	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 10. Does the scorecard/set of metrics include patient care experience indicators?	Y	
Question 12. What is the main message to staff from the leadership? CEO? Organisational governance? Question 13. Is the culture of the organisation supportive of change? Open to learning? Y Question 14. Are successes by staff visibly celebrated? Y Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 11. What is the mission/vision of the organisation?		
Question 14. Are successes by staff visibly celebrated? Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Question 18. Are any patient or family/carer representatives involved in any organisational committees? Question 19. Are there any future plans for engaging patients at a service level within the organisation? Question 20. Have there been any tragic events within the service from which lessons have been learnt? Question 21. What did the organisation learn from these events?			
Question 15. What is the current area of focus for staff development? Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Question 18. Are any patient or family/carer representatives involved in any organisational committees? Question 19. Are there any future plans for engaging patients at a service level within the organisation? Question 20. Have there been any tragic events within the service from which lessons have been learnt? Question 21. What did the organisation learn from these events?	Question 13. Is the culture of the organisation supportive of change? Open to learning?	Y ,	
Question 16. Have staff training activities included communication skills training or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 14. Are successes by staff visibly celebrated?	Y	
or patient-centred values? Question 17. Are patients and families considered 'partners' in care? Y Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 15. What is the current area of focus for staff development?		
Question 18. Are any patient or family/carer representatives involved in any organisational committees? Y Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?			
Question 19. Are there any future plans for engaging patients at a service level within the organisation? Y Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 17. Are patients and families considered 'partners' in care?	Y	
Question 20. Have there been any tragic events within the service from which lessons have been learnt? Y Question 21. What did the organisation learn from these events?	Question 18. Are any patient or family/carer representatives involved in any organisational committees?	Υ /	
Question 21. What did the organisation learn from these events?	Question 19. Are there any future plans for engaging patients at a service level within the organisation?	Υ /	
	Question 20. Have there been any tragic events within the service from which lessons have been learnt?	Υ /	
Question 22. Are families/carers considered 'visitors' to the service (e.g. restricted 'visiting' hours)?	Question 21. What did the organisation learn from these events?		
	Y		

- WHERE DO WE STAND? AN ASSESSMENT TOOL FOR HOSPITAL TRUSTERS, ADMINISTRATORS, PROVIDERS, AND PATIENT AND FAMILY LEADERS (IPFCC 2008)

 Initial assessment

 Organisational culture and philosophy of care

 And the organisation's vision, mission, and philosophy of care statements reflect the principles of patient and families reversed.

 Has the organisation defined quality health care, and does this definition include how patients and families will operatione care.

 Has the organisation defined quality health care, and does this definition include how patients and families will operatione care.

 Has the organisation of patients and families, and others in the community?

 Do the organisation's leaders model collaboration with patients and families?

 Are the organisation's leaders model collaboration with patients and families?

 Are the organisation's leaders model collaboration with patients and families are allies for patient health, after, and well being?

 Patient and family participation in organisational advisory roles

 Is there an organisational Patient and Family Advisory Council?

 If there is a Patient and Family Advisory Council, is patient safety a regular agenda item?

 Are patients and families members of committees and are they involved in initiatives for:

 patient safety?

 quality improvement?

 Lacility design?

 Lacility design.

 Lacility design.

 Lacility design.

 Lacility design.

 Lacili

- Patient and family access to information

 Are there systems in place to ensure that patients and families have access to complete, unbiased, and useful information?

 Do patients and families, in accordance with patient preference, have timely access to medication lists, clinical information (e.g., lab. x-ray and other test result), and discharge or transition nummaries?

 Are informational and educational resources available in a variety of formats and media, and in the languages and at the reading levels of the individuals served?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 Are patients and families provided with practical information on how to best assure safety in health care?

 The patients and families function on how to best assure safety in health care?

 The patients and families function on health care and families, including practical information on how to best assure safety in health care?

 The patients and families provided with practical information on how to best assure safety in hea

Appendix BSelect annotated bibliography

This annotated bibliography contains most of the citations from Section 1.3 regarding the evidence for patient-centred care.

The bibliography provides more information for individuals who may need to make a business case or support the development or implementation of patient-centred care strategies. The cited papers provide evidence about the relationship between patient-centred approaches and outcomes related to clinical quality, the experience of care, the business and operation of delivering health services, and the work environment. Citations from Section 1.3 that have not been published in peer-reviewed journals are not included in this bibliography. 42, 56–57

Arbuthnott A, Sharpe D. The effect of physicianpatient collaboration on patient adherence in non-psychiatric medicine. *Patient Education and Counselling* 2009;77:60–67.⁴⁷

Type of study: Meta-analysis.

Aims and methods: To investigate the effect of physician—patient collaboration in non-psychiatric patient treatment adherence. An initial search of the Cochrane, Medline and PsycInfo databases using the keywords 'adherence', 'compliance', 'nonadherence' and 'noncompliance' was conducted. Studies eligible for inclusion were published in English between 1968 and 2007, and fulfilled the following three criteria: (i) measured some aspect of physician—patient collaboration (e.g. communication, satisfaction with degree of involvement); (ii) had patient adherence as an outcome variable; (iii) comprised a non-psychiatric sample.

Results and analysis: Data from 48 studies (n = 34 000 patients) were included in the meta-analysis. Patients had a range of illnesses, including asthma, bacterial infection, breast cancer, congestive health failure, diabetes, fibromyalgia, HIV, hypertension, inflammatory bowel disease, renal disease and tuberculosis. Patient–physician interaction was found to be associated with patient adherence ($M_{\rm d}$ = 0.145).

Limitations: The authors acknowledged the following limitations: (i) study designs and research questions varied; (ii) few studies were found measuring only a single aspect of collaboration; (iii) the inclusion of both observational and interventional studies was necessitated by the interdisciplinary nature of the question.

Conclusions: Strong patient–physician collaboration is associated with adherence to treatment.

Arnetz JE, Winblad U, Höglund AT, Lindahl B, Spångberg K, Wallentin L, et al. Is patient involvement during hospitalization for acute myocardial infarction associated with post-discharge treatment outcome? An exploratory study. *Health Expectations* 2010; 13(3):298–311.³⁷

Type of study: Self-administered questionnaire.

Aims and methods: To investigate whether there is an association between patients' perceptions of their involvement during hospitalisation for acute myocardial infarction (AMI), and their health (e.g. shortness of breath) and behavioural (e.g. medication compliance) outcomes 6–10 weeks after hospital discharge. Cardiac patients attending their first follow-up visit were invited to complete a 53-item, forced-choice questionnaire rating their involvement on four-point Likert scales.

Results and analysis: Questionnaires were completed by 591 patients (age ≤75 years). Responses to questionnaire items were compared against ten medical and behavioural outcome variables from a national database of cardiac patients.

Limitations: Database follow-up information was missing for nearly one-quarter (n = 142) of questionnaire respondents, meaning their health outcomes were unknown. There was inconsistency in the timing of initial follow-up visits, so the time that patients received the questionnaire varied.

Conclusions: Higher ratings of patient involvement appear to be related to experiencing fewer cardiovascular symptoms (i.e. angina and shortness of breath). Patient ratings of involvement in care do not appear to be related to medication compliance.

Beach CM, Sugarman J, Johnson RL, Arbaleaz JJ, Duggan PS, Cooper LA. Do patients treated with dignity report higher satisfaction, adherence and receipt of preventive care? *Annals of Family Medicine* 2005;3(4):331–338.⁵²

Type of study: Cross-sectional analysis using data from the Commonwealth Fund 2001 Health Care Quality Survey.

Aims and methods: To assess the association between two measures of respect (treating patients with dignity and involving them in decision making) and positive patient outcomes (satisfaction, adherence and optimal use of preventive health care).

Results and analysis: Data from 5514 respondents who reported a medical encounter in the previous two years were used in analyses. Separate logistic regression analyses were conducted to investigate the association of the two measures of respect with the three outcome variables, in adjusted and unadjusted models. Patients who were treated with dignity or involved in decision-making were more likely to report a higher level of satisfaction, but there were no significant differences between the groups in the probability of receiving optimal preventive care.

Limitations: As with all cross-sectional analyses, no conclusions regarding causality can be drawn. The modest response rate (54 per cent) means a response bias is possible. Recall bias may also have influenced findings, as patients were asked to think back over a two-year period.

Conclusions: Involvement in decisions and being treated with dignity are independently associated with positive outcomes, including patient satisfaction and adherence benefits.

Boulding W, Glickman S, Manary M, Schulman K, Staelin R. Relationship between patient satisfaction with inpatient care and hospital readmission within 30 days. *American Journal of Managed Care* 2011;17(1):41–48.⁴¹

Type of study: Observational analysis comparing clinical performance, patient satisfaction and 30-day risk standardised readmission rates for three clinical outcomes (acute myocardial infarction, heart failure and pneumonia).

Aims and methods: To investigate whether patient satisfaction with inpatient care and discharge planning is associated with lower 30-day readmission rates for the three clinical outcomes. Data about readmission rates were sourced from Hospital Compare (for the period July 2005–June 2008). Data about patients' perceptions were sourced from the Hospital Care Quality Information from the Consumer Perspective database.

Results and analysis: Patients' overall satisfaction ratings and perceptions regarding the discharge process were negatively associated with 30-day readmission rates for acute myocardial infarction, heart failure and pneumonia. Satisfaction was strongly correlated with perceptions of quality of communication with nursing staff.

Limitations: The cross-sectional design means causality cannot be inferred. The authors acknowledge that it is possible that some patients were readmitted before completing the survey.

Conclusions: Patient experience data should be used to complement objective clinical measures in assessments of quality of health care.

Charmel P, Frampton S. Building the business case for patient-centered care. *Healthcare Financial Management* 2008;62(3):80–85.⁵⁴

Type of study: Discussion paper about patient-centred care.

Aims and methods: To present the argument that patient-centred care is fundamental to providing high-quality care. Patient-centred care encourages patients to be actively involved in their own care. Providing comfort through a sensitive organisational culture, a home-like physical environment, good nutrition and staff who are sensitive to patient's physical, emotional and spiritual needs is the essence of patient-centred care.

Content and scope: The authors describe the conception of Planetree 30 years ago. Planetree is a not-for-profit organisation dedicated to helping healthcare facilities achieve patient-centred care. The introduction of value-based purchasing in the US was anticipated to lead to healthcare facilities receiving reimbursement if they were able to demonstrate patient-centred care through providing excellent clinical outcomes and patient experiences.

Limitations: One of the authors is the president of Planetree, which may introduce bias.

Conclusions: Adopting a patient-centred care approach makes good business sense, as it is expected that consumers will be drawn to facilities that offer personalised care and value for money.

DiGioia AI, Greenhouse PK, Levison TJ. Patient and family-centered collaborative care: an orthopaedic model. *Clinical Orthopaedics and Related Research* 2007;463:13–19.²²⁹

Type of study: Prospective data collection from patient-reported forms and the hospital's health management database; no exclusion criteria.

Aims and methods: To investigate patient- and family-centred collaborative care among orthopaedic patients in the Total Joint Replacement Program at the University of Pittsburgh Medical Center in the US. Participants were 618 consecutive patients undergoing hip or knee surgery. Patient satisfaction was measured by a national survey (Press Ganey Survey) and an internally developed survey.

Results and analysis: The authors describe the patientand family-centred collaborative care process for total joint replacement in their program. Patient overall satisfaction ratings for the program were very high, with a mean rating of 91.4. Patient- and family-centred collaborative care was perceived as worthy by patients, with 99.7 per cent reporting that they would refer others to the program.

Limitations: There was no control group and most comparisons were made against national data. Baseline data from the hospital were not presented; therefore, it was unclear if improvements had been made.

Conclusions: Adopting a patient- and family-centred collaborative care approach can improve quality of health care, but requires reorganisation and restructuring of existing care delivery programs. However, this need not be complicated, expensive or highly technological. The authors also emphasise the importance of involving staff from all relevant departments, as well as the need for encouragement and support from hospital leadership.

Edgcumbe D. Patients' perceptions of hospital cleanliness are correlated with rates of methicillin-resistant *Staphylococcus aureus* bacteraemia. *Journal of Hospital Infection* 2009;71(1):99–101.⁴⁴

Type of study: Letter to the editor briefly detailing an observational analysis.

Aims and methods: To investigate the association between patients' perceptions of hospital cleanliness and rates of methicillin-resistant *Staphylococcus aureus* (MRSA) bacteraemia, using two publicly available data sources. Data relating to patient's perceptions came from Question 19 of the Healthcare Commission patient survey, which asked, 'In your opinion, how clean was the ward that you were in?' Response options were 'very clean', 'fairly clean', 'not clean' or 'not at all clean'. Bacteria rates, expressed as the number of MRSA bacteraemia reported per 1000 bed-days, were obtained from the Health Protection Agency's mandatory reporting scheme.

Results and analysis: A statistically significant correlation (-0.384; P<0.01) was found, suggesting that patients' perceptions of hospital cleanliness is related to MRSA rates.

Limitations: No limitations were identified by the author. Findings were in opposition to another study, which did not find that perceptions of cleanliness correlated with rates of MRSA bacteraemia; however, the author was unable to explain this discrepancy.

Conclusions: There is a paucity of research into the link between patients' perceptions of hospital cleanliness and clinical outcomes: in particular, hospital acquired infections. The author recommends that research be conducted to investigate this topic.

Epstein RM, Franks P, Shields CG, Meldrum SC, Miller KN, Campbell TL et al. Patient-centred communication and diagnostic testing. *Annals of Family Medicine* 2005;3(5):415–421.⁴⁹

Type of study: Observational cross-sectional study using patient surveys and audio recordings of clinical visits.

Aims and methods: To investigate the relationship between patient–physician interaction style and diagnostic test expenditure. Physicians (n = 100) were covertly recorded on two separate occasions, by standardised patients wearing concealed audio recorders. Information about diagnostic testing costs for each physician and their real patients was extracted from a managed-care claims database.

Results and analysis: Audio recordings were scored using the Measure of Patient-Centred Communication. Higher (better) scores were associated with lower diagnostic testing expenditure and longer consultations.

Limitations: Measure of Patient-Centred Communication scores were restricted to only two patient visits, as this was the maximum number of standardised patients that physicians would consent to seeing.

Conclusions: The links between patient-centred communication, consultation length and diagnostic testing expenditures may affect the willingness of health systems to endorse such an approach. Confirmation of results in randomised trials is warranted.

Flach SD, McCoy KD, Vaughn TE, Ward MM, BootsMiller BJ, Doebbeling BN. Does patient-centered care improve provision of preventive services? *Journal* of General Internal Medicine 2004;19(10):1019–1026.⁴⁵

Type of study: Observational data analysis.

Aims and methods: To investigate the relationship between patient-centred care and preventive service delivery using data from three national surveys. Eight domains of patient-centred care were measured, including access to care, inclusion of patient preferences, provision of patient education, coordination of visitors, coordination of care, courtesy, continuity of care and emotional support. Scores on the domains were compared against benchmarking scores on 12 preventive activities: (1-3) screening for hypertension, alcohol abuse and seat belt use; (4-6) counselling to quit smoking, increase physical activity and lose weight; (7-8) ordering tests for cholesterol and faecal occult blood; (9-12) ordering vaccinations against tetanus, influenza and pneumococcal infection, and undergoing flexible sigmoidoscopy.

Results and analysis: Patient-centred care data were sourced from the Veterans' Satisfaction Survey and data about receipt of preventive activities were sourced from the Veterans' Health Survey. Counselling about smoking cessation and ordering faecal occult blood tests were most strongly related to some of the patient-centred care components.

Limitations: The value that physicians place on preventive activities, and how this may affect initiation of the activity, is unknown. The accuracy of patients' self-reports of receiving preventive activities was not verified.

Conclusions: Some patient-centred care components are associated with hospital preventive activities.

Fremont AM, Cleary PD, Lee Hargraves J, Rowe RM, Jacobson NB, Ayanian JZ. Patient-centered processes of care and long-term outcomes of myocardial infarction. Journal of General Internal Medicine 2001;16(12):800–808.38

Type of study: Observational cohort study.

Aims and methods: To investigate the association between patients' experiences with nontechnical aspects of care, and health status and cardiac symptoms (overall health, physical health status, mental health status, presence of chest pain, shortness of breath) one year post-discharge. Participants (n=2272) were inpatients with AMI recruited from 23 New Hampshire hospitals. Surveys posted at 1, 3 and 12 months after discharge were completed by 59.2 per cent (n=1346), 46 per cent (n=1046) and 42.4 per cent (n=964) of participants, respectively. Primary analyses were conducted with data of the 762 patients who returned both the 1-month and the 12-month surveys.

Results and analysis: Patients returning surveys with lower ratings of experiences with care were more likely to report lower overall health and physical health; chest pain at 12 months was also more likely for these patients. However, this association may be offset by positive outpatient experiences.

Limitations: It is unclear if these findings can be generalised beyond AMI to other conditions. Data were incomplete for patients who died within the 12-month follow-up period, which reduced the number available to analyse.

Conclusions: Nontechnical processes of AMI hospital care are associated with long-term outcomes. Positive outpatient experiences may offset negative hospital experiences, for at least one outcome (chest pain at 12 months).

Glickman SW, Boulding W, Manary M, Staelin R, Roe MT, Wolosin RJ, et al. Patient satisfaction and its relationship with clinical quality and inpatient mortality in acute myocardial infarction. *Circulation:* Cardiovascular Quality and Outcomes 2010;3:188–195.³⁹

Type of study: Observational analysis.

Aims and methods: To investigate the relationship between patient satisfaction, hospital clinical quality and mortality in patients with AMI. Patient satisfaction data came from surveys sent to patients within a week of discharge from hospital. Clinical process-of-care data were sourced from a registry (CRUSADE) comprising inpatient care and outcomes pertaining to specific cardiac conditions.

Results and analysis: Surveys from 3562 patients and clinical data from 6467 patients were used in analyses. Even when clinical performance was controlled for, higher hospital-level patient satisfaction was associated with lower inpatient mortality rates.

Limitations: Only data from hospitals contributing to CRUSADE were analysed, which may introduce bias. It is unclear if findings can be generalised beyond AMI to other conditions. Data were incomplete for patients who died during their hospitalisation, which reduced the number of satisfaction surveys.

Conclusions: Patients' levels of satisfaction should be considered a source of information on quality of care, as higher patient satisfaction is associated with improved guideline adherence and lower inpatient mortality.

Isaac T, Zaslavsky AM, Cleary PD, Landon BE. The relationship between patients' perception of care and measures of hospital quality and safety. *Health Services Research* 2010;45(4):1024–1040.³³

Type of study: Secondary data analysis.

Aims and methods: To examine the relationship between patient experience and other measures of hospital safety and quality. Data from three sources were compared: patient experience data from Hospital Consumer Assessment of Healthcare Providers and Systems national database; and process of care measures from the Hospital Quality Alliance and Patient Safety Indicators to measure medical and surgical complication rates, sourced from Medicare claims data.

Results and analysis: The overall rating of the hospital and readiness to recommend the hospital, as rated by patients, were associated with technical quality of care for pneumonia, congestive heart failure, myocardial infarction and surgical care.

Limitations: The Hospital Quality Alliance measures, while widely used and accepted, are only a subset of medical and surgical care measures. The validity of patient safety indicators as safety measures has not been established.

Conclusions: Patient experience data are associated with some technical quality of care and hospital safety measures (as measured by patient safety indicators).

Jha AK, Orav EJ, Zheng J, Epstein AM. Patients' perception of hospital care in the United States. *New England Journal of Medicine* 2008;359(18):1921–1931.³⁴

Type of study: Secondary analysis of data from the Hospital Consumer Assessment of Healthcare Providers and Systems (H-CAHPS) survey.

Aims and methods: To investigate whether key hospital characteristics believed to improve patients' experiences (high ratio of nurses to patient days, being for-profit and not a teaching hospital) are associated with patient satisfaction. Data collected by the H-CAHPS survey about patients' experiences were linked with data about a range of key hospital characteristics collected annually by the American Hospital Association.

Results and analysis: Higher ratios of nurses to patient days predicted higher (better) patient experience scores. Hospitals that provided higher quality care also received higher patient satisfaction ratings. However, patients in for-profit hospitals gave lower satisfaction scores than patients in either private or public not-for-profit hospitals. Patients' ratings were not associated with whether they were in a teaching or non-teaching hospital.

Limitations: There is a possible nonresponse bias, as data were not provided by approximately 40 per cent of US hospitals.

Conclusions: High nurse to patient ratios appear to be associated with better patient experiences. Higher quality of care is also associated with greater patient satisfaction.

Little P, Everitt H, Williamson I, Warner G, Moore M, Gould C et al. Observational study of effect of patient-centeredness and positive approach on outcomes of general practice consultations. *British Medical Journal* 2001;323:908–911.⁵⁰

Type of study: Observational study using preconsultation and post-consultation questionnaires

Aims and methods: This study investigated patients' perceptions of patient-centredness and the relationship to, and ability to predict, three patient outcomes (enablement, satisfaction and burden of symptoms). Patients from three general practices were approached consecutively and invited to complete pre-consultation and post-consultation questionnaires. The questionnaires were designed to assess patients' expectations of what the doctor would do during the consultation, and tapped into five domains of patient-centredness: 'exploring the disease and illness experience, understanding the whole person, finding common ground, health promotion and enhancing the doctor–patient relationship'.

Results and analysis: A total of 865 patients consented and 661 (76 per cent) completed both questionnaires using seven-point Likert scales to rate how much they agreed or disagreed (very strongly agree to very strongly disagree) with each statement. The postconsultation questionnaire included items relating to the three outcome measures, reasons for consultation and demographic information.

Limitations: The study design precluded any conclusions regarding cause and effect, and the authors recommended the conduct of randomised trials or cohort studies with a homogenous sample. Patients wanted doctors to provide a patient-centred approach to care, and those who did not receive this were less satisfied, less enabled and may have had a greater symptom burden.

Conclusions: There are five components of patients' perceptions that can be reliably measured: communication and partnership; personal relationship; health promotion; positive approach to diagnosis and prognosis and interest in the effect on life. Measurement of these markers can indicate the quality of care.

Meterko M, Wright S, Lin H, Lowy E, Cleary PD. Mortality among patients with acute myocardial infarction: The influences of patient-centered care and evidence-based medicine. *Health Services Research* 2010;45(5p1):1188–1204.⁴⁰

Type of study: Secondary data analysis of the Veterans Affairs Survey of Healthcare Experiences of Patients.

Aims and methods: To examine the association between patient-centred care and technical care quality on one-year mortality among patients with AMI. Patient assessments of patient-centred care were from the national Veterans Affairs Survey of Healthcare Experiences of Patients, which was administered to veterans who had been hospitalised with AMI. Data on 14 measures of technical quality of care were obtained through a medical record review.

Results and analysis: Higher patient-centred care ratings were modestly associated with lower hazard of death one year after the AMI.

Limitations: As the sample was predominantly male and all participants were veterans, it is unclear if these findings can be generalised to other AMI patients.

Only data from AMI patients with survey scores were analysed. This creates a response bias, as those missing survey data were an average of two years older. This has particular implications to the finding that increased age was significantly associated with higher risk of one-year mortality.

Conclusions: Providing patient-centred care may result in important clinical outcomes, such as a reduced risk of death one year post-hospitalisation for AMI among veterans.

Schneider EC, Zaslavsky AM, Landon BE, Lied TR, Sheingold S, Cleary PD. National quality monitoring of Medicare Health Plans. *Medical Care* 2001; 39:1313–1325.³⁶

Type of study: Observational cohort study.

Aims and methods: To examine the relationships between reports on the quality of healthcare plans as measured by two publicly available measurement systems: the Consumer Assessment of Healthcare Providers and Systems (CAHPS) survey and the Health Plan–Employer Data and Information Set (HEDIS). CAHPS and HEDIS collect unique data. However, it would be expected that some data would be complementary; that is, ratings of experiences about a health plan (collected through CAHPS) may be related to some clinical measures (collected through HEDIS). Data from five composite measures and four ratings from CAHPS, and six measures of clinical quality from HEDIS, were compared.

Results and analysis: CAHPS items 'getting needed care' and 'health plan information and customer service' were positively and significantly associated with all HEDIS measures, except for the proportion of individuals who continued antidepressant medications after commencing treatment. In this group, health plan information and customer service was the most important factor.

Limitations: Not all HEDIS plans were complete, which could have implications for the power of the study.

Conclusions: Patient experience ratings relating to access to care and customer service provide markers for some measures of health plan quality.

Schneider J, Kaplan SH, Greenfield S, Li W, Wilson IB. Better physician-patient relationships are associated with higher reported adherence to antiretroviral therapy in patients with HIV infection. *Journal of General Internal Medicine* 2004;19:1096–1103.⁵³

Type of study: Cross-sectional analysis.

Aims and methods: To investigate the aspects of physician–patient interactions associated with high adherence to antiretroviral medication regimens in individuals with HIV infection. Patients with normal cognitive function who could read English were recruited from 22 outpatient HIV practices in metropolitan Boston. Patients completed six previously evaluated scales to assess the quality of interactions with their physician: (1) general communication;

(2) provision of HIV-specific information; (3) decision-making style; (4) global satisfaction with care; (5) readiness to recommend physician to others; and (6) trust. A seventh item was developed by the authors that assessed patients' perceptions of their physician's ability to help them adhere to their medication regimens. Adherence to medication was obtained through four items with Likert-type response options.

Results and analysis: Analyses were conducted on data from 552 patients who completed a patient–physician interaction questionnaire, were using antiretrovirals and had an adherence score. Of the seven measures of patient–physician interaction, six were independently associated with adherence score. Decision-making style was the only measure that was not significantly associated; however, a trend towards significance was reported.

Limitations: Adherence was assessed via self-reporting, and may therefore have been overestimated.

Conclusions: Multiple measures of interpersonal care show that better physician–patient relationships and physician–patient communication can produce better adherence with antiretroviral therapies.

Sequist TD, Schneider EC, Anastario M, Odigie EG, Marshall R, Rogers WH et al. Quality monitoring of physicians: linking patients' experiences of care to clinical quality and outcome. *Journal of General Internal Medicine* 2008;23(11):1784–1790.³⁵

Type of study: Observational study.

Aims and methods: To examine the association between clinical performance (processes of care and outcomes of care) and patient experiences of care. Patient ratings of experience with physicians were obtained from the Ambulatory Care Experiences Survey instrument. Individual physician performance data were derived from an electronic medical record system containing clinician notes, laboratory reports, and diagnostic and procedure codes.

Results and analysis: Scores (o–100) for patient experience and clinical quality measures were calculated and correlations conducted. Clinical measures of care (cholesterol management, poor HbA1c control, LDL cholesterol control and control of high blood pressure) were not significantly associated with patient experiences of care.

Limitations: Data used were from five major health plans in Massachusetts; therefore, the ability to generalise the results may be limited.

Conclusions: Patient experiences data are not correlated with clinical outcomes and are only modestly correlated with clinical process measures.

Stewart M, Brown JB, Donner A, McWhinney IR, Oates J, Weston W et al. The impact of patient-centred care on outcomes. *Journal of Family Practice* 2000;49(9):796–804.⁵¹

Type of study: Observational cohort study.

Aims and methods: To investigate whether patients who experienced patient-centred communication at an initial consultation would, two months later, experience faster recovery, better self-reported health and less subsequent medical care. Participants included 39 family physicians and 315 patients. Adult patients with one or more recurring problems were approached in physician practices and invited to participate. Consultations were audio recorded and later scored for three components of patient-centred communication: (1) extent to which disease and illness experience was explored; (2) extent to which physician attempted to understand the whole person; (3) extent to which the physician attempted to find common ground. Participants were interviewed by a research assistant to investigate how patient-centred they perceived the consultation to be. Medical care during the two-month follow up was assessed by chart review and a telephone interview with the patient.

Results and analysis: Patient-centred communication scores (from audio recordings) were not significantly related to any of the health outcomes (recovery, health status) or medical care outcomes (total number of visits over two months, diagnostic tests ordered over two months, and number and type of referrals made over two months). Patient-centred communication scores (from audio recordings) were significantly correlated (r = -0.16) with patients' perceptions that they and the physician had found common ground. Patients who perceived their visit as patient-centred received fewer diagnostic tests and referrals in the following two months.

Limitations: Approximately 30 per cent of patients approached did not consent. Male patients were overrepresented; however, gender was not identified as a confounding factor.

Conclusions: Doctors should not underestimate the value of attempting to find common ground with a patient. The mechanism(s) by which patient-centred communication and efficiency of care (reduced diagnostic tests and referrals) are associated requires further study.

Stone S. A retrospective evaluation of the impact of the Planetree patient-centered model of care on inpatient quality outcomes. *Health Environments Research and Design Journal* 2008(summer):55–69.46

Type of study: Retrospective quasi-experimental study.

Aims and methods: To evaluate the Planetree patient-centred model of care, specifically the impact on patient satisfaction, clinical outcomes (length of stay and readmission) and care costs (per case and productive nursing hours per patient day). Adult patients (mean age 67.33, range 40–88 years) who had undergone primary elective surgical knee or primary hip total joint replacement between 2002 and 2006 were recruited from two hospitals. The treatment sample (n = 587) came from a 30-bed medical/surgical telemetry unit that had followed the Planetree model of patient-centred care since 2002. The control sample (n = 282) came from a 22-bed surgical telemetry unit, located within another hospital 15 miles away.

Results and analysis: Length of stay was significantly shorter at the Planetree facility (mean 3.26 vs 3.51 days), and cost per case was significantly lower (mean \$5732.91 vs \$7823.20) each year from 2002 to 2006. Patient satisfaction with their physician was significantly higher in the control unit. The authors noted that more Planetree patients than comparison patients were assigned to their physician, and this lack of choice may partially explain this difference.

Limitations: Sample sizes were uneven between groups (587 vs 282) because of the lower return of satisfaction groups in the control unit. Raw data regarding productive nursing hours were not available, limiting the analyses to summarisation and discussion of general differences only. No data for this variable were available for the year 2002 in the control unit. Health status was self-reported by patients and not independently verified.

Conclusions: A Planetree patient-centred model of care may positively influence patients' level of satisfaction, and reduce length of stay and cost per case.

Sweeney L, Halpert A, Waranoff J. Patient-centered management of complex conditions can reduce costs without shortening life. *American Journal of Managed Care* 2007;13:84–92.⁵⁵

Type of study: Prospective cohort study.

Aims and methods: To examine the effect of intensive patient-centred management on patient access of key healthcare resources (i.e. admissions, inpatient days, emergency room visits, rehabilitation days, hospice days and home care days) and patient outcomes. Patients (n = 756) with a terminal diagnosis and multiple comorbidities (75% were oncology patients) were included in the patient-centred management intervention cohort (n = 358) or usual case management cohort (n = 398) based on their health plan membership. Each patient-centred management patient had a care team comprising a case manager, team leader and a physician. The care team met weekly to provide advice, ongoing assessment, anticipate medical problems, liaise with the patient's treating physician and suggest alternative treatment plans if appropriate. Goals were set with patients in an initial home visit, and patientcentred management ended when patients achieved their goals.

Results and analysis: Substantial reductions in hospital admissions (38 per cent), hospital days (36 per cent) and emergency visits (30 per cent) were noted among the patient-centred management group. Increase in use of hospices (62 per cent) and home care (22 per cent) were also noted. Cost analysis suggested an average saving of US\$18 599 per patient when patient-centred management was used. Chemotherapy and radiation rates were higher among oncology patients in the usual care group than in the patient-centred management group. Membership of the patient-centred management group was also associated with lower inpatient symptoms, including nausea, anaemia, fluid disorder and fever.

Limitations: The authors minimised the possibility of allocation bias by conducting intent-to-treat analyses.

Conclusions: Patient-centred management was associated with reduced costs, lower rates of admissions, fewer days in hospital, and more home care and hospice days.

Weingart SN, Zhu J, Chiapetta L, Stuver S, Schneider E, Epstein AM, et al. Hospitalized patients' participation and its impact on quality of care and patient safety. *International Journal for Quality in Health Care* 2011; 13(3):269–267.

Type of study: Observational study using random sample telephone interview and medical chart review.

Aims and methods: To investigate the extent of hospitalised patients' participation in their own care, and associations between participation, quality of care and patient safety. Patients who had been hospitalised in the last 6–12 months underwent a telephone interview about their hospitalisation and any complications, and were asked seven questions about their level of participation while in hospital. Of the 2025 included respondents, 788 also had their medical charts reviewed to identify adverse events.

Results and analysis: Almost all patients (99.9%) reported at least one participatory activity. Patients who reported five or more of the seven forms of participation consistently gave higher ratings for quality of care and were less likely to experience an adverse event.

Limitations: The authors point out several study limitations, including that patients who died during hospitalisation or before the survey were not included, and sicker patients may be less likely to participate in their care. Nonresponse and recall biases are possible.

Conclusions: Most hospitalised patients participate in their care. Higher participation correlates with favourable ratings for quality of care and reduced adverse events.

Appendix CSupportive resources

This appendix provides links to key organisations and resources relevant to patient-centred care approaches and strategies.

C1.

KEY ORGANISATIONS

Links to leading international organisations that focus on patient-centred approaches to care, as well as Australian organisations that are also beginning to focus on patient-centred care are listed below.

Leading international organisations

Dana-Farber Cancer Institute

www.dana-farber.org

Institute for Healthcare Improvement

www.ihi.org

Institute for Patient- and Family-Centered Care

www.ipfcc.org

Kenneth B Schwartz Center

www.theschwartzcenter.org

The King's Fund

www.kingsfund.org.uk

Picker Institute

www.pickerinstitute.org

Picker Institute Europe

www.pickereurope.org

Planetree

www.planetree.org

Studer Group

www.studergroup.com

World Health Organization — Patients for Patient Safety

www.who.int/patientsafety/patients_for_patient/statement/en

Key Australian organisations

Australian Institute for Patient and Family Centred Care

www.aipfcc.org.au

Australian organisations for health consumers

Carers Australia

www.carersaustralia.com.au

Centre for Health Communication and Participation

www.latrobe.edu.au/chcp

Cochrane Consumers and Communication Review Group

www.latrobe.edu.au/chcp/cochrane

Cochrane Consumer Network

www.cochrane.org/consumers

Consumers Health Forum of Australia

www.chf.org.au

Health Care Consumers' Association of the ACT

www.hcca.org.au

Health Consumers Alliance of South Australia Inc.

www.hcasa.asn.au

Health Consumers' Council (WA) Inc.

www.hconc.org.au

Health Consumers NSW

Email hcnsw@internode.on.net

Health Consumers Queensland

www.health.qld.gov.au/hcq

Health Consumers of Rural and Remote Australia Inc.

www.ruralhealth.org.au

Health Issues Centre Inc. (Victoria)

www.healthissuescentre.org.au

National, state and territory government organisations and programs

Department of Health and Ageing

www.health.gov.au

Department of Health and Ageing — National Primary Health Care Strategy

www.health.gov.au/primaryhealthstrategy

Department of Health and Ageing — National Chronic Disease Strategy and National Service Improvement Framework

www.health.gov.au/internet/main/publishing.nsf/content/7E7E9140A3D3A3BCCA257140007AB32B/\$File/stratal3.pdf

Department of Health and Ageing — Fourth National Mental Health Plan

www.health.gov.au/internet/publications/publishing.nsf/Content/mental-pubs-f-planog-toc

Department of Health and Ageing — Fifth Community Pharmacy Agreement

www.health.gov.au/internet/main/publishing.nsf/Content/fifth-community-pharmacy-agreement-copy

Australian Commission on Safety and Quality in Health Care

www.safety and quality.org. au

New South Wales

Health Care Complaints Commission

www.hccc.nsw.gov.au

NSW Bureau of Health Information

www.bhi.nsw.gov.au

NSW Clinical Excellence Commission

www.cec.health.nsw.gov.au

NSW Health

www.health.nsw.gov.au

NSW Health Essentials of Care program

www.health.nsw.gov.au/nursing/projects/eoc.asp

NSW Health Patient and Carer Experience

www.health.nsw.gov.au/performance/pcexperience.asp

NSW Health Patient Survey

www.health.nsw.gov.au/hospitals/patient_survey

NSW Health 'Take the Lead' project

www.health.nsw.gov.au/nursing/projects/take the lead.asp

Australian Capital Territory

ACT Health Services Commissioner

www.hrc.act.gov.au/content.php/content.view/id/6

ACT Health

www.health.act.gov.au/c/health?a=da&did=11143891&pid=1288659671

ACT Health Healthcare Survey

www.health.act.gov.au/c/health?a=da&did=10101609&pid=1254789026

ACT Health Access Improvement Program

www.health.act.gov.au/c/health?a=&did=10109188

Queensland

Health Quality and Complaints Commission

www.hqcc.qld.gov.au

Queensland Health

www.health.qld.gov.au

Queensland Health Healthcare Experience Improvement Program

www.health.qld.gov.au/cpic/hlthcr_exp_improve/hlthcr_exp_improve.asp

Queensland Health Patient Satisfaction Survey

www.health.qld.gov.au/quality/pat sat survey/patsat.asp

Queensland Health Redesign Projects

www.health.qld.gov.au/cpic/service improve/projects.asp

Victoria

Health Services Commissioner

www.health.vic.gov.au/hsc

Victorian Department of Health

www.health.vic.gov.au

Victorian Department of Health — Improving the Patient Experience Program

www.health.vic.gov.au/management innovation/resources

Victorian Department of Health — Victorian Patient Satisfaction Monitor

www.health.vic.gov.au/patsat

Victorian Quality Council

www.health.vic.gov.au/qualitycouncil/activities

Northern Territory

Health and Community Services Complaints Commission

www.hcscc.nt.gov.au

Northern Territory Department of Health and Families

www.health.nt.gov.au

South Australia

Health and Community Services Complaints Commissioner

www.hcscc.sa.gov.au

SA Health

www.sahealth.sa.gov.au

SA Health Patient Evaluation of Health Services Survey

www.health.sa.gov.au/pros/Default.aspx?PageContentMode=1&tabid=44

Tasmania

Health Complaints Commissioner Tasmania

www.healthcomplaints.tas.gov.au

Department of Health and Human Services Tasmania

www.dhhs.tas.gov.au

Western Australia

Western Australian Office of Health Review

www.healthreview.wa.gov.au

WA Health

www.health.wa.gov.au

WA Office of Safety and Quality in Healthcare & Patient First program

www.safetyandquality.health.wa.gov.au/involving patient/patient 1st.cfm

WA Health Toolkit for Collecting and Using Patient Stories for Service Improvement

www.health.wa.gov.au/hrit/docs/A toolkit for collecting and using patient stories.pdf

C2.

KEY RESOURCES AND TOOLS

A variety of online international and Australian resources, including assessment tools and checklists for health service providers, are listed below.

General patient-centred care strategies and approaches

Partnering with Patients and Families to Design a Patient and Family-Centred Health Care System — Recommendations and Promising Practices (2008)

www.ihi.org/IHI/Topics/PatientCenteredCare/PatientCenteredCareGeneral/Literature/PartneringwithPatients and Families.htm

A comprehensive review of the worldwide evidence of what works to engage patients and the public in healthcare

www.investinengagement.info

Seeing the Person in the Patient: Point of Care Review Paper

www.kingsfund.org.uk/current_projects/the_point_of_care

The Patient-Centered Care Improvement Guide

www.planetree.org/publications.html

Patient-Centered Care Resource Compendium

www.planetree.nl/media/Ledenge deel te/Patient-Centered % 2520 Care % 2520 Improvement % 2520 Evidence % 2520 Base.pdf

Self-assessment tool

www.patient-centeredcare.org/inside/selfassessment.html

Advancing the Practice of Patient- and Family-Centred Care — How to Get Started

www.ipfcc.org/pdf/getting started.pdf

Partnering with Patients and Families to Design a Patient- and Family-Centred Health Care System — Recommendations and Promising Practices

www.ipfcc.org/pdf/PartneringwithPatientsandFamilies.pdf

Partnering with Patients and Families To Design a Patient- and Family-Centered Health Care System: A Roadmap for the Future — A Work in Progress

www.ipfcc.org/pdf/Roadmap.pdf

Advancing Effective Communication, Cultural Competence, and Patient- and Family-Centered Care www.jointcommission.org/assets/1/6/ARoadmapforHospitalsfinalversion727.pdf

A Toolkit to Assist Organisations in Initiating Patient Safety Rounds

www.dana-farber.org/pat/patient-safety/patient-safety-resources/patient-rounding-toolkit.html

Schwartz Center Rounds

www.theschwartzcenter.org/programs/rounds.html

Carepages

www.carepages.com

Picker Institute Always Events Program

www.alwaysevents.pickerinstitute.org

Resources on collecting patient feedback

UK Care Quality Commission

www.cqc.org.uk/usingcareservices/healthcare/patientsurveys.cfm

UK Survey Coordination Centre

www.nhssurveys.org

UK General Practice Patient Survey results

www.results.gp-patient.co.uk/report/main.aspx

UK National Cancer Action Team Patient Experience

www.cancerinfo.nhs.uk

UK National Cancer Patient Experience Survey

www.dh.gov.uk/prod_consum_dh/groups/dh_digitalassets/documents/digitalasset/dh_122520.pdf

UK National Institute for Health and Clinical Excellence patient experience standard

www.nice.org.uk/guidance/index.jsp?action=byID&o=13259

Understanding What Matters — A Guide to Using Patient Feedback to Transform Services, UK Department of Health

www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicyAndGuidance/DH_099780

Agency for Healthcare Research and Quality (AHRQ) Consumer Assessment of Healthcare Providers and Systems (CAHPS) survey

www.cahps.ahrq.gov/default.asp

AHRQ National CAHPS Benchmarking Database

www.cahps.ahrq.gov/content/ncbd/ncbd Intro.asp?p=105&s=5

Royal Australian College of General Practitioners Patient Feedback Guide

 $www.racgp.org. au/Content/Navigation Menu/Practice Support/Standards for General Practices/Standards 4 th Edition/Patient feedback/Standards 4_Patient Feedback Guide.pdf$

Real-time patient feedback

UK websites offering patient reviews for healthcare services

www.iwantgreatcare.com and www.patientopinion.org.uk

Patient choice of provider

NHS Choices

www.nhs.uk

NHS survey results for patients' experience of choice

www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsStatistics/DH 085329

US Department of Health and Human Services Hospital Compare

www.hospitalcompare.hhs.gov

US Department of Health and Human Services Physician Compare

www.medicare.gov/find-a-doctor/provider-search.aspx

Rights-based charters and codes

The NHS Constitution

www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicyAndGuidance/DH_113613

American Hospital Association Communicating with patients

www. aha.org/aha/issues/Communicating-With-Patients/index. html

Centers for Medicare and Medicaid Services — Important Message From Medicare Notice

www.virginia.edu/uvaprint/HSC/pdf/CMS-R-193.pdf

Code of Health and Disability Services Consumers' Rights Regulation 1996 New Zealand

www.legislation.govt.nz/regulation/public/1996/0078/latest/DLM209080.html

Australian Charter of Healthcare Rights

www.safetyandquality.gov.au/internet/safety/publishing.nsf/Content/com-pubs_ACHR-pdf-o1-con/\$File/17537-charter.pdf

London Declaration for Patient Safety

www.who.int/patientsafety/patients_for_patient/London_Declaration_EN.pdf

Patient Rights (Scotland) Bill (SP Bill 42)

www.scottish.parliament.uk/s3/bills/42-PatientRights

ACT Human Rights Act

www.legislation.act.gov.au/a/2004-5/default.asp

Victorian Charter of Human Rights and Responsibilities

www.austlii.edu.au/au/legis/vic/consol_act/cohrara2006433/

SA Charter for Health and Community Services

www.hcscc.sa.gov.au/cgi-bin/wf.pl

Redesign and experience-based design

NHS Institute for Innovation and Improvement Case Studies

www.institute.nhs.uk/quality and value/experienced based design/case studies.html

Victorian Health Service Management Innovation Council — redesign website

www.health.vic.gov.au/managementinnovation/resources/what.htm

A Toolkit for Redesign in Health Care

www.ahrq.gov/qual/toolkit/index.html

Improving complaints processes

UK Department of Health guide and advice sheets

www.dh.gov.uk/en/Publicationsandstatistics/Publications

Listening, Responding, Improving: A Guide to Better Customer Care, UK Department of Health

www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicyAndGuidance/DH 095408

Consumer engagement and supporting consumer participation

Joint Commission Speak Up Program

www.jointcommission.org/speakup.aspx

Picker Institute Europe Invest in Engagement website

www.investinengagement.info/

QLD Health Community Engagement Handbook

www.healthissuescentre.org.au/documents/items/2008/04/204586-upload-00001.pdf

QLD Health Consumer and Community Participation Toolkit

www.health is suescentre. or g. au/documents/items/2008/08/226706-upload-00001.pdf

Improving Health Services through Consumer Participation: A Resource Guide

www.healthissuescentre.org.au/documents/detail.chtml?filename_num=226531

Consumer and Community Participation Self-Assessment Tool for Hospitals

www.healthissuescentre.org.au/documents/items/2008/04/204590-upload-00001.pdf

Standards

Australian Commission on Safety and Quality in Health Care — Draft National Safety and Quality Health Service Standards

www.safetyandquality.gov.au/internet/safety/publishing.nsf/Content/PriorityProgram-07

Royal Australian College of General Practitioners — Standards for General Practices (Fourth Edition) www.racgp.org.au/Content/NavigationMenu/PracticeSupport/StandardsforGeneralPractices/ Standards4thEdition.pdf

National Institute for Health and Clinical Excellence — Quality Standard for Patient Experience http://guidance.nice.org.uk/?action=byID&o=13259

Australian Commission on Safety and Quality in Health Care — Open Disclosure Standard www.http://www.health.gov.au/internet/safety/publishing.nsf/Content/3D5F114646CEF93DCA2571D5000BFEB7/\$File/OpenDisclosure web.pdf

Shared decision-making tools

Implementing Shared Decision-Making in the UK

www.health.org.uk/public/cms/75/76/313/595/Implementing%20 shared%20 decision%20 making%20 in%20 the%20 UK.pdf?realName=vgvUMW.pdf

Shared Decision Aids website — Ottowa Hospital Research Institute http://decisionaid.ohri.ca/

Sydney Health Decision Group

http://sydney.edu.au/medicine/public-health/shdg/about/aims.php

Interventions for Improving the Adoption of Shared Decision Making by Healthcare Professionals www2.cochrane.org/reviews/en/aboo6732.html

Health literacy

Invest in engagement — health literacy website www.investinengagement.info/HealthLiteracytop

US Department of Health and Human Services — health literacy website

ATTROTT III THE LOCAL TO A TOTAL TOTAL TO A TOTAL TOTAL TO A TOTAL

AHRQ Health Literacy and Cultural Competency website

www.health.gov/communication/literacy/

Appendix D Australian Charter of Healthcare Rights

AUSTRALIAN CHARTER OF HEALTHCARE RIGHTS

The Australian Charter of Healthcare Rights describes the rights of patients and other people using the Australian health system. These rights are essential to make sure that, wherever and whenever care is provided, it is of high quality and is safe.

The Charter recognises that people receiving care and people providing care all have important parts to play in achieving healthcare rights. The Charter allows patients, consumers, families, carers and services providing health care to share an understanding of the rights of people receiving health care. This helps everyone to work together towards a safe and high quality health system. A genuine partnership between patients, consumers and providers is important so that everyone achieves the best possible outcomes.

Guiding Principles

These three principles describe how this Charter applies in the Australian health system.

1 Everyone has the right to be able to access health care and this right is essential for the Charter to be meaningful.

2 The Australian Government commits to international agreements about human rights which recognise everyone's right to have the highest possible standard of physical and mental health.

Australia is a society made up of people with different cultures and ways of life, and the Charter acknowledges and respects these differences.



For further information please visit www.safetyandquality.gov.au

AUSTRALIANCOMMISSIONON SAFETYANDQUALITYINHEALTHCARE

What can I expect from the Australian health system?

MY RIGHTS	WHAT THIS MEANS
Access	
I have a right to health care.	I can access services to address my healthcare needs.
Safety	
I have a right to receive safe and high quality care.	I receive safe and high quality health services, provided with professional care, skill and competence.
Respect	
I have a right to be shown respect, dignity and consideration.	The care provided shows respect to me and my culture, beliefs, values and personal characteristics.
Communication	
I have a right to be informed about services, treatment, options and costs in a clear and open way.	I receive open, timely and appropriate communication about my health care in a way I can understand.
Participation	
I have a right to be included in decisions and choices about my care.	I may join in making decisions and choices about my care and about health service planning.
Privacy	
I have a right to privacy and confidentiality of my personal information.	My personal privacy is maintained and proper handling of my personal health and other information is assured.
Comment	
I have a right to comment on my care and to have my concerns addressed.	I can comment on or complain about my care and have my concerns dealt with properly and promptly.

Appendix E Australian Safety and Quality Framework for Health Care

.....

Australian Safety and Quality Framework for Health Care

Safe and high quality health care for Australia

Australian Health Ministers endorsed the Australian Safety and Quality Framework for Health Care in 2010. The Framework describes a vision for safe and high-quality care for all Australians and sets out the actions needed to achieve this vision. The Framework specifies three core principles for safe and high-quality care. These are that care is **consumer centred**, **driven by information**, and **organised for safety**.

The Framework sets out twenty-one areas for action that all people in the health system can take to improve the safety and quality of care provided in all healthcare settings over the next decade.

The Framework should:

- be used as the basis of strategic and operational safety and quality plans
- provide a mechanism for refocusing current safety and quality improvement activities and designing goals for health service improvement
- be used as a guide for reviewing investments and research in safety and quality
- promote discussion with consumers, clinicians, managers, researchers and policy makers about how they might best form partnerships to improve safety and quality



Tools, resources, and examples to support local use of the Framework are available from the Australian Commission on Safety and Quality in Health Care at www.safetyandquality.gov.au

www.saretyanuquanty.gov.a

AUSTRALIANCOMMISSIONon SAFETYANDQUALITYINHEALTHCARE

December 2010

Australian Safety and Quality Framework for Health Care

Safe, high-quality health is always:

What it means for me as a consumer or patient:

Areas for action by people in the health system:



CONSUMER CENTRED

This means

Providing care that is easy for patients to get when they need it.

Making sure that healthcare staff respect and respond to patient choices, needs and values

Forming partnerships between patients, their family, carers and healthcare providers.

I can get high-quality care when I need it.

I have information I can understand. It helps me to make decisions about my health care.

I can help to make my care safe.

My health care is well organised. The doctors, nurses and managers all work together. I feel safe and well cared for.

I know my healthcare rights.

If something goes wrong, my healthcare team look after me. I receive an apology and a full explanation of what happened. 1.1 Develop methods and models to help patients get health services when they need them.

- 1.2 Increase health literacy.
- 1.3 Partner with consumers, patients, families and carers to share decision making about their care.
- **1.4** Provide care that respects and is sensitive to different cultures.
- 1.5 Involve consumers, patients and carers in planning for safety and quality.
- **1.6** Improve continuity of care.
- 1.7 Minimise risks at handover.
- 1.8 Promote healthcare rights.
- 1.9 If something goes wrong, openly inform and support the patient.



DRIVEN BY INFORMATION

This means:

Using up-to-date knowledge and evidence to guide decisions about care.

Safety and quality data are collected, analysed and fed back for improvement.

Taking action to improve patients' experiences.

My care is based on the best knowledge and evidence.

The outcome of my treatment and my experiences are used to help improve care.

- 2.1 Use agreed guidelines to reduce inappropriate variation in the delivery
- **2.2** Collect and analyse safety and quality data to improve care.
- 2.3 Learn from patients' and carers' experiences.
- **2.4** Encourage and apply research that will improve safety and quality.

ORGANISED FOR SAFETY

This means making safety a central feature of how healthcar facilities are run, how staff work and how funding is organised.

I know that the healthcare team, managers and governments all take my safety seriously.

The health system is designed to provide safe, high-quality care for me, my family and my carers.

When something goes wrong, actions are taken to prevent it happening to someone else.

- **3.1** Health staff take action for safety.
- **3.2** Health professionals take action for safety.
- 3.3 Managers and clinical leaders take action for safety.
- 3.4 Governments take action for safety.
- 3.5 Ensure funding models are designed to support safety and quality.
- **3.6** Support, implement and evaluate e-health.
- **3.7** Design and operate facilities, equipment and work processes for safety.
- **3.8** Take action to prevent or minimise harm from healthcare errors.

AUSTRALIANCOMMISSIONon SAFETYANDQUALITYINHEALTHCARE

December 2010

Appendix FPerth Declaration for Patient Safety

- PERTH DECLARATION FOR PATIENT SAFETY

 The Perth Declaration for Patient Safety and the Australian workshop information is available at www.hcon.corg.au.

 We, the participants of the inaugural Australian Patients for Patient Safety Workshop, convened in July 2009 to share profound health care experiences in our lives and to take forward our call for action to improve patient safety in Australia.

 We are patients, family members, carers and health professionals—people from all walks of life. Each one of us is a testament to the personal experience of unintended harm in health care and the continuing. Much of that harm was preventable.

 We declare

 Policies and protocols alone have not made us safer. This problem is systemic, widespread and deep-rotocol. The fact that any person or family could one day experience needless devastating harm within the health care system is unacceptable.

 Action must be taken now across all aspects and all levels of health care to prevent more harm occurring to others.

 Our trusted health care to prevent more harm occurring to others.

 Our trusted health care workers and managers must receptise that we, your patients and our families, are an invaluable asset and resource for improving patients afterly. We offer our form our hard-wow misdom and partner will us to make lasting change.

 We are the owners and funders of our hard-wow misdom and partner will us to make lasting change.

 We are the owners and funders of our hard-wow misdom and partner will us to make lasting change.

 We are the owners and funders of our hard-wow misdom and partner will us to make lasting change.

 We are the owners and funders of our hard-wow misdom and partner will us to make lasting change.

 We are the owners and funders of our hard-wow misdom and experiences. Seek to learn from our hard-wow misdom and partner will us to make lasting change.

 We need to receive a seek to learn from our hard-wow misdom and partner will us to make lasting change.

 We are patient for the mass of the downer will be a seek to th

Index	
	1111
A	C
Aboriginal and Torres Strait Islander Peoples, 35, 39–40	CAHPS surveys and database, 9, 14, 15
accountability, 26–7, 31, 32, 48, 54–5	Calvary Health Care Simply Better Program, 42–3
accreditation systems, 31, 43	Camden Hospital, 37
advanced care planning programs, 39, 44	Campbelltown Hospital, 37
adverse clinical events, 9–10, 21, 37	Canada, 11, 19, 21
Agency for Healthcare Research and Quality (US), 14	Canberra Hospital, 37
Always Events program, 28	carers, 22–3, 25, 26
ambulance service, 38	case studies
Australia see also specific states and territories adverse clinical incidents inquiries, 37	Calvary Health Care Simply Better Program, 42–3 Flinders Medical Centre Emergency Dept
health literacy, 21	Redesign, 43
health records, 22, 39	Medical College of Georgia Health System, 27
healthcare system, 31–5	Mental Health Peer Service, 44–5
key organisations, 77–80	Patient First Program, 43–4
national strategies and initiatives, 1, 32–5	RACGP quality framework, 43
patients' rights, 19, 32, 82, 83, 86, 91 performance monitoring, 1, 32, 48, 49	Respecting Patient Choices Program, 44 Victorian consumer advisory councils, 17–18
standards, 1, 21, 31, 33–4, 38, 84	Centers for Medicare and Medicaid Services (US), 14, 15, 16
state and territory initiatives, 37–45	Centre for Health Communication and Participation
Australian Bureau of Statistics, 21, 37	(Victoria), 41
Australian Capital Territory	change management, 25–7, 50–1
ACT Health initiatives, 38–9	checklists see tools
key organisations and programs, 79	children's health care, 8–9 see also family-centred care
Australian Charter of Healthcare Rights, 19, 32, 86	chronic disease strategy, 34
Australian Institute for Patient and Family Centred	clinical outcomes associated with patient-centred
Care, 38	care, 1, 9–10, 27
Australian Nurse Family Partnership Program, 35	collaborative patient-centred care, 11
Australian Patients for Patient Safety Workshop, 42, 91	Commissioning for Quality and Innovation (CQUIN)
Australian Safety and Quality Framework for Health	schemes, 14, 15–16
Care, 33, 47, 88–9	community Pharmacy Agreements (Australia) 24.5
В	Community Pharmacy Agreements (Australia), 34–5
'best buys' (improving patient experience), 18	complaints processes, 20, 38, 42, 83 Consumer Assessment of Healthcare Providers and
Bundaberg Hospital, 37	Systems (CAHPS) surveys, 9, 14, 15
	consumer-centred care, defined, 8
	consumer choice 14-15, 20, 44, 82

consumer engagement <i>see also</i> patient engagement consumer advisory groups, 17–18, 23, 27, 39, 40	healthcare professionals education and training, 28–9, 53–4
importance of, 17–18	healthcare rights, 19, 32, 82–3, 86, 91
peer support in mental health, 44–5	Hospital Consumer Assessment of Healthcare Providers
resources and tools, 83	and Systems (H-CAHPS) survey, 9, 14
state and territory initiatives, 38–42	hospitals see also health service providers
consumer organisations, 40, 77–8	adverse clinical incidents, 37
consumer organisations, 40, 77	advisory councils, 23, 27
D	patient-centred care approaches and benefits,
decision-making tools, 84	1, 9–10, 27
Declaration of Alma Ata, 17	patient feedback influences, 24, 27, 38, 39, 41–3, 53
definitions, 7, 8–9, 10, 11	safety improvement, 20–1 see also quality and safety
design of the care environment, 24, 27, 38, 39, 41–3, 53, 83	of health care
acoign of the care environment, 24, 27, 30, 39, 41 3, 33, 03	staff communication skills, 24–5, 29, 42
E	values and employee behaviour, 25
education	Hunter New England Health Service, 20
consumer health literacy, 21–2, 41, 44, 84	Truffeet IVEVV England Health Service, 20
of healthcare professionals, 28–9, 49–50, 53–4	T
efficiency gains, 17	Improving Patient and Staff Experience Program
England see United Kingdom	(NSW), 26
Enhancing Practice Program, 26	Improving the Patient Experience Program (Victoria), 41
Essentials of Care Program, 39	Indigenous Australians see Aboriginal and Torres
evidence-based care	Strait Islander Peoples
evidence-based care/patient-centred care tension, 25	information resources, 21–2, 39–40, 77–84
evidence for patient-centred care, 1, 9–10, 67–75	Institute for Healthcare Improvement (US), 28
research needs, 50	Institute for Patient- and Family-Centered Care (US),
experience-based design, 24, 27, 38, 39, 41, 42–3, 53, 83	28, 48
ent errorses senses errorer, -4, -4, 20, 23, 4-, 4- 3, 23, 62	Institute of Medicine (US), 1, 10, 47
F	International Alliance of Patients' Organizations, 8, 48
family-centred care, defined, 8–9	international organisations, 27–9, 77
family member involvement, 22–3 see also carers	international organisations, 27 9, 77
feedback <i>see</i> patient experience data collection	K
financial incentives for quality improvement, 15–16	Kenneth B Schwartz Center, 29
Flinders Medical Centre Emergency Department	key organisations, 27–9, 77–80
Redesign, 43	King Edward Memorial Hospital, Perth, 37
Four Hour Rule Program, 42	The King's Fund, 11, 29
3 1	
G	L
general practice, 9, 23, 31, 43	leadership, 25, 50–1
governance	Listening Posts initiative, 20
leadership and change management, 25–7, 50–1	literacy see health literacy
patient engagement in, 8, 17, 21, 23, 25, 26, 33	
	M
H	malpractice claims, reduction in, 9–10
Health and Disability Code (NZ), 11	Medical College of Georgia Health System, 10, 27
health care see healthcare professionals; patient-centred	medical records, 22, 39
care; quality and safety of health care; in specific	mental health, 22, 34, 41, 44–5
countries see names of countries	
Health Issues Centre (Victoria), 23, 41	N
health literacy, 21–2, 41, 44, 84	National Aboriginal Community Controlled Health
health records, 22, 39	Organisation, 35
health service providers see also hospitals	National Chronic Disease Strategy (Australia), 34
accreditation, 31, 43	National Health Performance Authority (Australia), 32
complaints processes, 20, 38, 42, 83	National Health Service (UK)
efficiency, 17	Constitution (patients' rights), 19
patient-centred care benefits, 1, 9–10, 27	public value, 16–17
performance monitoring and reporting, 14–16	quality indicators and outcomes, 11, 13–16
service design/redesign, 24, 27, 38, 39, 41–3, 53, 83	service redesign, 24, 83
staff development and satisfaction, 25, 26, 29, 53–4	

National Healthcare Agreement (Australia), 32	leadership and change management, 25–7, 50–1
National Institute for Health and Clinical Excellence	patient/consumer engagement see consumer
(UK), 14	engagement; patient engagement
National Mental Health Plan (Australia), 34	patient feedback see patient experience data
National Primary Health Care Strategy (Australia), 34	collection
National Safety and Quality Health Service Standards	rights-based charters, 19, 32, 82–3, 86, 91
(Australia), 31, 33–4, 84	staff development and satisfaction, 25, 26, 29, 53–4
National Service Improvement Frameworks	patient choice in health care, 14–15, 39, 44, 82
(Australia), 34	patient engagement, 20–5 see also consumer engagement
New South Wales	peer support in mental health, 44–5
adverse clinical incidents inquiries, 37	resources and tools, 83
NSW Health initiatives, 39	state and territory initiatives, 38–42
organisations and programs, 21, 22, 26, 39, 78–9	strategies and recommendations, 51–2
patient experience reporting, 20, 37, 38, 49	WHO programs, 27–8
New Zealand, 11, 19	patient experience
Northern Territory	improvement strategies, 18, 28–9, 38–42
Department of Health and Families initiatives,	influencing design of services/environment, 24, 27,
39-40	38, 39, 41–3, 53, 83
key organisations and programs, 80	as outcome metric, 15, 16–17
'nothing about me without me' principle, 8	and public value, 16–17
nurse family partnership home visiting models, 35	quality of care/patient experience association, 9–11
naise family partitioning frome visiting fromeis, 55	satisfaction ratings, 9–10, 27, 37–8, 42
0	patient experience data collection see also performance
open disclosure policies, 21	monitoring and reporting
Open Notes project, 22	methods, 13–14, 19–20, 29, 48–9
Organisation for Economic Co-operation and	patient feedback, 14–15, 19–20, 24, 37–8, 40–2
Development, 11	resources and tools, 81–2
organisational culture and values, 25, 26, 54–5	strategies and recommendations, 52–3
organisational readiness to implement patient-centred	surveys, 13–14, 28, 32, 37–8, 40, 42, 48
care (checklists), 63–6	Patient First Program, 42, 43–4
organisational strategies, 25–7, 50–5 <i>see also</i> governance;	Patient Voice for Compassionate Care program, 29
patient-centred care strategies	patients' rights, 19, 32, 82–3, 86, 91
organisations (key organisations), 27–9, 77–80	'pay for performance,' 15, 16
organisations (key organisations), 27 9, 77 00	peer support in mental health, 44–5
P	people-centred care concept, 11
P4P approaches, 15, 16	Performance and Accountability Framework (Australia),
Partnering with Patients program, 21	32, 48
patient-centred care see also consumer engagement;	performance monitoring and reporting <i>see also</i> patient
patient engagement	experience data collection; quality and safety of
in Australia see Australia	health care
benefits, 1, 9–10, 27	
case studies, 17–18, 27, 42–5	Australia, 1, 32, 48, 49
concepts and definitions, 1, 7–9, 10, 11	health service providers, 14–16
as dimension of high-quality care, 1, 10–11, 47	other countries, 11, 13–16
evidence base for, 1, 9–10, 50, 67–75	strategies and recommendations, 1, 48–9, 52–3
evidence-based care/patient-centred care tension, 25	transparency, 49
key organisations, 27–9, 77–80	person-centred care, defined, 8
	personalised care, defined, 8
key resources and tools, 80–4	Perth Declaration for Patient Safety, 42, 91
policy drivers, 13–16	pharmacists, 34–5
public value, 16–17	Physician Compare web-based tool, 15
in specific countries see names of countries	Physician Quality Reporting System, 15
patient-centred care strategies, 1, 18–27	Picker Institute, 7, 28, 29
accountability strategies, 26–7, 32, 48, 54–5	Picker Institute Europe, 14, 18, 21, 28
in Australia, 1, 32–5, 37–45 see also Australia	Planetree, 28–9, 53
communication styles and skills, 24–5, 29, 42	policy drivers, 13–16
experience-based design, 24, 27, 38, 39, 41–3, 53, 83	primary health care, 9, 34, 35
implementation of, 1, 47–55, 63–6	Productive Ward program, 40
information and education for consumers, 21–2	providers see health service providers
key resources and tools, 80–4	public value, 16–17

0 10 10 1 7 7 1 (7777)	. 1 1. 1. 1 1. 1 1. 0
Quality and Outcomes Framework (UK), 14, 15	standardised tools and indicators needed, 48
quality and safety of health care	Top 5 program, 22
adverse clinical incidents inquiries, 37	Torres Strait Islanders see Aboriginal and Torres Strait
Australian initiatives, 31–5, 38–42	Islander Peoples
consumer/patient engagement in, 17–18, 20–1	Transforming Care program, 40
financial incentives, 15–16, 31	trust see public value
high-quality care and patient-centredness,	
1, 10–11, 47	U
models and frameworks, 43, 47, 88–9	United Kingdom
patient experience/quality of care association, 9–11	leading organisations, 29
patients' rights, 19, 32, 82–3, 86, 91	NHS reform/improvement, 11, 13–14, 24, 83
public value and efficiency gains, 16–17	open disclosure policies, 21
standards, 31, 33–4, 38, 84	patient experience data collection, 13–14, 19
Queensland	patients' rights, 19, 82
hospital advisory groups, 24	performance reporting and choice of provider, 14–15
organisations and programs, 40, 79	quality improvement incentives, 15
Queensland Health initiatives, 37, 40	United States
Queensland Children's Hospital Youth Advisory	leading organisations, 28–9
Council, 24	medical records accessibility, 22
	open disclosure policies, 21
R	patient-centred organisations, 10, 27
relationship-centred care, defined, 9	patient engagement, 21
research findings, 1, 9–10, 67–75 see also information	patient experience data collection, 14
resources	patients' rights, 19, 82
research needs, 50	performance reporting and choice of provider, 15
Respecting Patient Choices Program, 39, 44	quality care, 10, 16
'responsiveness' concept, 7	user-centred design, 24 see also experience-based design
rights of patients, 19, 32, 82–3, 86, 91	
Royal Australian College of General Practitioners,	V
38, 43, 84	values see organisational culture and values
Royal Melbourne Hospital, 37	Victoria
Royal North Shore Hospital, 37	consumer advisory councils, 17–18
2	Department of Health initiatives, 41
S	Enhancing Practice Program, 26
safety see quality and safety of health care	Health Issues Centre, 23, 41
Schwartz Center Rounds, 29	key organisations and programs, 79
Scotland, 19 see also United Kingdom	
South Australia	W
key organisations and programs, 80	web sites, 14–15, 28, 29, 77–84
SA Health initiatives, 40	Western Australia
Southern Mental Health Peer Service (SA), 44–5	key organisations and programs, 80
Speak Up program, 21	WA Health initiatives, 19, 41–2, 43–4
staff development and satisfaction, 25, 26, 29, 53–4	Work Foundation (UK), 16
standards see quality and safety of health care	workplace environment
strategies see patient-centred care strategies	culture, 25, 26, 42–3, 50–1, 54–5
Studer Group, 29, 39, 42, 53	physical care environment, 24, 27, 38, 39, 41, 42, 53–4
surveys see patient experience data collection	World Health Organization, 7, 11, 27–8
т	Y
	youth advisory groups, 24
Tasmania Department of Health and Human Services	1 marroor 1 Broaks, - 1
Department of Health and Human Services initiatives, 40–1	
key organisations and programs, 80	
terminology (definitions), 7, 8–9, 10, 11	
tools, 28, 29, 80–4	
consumer information and choice websites, 14–15, 29 organisational readiness checklists, 63–6	







Australian Commission on Safety and Quality in Health Care Level 7, 1 Oxford Street

Darlinghurst NSW 2001

Telephone: (02) 9126 3600 Email: mail@safetyandquality.gov.au www.safetyandquality.gov.au

AUSTRALIANCOMMISSIONON SAFETYANDQUALITYINHEALTHCARE